Ŋ	AD-A1	41 144	ENH	ANCED	TRACK	ING OF		RNE TE		USING	NULTI	PLE	1/4	4	
	UNCLAS	SIFIE	NRI DEC	GHT-PF 83 AF	TTERS	ON AFB	OH SC M-4	HOOL C	FENG	I R	i SUI F/G 1	ZU 2/1	NL		
														-	





(2)



AFIT/GE/EE/84M-4

Start and a second s

1 4 A A

81

- - - C-

ENHANCED TRACKING OF AIRBORNE TARGETS USING MULTIPLE MODEL FILTERING TECHNIQUES FOR ADAPTIVE FIELD OF VIEW EXPANSION

THESIS

Robert I. Suizu First Lieutenant, USAF

AFIT/GE/EE/84M-4



ć. u



Approved for public release; distribution unlimited.

AFIT/GE/EE/84M-4

ENHANCED TRACKING OF AIRBORNE TARGETS USING MULTIPLE MODEL FILTER TECHNIQUES FOR ADAPTIVE FIELD OF VIEW EXPANSION

THESIS

Presented to the Faculty of the School of Engineering of the Air Force Institute of Technology Air University In Partial Fulfillment of the Requirements for the Degree of Master of Science in Electrical Engineering

Robert I. Suizu

Accession For GPA&I NTIS DTIC TAB Nersel Scotter 1111

First Lieutenant, USAF

December 1983

Approved for public release; distribution unlimited

Preface

This study was part of an ongoing effort at the Air Force Institute of Technology to design a tracking algorithm for use with the Air Force Weapons Laboratory's high energy laser weapon system. The purpose of this $\frac{f_{A} e_{S} f_{A}}{\text{study}}$ was to take previously developed tracker algorithms and incorporate a multiple model adaptive filter algorithm into the existing structure. This approach was intended to provide adaptive expansion of the effective tracker field of view, which in turn would increase the tracker's ability to maintain lock on highly dynamic, close range targets.

I wish to express my deepest thanks to my thesis advisor, Dr. Peter S. Maybeck, for his guidance, motivation, and above all, patience. His willingness to expend that extra effort to keep this research on track was invaluable to the completion of this study.

I would also like to thank my family, especially my parents, whose love and support keep me going when things aren't going well, and inspire me to do more when they are. I would also like to thank my fellow classmates in GE-83D, their friendship made the last 18 months at AFIT a bearable burden.

Table of Contents

1. 22

Sec. 20

5. 6. 7

and the second second

a a section a section of

Ø

3

				Page
Prefa	.ce.	• • • • •	•••••	ii
List	of F	igures .		vi
List	of Ta	ables	•••••	x
List	of S	ymbols .		xi
Absti	act	• • • • •		xvi
I.	Intro	oduction .	• • • • • • • • • • • • • • • • • •	I-1
	1.1	Background	9	T-2
	1 2	Drohlam		T _ 7
	1 • 4	1 0 1	V · · · · · · · · · · · · · · · · · · ·	1-7 T_0
			Extended Kalman Filter Tracker .	7-0
		1.2.2	Linear Kalman Filter/Correlator	
			Tracker	I-12
	1.3	Overview		I-15
II.	Trutl	n Model .	•••••	11-1
	2.1	Introduct	ion	II-1
	2 2	Target Cel	ntroid Offget Model	TT_1
	2.2	larget te		
	2.3	Atmospher	LC Disturbances	11-3
	2.4	Target Dy	namics Models	11-7
	2.5	Overall St	tate Space Model	II - 13
	2.6	Target Co	ordinate Frames	II-16
	2.7	Target Tra	njectories	II-17
	2.8	Neagilreme	nt Model	TT-20
	2 0	Mercat Inc		TT_22
	4.7	Taryet Im		11-22
	2.10	Spatially	correlated Background Noise	11-25
	2.11	Summary .	• • • • • • • • • • • • • • • • •	11-28
III.	Trac	er Config	urations	III-1
	•			^
	3.1	Introduct.		III-1
	3.2	Extended	Kalman Filter Tracker	III-3
		3.2.1	Gauss-Markov Target Acceleration	
			Model	III-3
		3.2.2	State Propagation of Gauss-Markov	
			Nodel	TTT-5
		2 2 2	Constant Murn Data Marsat	*** 3
].2.]	Constant Turn Rate Target	
		• • •	ACCELERATION MODEL	111-1
		3.2.4	Measurement Update	111-10
		3.2.5	Derivation of Non-linear and	
			Linearized Intensity Functions .	III-12
		3.2.6	Two Dimensional Fourier Transform	III-13
		3.2.7	Shifting Property of the Fourier	
			Trangform	TTT-16
		2 2 0	Funnantial Cmonthing	
		J & Z & D		TTT

NEW MARKETER PROPERTY AND A SALE A

Table of Contents

STATE WHEN TO STATE AND A STATE

B

Y

Page

CLIPC COL

		3.2.9	Deriva	ative	Proper	rty o	f the	e Fo	urie	er
		_	Trans	form .	• • •		• •	• •	• •	. III-19
	3.3	Linear Kal	lman F	ilter/	Correl	lator	Trac	cker	• •	. III-19
		3.3.1	FFT C	orrela	tor .		• •	• •	•	. III-20
	3.4	Estimation	n of D	ynamic	Drivi	ing N	oise	• •	•	. III-23
	3.5	Summary .	• • •	• • •	• • •		• •	• •	• •	. III-26
IV.	Mult	iple Model	Adapt:	ive Fi	lter .	• • •	• •	••	• •	. IV-1
	4.1	Introduct	ion .	• • •	• • •		• •	• •	•	IV-1
	4.2	Multiple N	Model 1	Adapti	ve Fil	lter	Algor	ith	m .	. IV-2
	4.3	Implementa	ation (of the	Multi	iple	Mode]	L		
		Algorithm	• • •	• • •	• • •		• •	• •	•	. IV-8
		4.3.1	Imple	mentat	ion in	n the	Koze	emch	ak	
			Track	er Con	figura	ation	• •	• •	•	. IV-10
		4.3.2	Imple	mentat	ion ir	n the	Mill	lner		
			Track	er Con	figura	ation	• •		•	IV-14
	4.4	Summary .		• • •	•••		• •		• •	IV-15
		-								
v.	Perf	ormance Ana	alysis	• • •	• • •	• • •	•••	• •	•	V-1
	5.1	Introduct	ion .	• • •	• • •		• •		•	. V-1
	5.2	Derivation	n of T	racker	Stati	istic	s .	• •	•	. V-1
	5.3	Performance	ce Plo	ts	• • •		• •	• •	•	. V~5
	5.4	Parameter	Value	5			• •	• •	•	. v-9
		5.4.1	Truth	Model	Para	neter	s.		• •	. v-9
		5.4.2	Data 1	Proces	sing I	Param	eters	3.	•	v-10
		5.4.3	Filte	r Para	meter	Valu	es ,			V-12
	5.5	Tracker Pe	erform	ance A	gainst	t Tar	get	• •	•	
		Trajector	les .		·		, , ,		•	v−18
		5.5.1	Evalu	ations	Usind	7 Tra	iecto	nrv	1	V-19
		5.5.2	Evalua	ations	Using	, <u></u> - , Тга	jecto		2	V-22
		5.5.3	Evalu	ations	Usind	7 Tra	jecto)rv	3	V-28
		5.5.4	Evalu	etione	lleind	7 Tra	jecto)TV	Δ Δ	V-33
		5.5.5	Change	actons	Gianal]_+^_	Noiee	∕⊢j Da	tia tia	V-33
		5 5 6	1d Ho	c Chan	a a g + c	- 20- - 211	+0-			••55
		3.3.0	Condi	tional	yes u Drohi	, ,,, ,,,,,,	tion			V-25
			Condi	CIONAI	FLODE		LIES	• •	• •	v-33
VI.	Conc	lusions and	i Reco	mmenda	tions	• •	• •	• •	•	. VI-1
Bibli	logra	phy	• • •	• • •	• • •		• •	• •	•	BIB-1
1		1	Vel me		/	- 1-	+o- "		ka -	
vbbei	MTDI	ni Llnedi Domform	SmlbA	U 511C Dlate	er/CO	reig	COT	Tac	Ker	3_4
		refior	uance	LIOLS	• • •	• • •	• •	• •	•	-A-T
Inner	dir '	R: Extands	d Kal	man Pi	lter "	Frack	er (lane	e	
whher		Marban Marban	local	aratiri.	n Mode	all D	arfor	man	~~	
		DI AFO	N0021	et a LTV		31) E	GLIVI	. metil		B_1
		FICE	• • •	• • •	• • •	• • •	• •	• •	• •	, <u>p-</u> 1

Table of Contents

1. 1. 1. A. A.

B

ኯዄ፟ጟ፟ቘጞዀጞቘጞዄጞ፟ዀዀጞቘጞዀጞዀጞቘጞዀጞቚጞቚጞዀጞዀጞዀጞዀጞ፟፟፟፟፟፟፟ቜዀዸዄ

Page

Appendix C: Extended Kalman Filter Tracker (Constant Turn Rate Acceleration Model) Performance Plots C-1

r e

£ 4

R

Figure		Page
I - 1	Data Processing Algorithm for the Extended Kalman Filter Tracker	I-9
1-2	Data Processing Algortihm for the Linear Kalman Filter/Correlator Tracker	I-13
II-1	Inertial Coordinate Frame	II-10
II-2	Azimuth Geometry	II-10
II-3	Elevation Geometry	II - 11
II-4	Trajectory 1	II-17
II - 5	Trajectory 2	II-18
II-6	Distribution of Hot-Spots	II-21
II-7	Image Projection	II-23
II-8	Pixel Numbering Scheme	II-26
IV-1	Multiple Model Filter Algorithm	IV-1
V-1	Y-Minus Error, 20g Pull-up Maneuver	v- 7
V-2	Y-Plus Error, 20g Pull-up Maneuver	V-8
V-3	Simulation Mnemonic Codes	V-16
A-1	Performance Plot for LT1MF	A-2
A-2	Performance Plot for LT1F1	A-4
A-3	Performance Plot for LT1F2	A-6
A-4	Performance Plot for L2T2MF	A-8
A-5	Performance Plot for L2T2F1	A-10
A-6	Performance Plot for L2T2F2	A-12
A-7	Performance Plot for L10T2MF	A-14
A-8	Performance Plot for L10T2F2	A-16
A-9	Performance Plot for L20T2MF	A-18
A-10	Performance Plot for L20T2F2	A-20

đ

Figure	Pac	je
A-11	Performance Plot for L2T3MF A-2	22
A-12	Performance Plot for L2T3F1 A-2	24
A-13	Performance Plot for L2T3F2 A-2	26
A-14	Performance Plot for L10T3MF A-2	28
A-15	Performance Plot for L10T3F2 A-:	30
A-16	Performance Plot for L20T3MF A-3	32
A-17	Performance Plot for L20T3F2 A-3	34
A-18	Performance Plot for L2T4MF A-:	36
A-19	Performance Plot for L2T4F1 A-4	40
A-20	Performance Plot for L2T4F2 A-4	42
A-21	Performance Plot for L10T4MF A-4	44
A-22	Performance Plot for L10T4F2 A-4	46
A-23	Performance Plot for L20T4MF A-4	18
A-24	Performance Plot for L20T4F2 A-S	50
A-25	Performance Plot for LS2T2MF A-S	52
A-26	Performance Plot for LS10T2MF A-S	54
A- 27	Performance Plot for LS20T2MF A-S	56
A-28	Performance Plot for LA10T2MF A-	58
A-29	Performance Plot for LA20T2MF A-0	50
B-1	Performance Plot for GT1MF B	-2
B-2	Performance Plot for GT1F1 B-	-4
B-3	Performance Plot for GT1F2 B.	-6
B-4	Performance Plot for G2T2MF B.	-8
B-5	Performance Plot for G2T2F1 B-:	LO
B-6	Performance Plot for G2T2F2 B-1	L2

NARANA NARANA NA MANANA NA MAA

.....

\$

Figure		Page
B- 7	Performance Plot for G10T2MF	B-14
B-8	Performance Plot for G10T2F2	B-16
B-9	Performance Plot for G20T2MF	B-18
B-10	Performance Plot for G20T2F2	B-20
B-11	Performance Plot for G2T3MF	B-22
B-12	Performance Plot for G2T3F1	B-24
B-13	Performance Plot for G2T3F2	B-26
B-14	Performance Plot for G10T3MF	B-28
B-15	Performance Plot for G10T3F2	B-30
B-16	Performance Plot for G20T3MF	B-32
B-17	Performance Plot for G20T3F2	B34
B-18	Performance Plot for GA10T2MF	B-36
B-19	Performance Plot for GA20T2MF	B-38
C-1	Performance Plot for CT1MF	C-2
C-2	Performance Plot for CT1F1	C-4
C-3	Performance Plot for CT1F2	C-6
C-4	Performance Plot for C2T2MF	C-8
C-5	Performance Plot for C2T2F1	C-10
C-6	Performance Plot for C2T2F2	C-12
C-7	Performance Plot for C10T2MF	C-14
C-8	Performance Plot for C10T2F2	C-16
C-9	Performance Plot for C20T2MF	C-18
C-10	Performance Plot for C20T2F2	C-20
C-11	Perfc nc. Plot for C2T3MF	C-22
C-12	Performance Plot for C2T3F1	C-24

12.00 (2.00

Ø

Figure												Page
C-13	Performance	Plot	for	C2T3F2	•	•	•	•	•	•	•	C-26
C-14	Performance	Plot	for	C10T3MF	•	•	•	•	•	•	•	C-28
C-15	Performance	Plot	for	C10T3F2	•	•	•	•	•	•	•	C-30
C-16	Performance	Plot	for	C20T3MF	•	•	•	•	•	•	•	C-32
C-17	Performance	Plot	for	C20T3F2	•	•	•	•	•	•	•	C-34
C-18	Performance	Plot	for	CA10T2ME	7	•	•	•	•	•	•	C-36
C-19	Performance	Plot	for	CA20T2M	7	•	•	•	•	•	•	C-38

List of Tables

1000

Table		Page
V-1	Individual Filter Evaluations	V-16
V-2	Trajectory 1 Evaluations	V-21
V-3	Trajectory 2 Evaluations (2g)	V-23
V-4	Trajectory 2 Evaluations (10g)	V-25
v- 5	Trajectory 2 Evaluations (20g)	V-26
V-6	Trajectory 3 Evaluations (2g)	V-29
V- 7	Trajectory 3 Evaluations (10g)	V-30
V-8	Trajectory 3 Evaluations (20g)	V-31
V-9	Trajectory 4 Evaluations	V-34
V-10	Trajectory 2 Evaluations (SNR=10)	V-37

and the second second

L	is	t (of	Sym	bols	
-						

- T I

E)

C.S.S.

RINA

1

Maria and

Ø

Ţ

Symbol	
AR	aspect ratio
<u>A</u>	direction cosine matrix
Ap	area of a single pixel
a,b	atmosphere break frequencies
<u>a</u>	vector of uncertain parameter values
aı	inertial acceleration
a	smoothing constant in averaging process
a(t)	azimuth
β (t)	elevation
B	control input matrix
<u>C</u> T	truth model output matrix
E	sample mean error
E{ }	expected value
<u>e</u>	unit vector
e	error between truth model system output and fill system output
F	system plant matrix
<u>F</u> []	linearized function of the filter states
F()	Fourier transform operation
F-1()	inverse Fourier transform operation
<u>f</u> []	nonlinear function of filter states
f _x ,f _y	spatial frequencies
G	system noise input matrix
ũ	complex-valued function in spatial frequency domain
ā	complex-valued function in the space domain

an and the lot

Ţ

.....

Symbol	
<u>h</u> []	nonlinear intensity function
<u>H</u> []	linearized intensity function
H	system output matrix
ī	identity matrix
I _{max}	maximum intensity of the hot-spot
K	Kalman filter gain
<u>•</u>	system state transition matrix
NXN	size of data array to be processed
ω	constant turn rate
ρ	range to target
<u>P</u>	state covariance matrix or dispersion matrix of target image
P	conditional probability
Qfd	dynamics driving noise matrix
Q _{TD}	noise covariance kernel descriptor for truth model dynamics
<u>R</u>	measurement noise covariance matrix
r	residual vector
r _h	horizontal range
^r k,1	correlation coefficient between k-th and i-th pixels
đ	standard deviation of a process or dispersion of an intensity distribution
T _{AF}	correlation time assumed for atmospheric jitter
^T DF	correlation time assumed for target dynamics
t,	time
Δt	sample time

xii

E

Ö

S

Symbol	
0	orientation angle in the FLIR image plane
<u>u</u>	deterministic control input
<u>v</u>	measurement noise vector - zero-mean, white Gaussian noise
<u>v</u> 1	inertial velocity vector
<u>v</u> '	unit variance, zero mean, white, Gaussian noise vector
<u>w</u>	white Gaussian noise vector
<u>x</u>	general state vector
<u>x_I, y_I, z_I</u>	inertial axes
Δ <u>×</u>	state vector update
ŧ	angle between inertial velocity vector and the plane that is perpendicular to the line of sight
(x,y)	FLIR frame coordinates
<u>\$</u> (t)	current averaged data frame
<u>y</u> (t)	current data frame
<u>Z</u>	measurement process
<u>z</u>	measurement vector
1	perpendicular
LOS	line of sight
rms	root mean squared
	Cholesky square root
Subscript	<u>.8</u>
A,a	atmospheric disturbances
AF	filter model for atmospheric jitter

Symbol	
a	azimuth direction
ß	elevation direction
с	centroid
D	target dynamics
DF	filter model for target dynamics
đ	discrete time
F	filter
I	inertial reference frame
i	i-th time frame
k	k-th term or filter
kl	kl-th pixel
m	m-th hot-spot or ellipsoid
n	normalized
0	initial value
pv	direction perpendicular to the target velocity
ppv	direction mutually perpendicular to the target velocity and pv direction above
r	direction along line of sight to the target from the tracker
T	truth model
v	direction of a arget velocity vector

Superscripts

以上为无法无法法无法法无法无法无法无法不能不能不能不能不能不能不能不能。

P.R. C. B. C.

San Real Property in

- A4

ĩ

c value after control application

<mark>Maddalada Balana a shinta shinta shinta babala an asar a shinta babada a shinta a shinta a shinta a shinta a sh</mark>

estimate



Symbol

- before measurement update
- + after measurement update
 - complex variable

Abstract

Previous studies at the Air Force Institute of Technology have developed two tracker algorithms which provide significant improvements in tracker performance against close-range, highly-dynamic, airborne targets, over a currently used direct correlation method. Digital signal processing techniques are used to derive a target shape function from available sensor information. In one formulation, this shape function is used in the measurement update portion of an extended Kalman filter to determine the target position offsets from the center of the sensor field of view. In the other tracker, the offsets are derived and incorporated into the tracking algorithm by using the shape function as a template for an enhanced correlator/linear Kalman filter structure. Combining these offsets with any a priori target information allows the tracker to produce better target position estimates than achievable from a conventional correlator. This research investigates using a multiple model approach for the adaptive expansion of the effective tracker field of view as a means of increasing the dynamic range of the tracker. Two independent Kalman filters, each receiving measurement information from a shared sensor, generate target position estimates. The multiple models are created by tuning the respective filters for "best" performance at differing conditions of exhibited target behavior and differing the physical size of their respective fields of view. Adaptive expansion of the

xvi

tracker field of view is obtained by summing the weighted estimates of the two filters.

Sec. Sec.

5

Sec. No.

48

xvii

ENHANCED TRACKING OF AIRBORNE TARGETS USING MULTIPLE MODEL FILTERING TECHNIQUES FOR ADAPTIVE FIELD OF VIEW EXPANSION

I. Introduction

Even though lasers were first developed in the late 1950s, they have generally been considered as part of future technology by the public at large. Recent advances in laser technology have allowed their use in everyday life to increase steadily. Today, lasers are widely used in surgery, scientific laboratories, industry, and military applications, because of their ability to deliver almost instantaneously, finely focused, concentrated beams of energy onto a particular spot.

It is these charactristics that make the laser very attractive as a possible weapon system. Because of its almost instantaneous transmission of energy from the weapon to the target, it eliminates the need for computing the lead angle necessary for a ballistic projectile to intercept the target. However, a number of factors will affect a laser's effectiveness against any target. Some of these factors are associated with the laser itself, such as the strength or power of the beam. Others are related to the atmosphere, or the medium through which the beam must pass on its way to the target. These include diffusion, or spreading of the beam, as well as any other condition which alters or distorts the beam in any way. Finally, there is the target itself. Its composition and sensitivity to the energy de-

posited by the laser on the target will affect the laser's ability to neutralize it. These factors will determine the necessary power, concentration, and duration of the laser's radiation of a particular spot on the target needed to disable it effectively. Add to all the above factors any evasive maneuvers performed by the target, and the development of a ground-based, anti-aircraft/anti-missile laser weapon system becomes a task replete with obstacles.

~~~~~

Two significant obstacles to the development of an effective laser weapon system are precision pointing of the laser and accurate tracking of the target. It is not sufficient simply to "paint" the entire surface of the target with laser energy, nor is it currently possible to have a laser powerful enough to achieve the instantaneous, spectacular destruction of the target portrayed in fictional depictions of future warfare. As stated above, practical limitations of the laser, the distortion of the beam as it passes through the atmosphere, and the nature of the target itself, all make it necessary for the laser energy to be concentrated on a specific spot on the target for some finite amount of time before the target is disabled.

#### 1.1 Background

1222

STRATE STRATES

ASSAGE RECORD PROVIDE AND AND A

The Air Force Weapons Laboratory at Kirtland AFB, New Mexico, is currently testing a high energy laser weapon system against airborne targets. These targets must be tracked despite the presence of several factors which can cause relative motion between the emitted beam and the target. These include: target motion, mirror vibration, atmospheric jitter, and sensor measurement errors (15:2).

Target measurements are supplied by a Forward Looking Infra-Red sensor (FLIR). This sensor was chosen because it allows the tracker to gather target information passively, thereby preventing the target from detecting that he is being tracked.

The current research effort at the Weapons Laboratory uses a correlation tracker to process sensor information. At any one time, the tracker takes predetermined or previously gathered FLIR data and compares it with new information gathered at the current time. Cross correlation between these two sets of data is used to generate the relative position offsets from one frame of data to the next. These offsets are used to control gimbals which position the FLIR so that the target is centered within the sensor's field of view. Since the laser has been coupled to the FLIR through a shared aperture, this centering of the target within the FLIR's field of view also points the laser toward the target.

Though this correlation technique can be applied to a wide variety of targets because it relies only upon measured target data, it is highly susceptible to noise and takes no advantage of any a priori knowledge about the target. In most practical applications, however, information about the target can be supplied or at least estimated. If this information could be incorporated into the tracking algo-

rithm, performance should be improved. Furthermore, the effects of atmospheric disturbances on radiated waveforms is well-known and can be described statistically. This will allow separation of true target motion from apparent motion caused by the distortion of the infrared wave while it travels from the target to the sensor. This separation is important since the laser beam will not undergo the same distortion as it travels toward the target. Finally, statistical information about the FLIR measurement noise and background noise is available and can be used to improve the estimate of the target's position further.

In recent years, considerable effort has been spent at the Air Force Institute of Technology, to demonstrate the feasibility and accuracy of a tracker for this application that uses an extended Kalman filter. This algorithm has been tested in simulations against both long range targets (12) and short range air-to-air missile targets (2,3,5,12,13,15,17). This algorithm results in a significant improvement in tracking capability over the currently used correlation tracker.

Specifically, the study by Mercier (12) demonstrated the feasibility of using the extended Kalman filter in an algorithm against long range targets whose FLIR image plane intensity shape function could be modelled as an Airy disc (well approximated as a bivariate Gaussian) pattern due to a point source. Also due to these long ranges, the exhibited target dynamics would be very benign.

The next study, by Harnly and Jensen (3), concluded that while estimates of target position and atmospheric jitter were sufficient to track very benign target trajectories, additional estimates of target velocities and accelerations were needed to enable tracking of more maneuverable targets. Target intensity images were no longer portrayed as having circular equal intensity contours, but were now described as being elliptical. Additional modifications were made to the algorithm to allow adaptive estimation of the true sizes and shapes of ellipses.

Karalalalalala istaisia istaisi si si si alalalala istaisi

Studies by Worsley (17) and Flynn (2) demonstrated that a constant turn rate target acceleration model was a more descriptive model of maneuvering airborne targets than either a Brownian motion or a first order Gauss-Markov target acceleration model. This model consistently produced less biased estimates than those produced by filters using the other acceleration models. However, the improvements in performance were not significant enough relative to the additional computational burden imposed by using the constant turn rate model, to warrant its use in all applications. Flynn's study (2) utilized the multiple model filter algorithm which was used in this research. The primary purpose of the research was to investigate if using a bank of independent Kalman filters (each tuned for optimum performance against a different target dynamics condition) and optimally combining the estimates produced by each filter, could produce a better estimate of target position than a

non-adaptive filter could. Combination of the state estimates is accomplished by weighting the estimates of each filter with a weighting factor based on the "correctness" of each filter dynamics model relative to true target dynamics. The major problem of Flynn's study was that, because of the similarity of the residuals from all the filters, the algorithm was consistently unable to identify the filter with the "best" model.

Studies by Singletery (15) and Rogers (14) implemented algorithms which made no claims as to prior knowledge about the size or shape of the target hot-spots. Digital signal processing techniques were used to take FLIR data and compute an estimated target shape. In addition, Rogers' thesis demonstrated the feasibility of this technique against multiple hot-spot targets performing very benign trajectories. Rogers also developed an alternative tracker form, one which used the estimated target shape as a template against which later measurement data could be compared. This comparison is performed by an enhanced correlation algorithm whose output is provided as "measurements" to a linear Kalman These "measurements" are the offset distances from filter. center of the template to the point of maximum correlation in terms of FLIR image plane coordinates. A linear Kalman filter, as opposed to the non-linear extended Kalman filter, could be used because the offset distances are linear functions of the chosen filter states.

Because both techniques demonstrated similar capability against multiple hot-spot targets, follow-on studies by Kozemchak (5) and Millner (13) further developed these configurations by testing them against more realistic target trajectories. These trajectories included benign constant velocity trajectories and more dynamic constant-g, pull-up manuevers. The targets were also allowed to perform rolling maneuvers to test the trackers against multiple hot-spot targets whose target intensity functions in the FLIR image plane were changing constantly.

While the trackers in both of the above studies performed well against the simulated target maneuvers, both trackers had difficulty maintaining lock on targets performing pull-up maneuvers in excess of 5 g's. It is for this reason that investigation of additional techniques for processing target information was performed.

## 1.2 Problem

The purpose of this research was to take the tracker formulations developed by Kozemchak (5) and Millner (13) and determine the feasibility of implementing a multiple model filter structure as a means of adaptively adjusting the aperture of the tracker field of view to permit the tracking of targets performing highly dynamic maneuvers. As with Flynn's thesis (2), the different filters were tuned to achieve optimal performance at different degrees of target dynamics.

The data processing algorithms for the tracker config rations will now be described. The first model is t extended Kalman filter tracker used by Kozemchak (5). T second model is the linear Kalman filter/correlator track used by Millner (13).

1.2.1 Extended Kalman Filter Tracker. Figure I illustrates the algorithm for the extended Kalman filt tracker implemented by Kozemchak (5). In this configurati the center of the FLIR field of view is positioned at t filter predicted target centroid location due to targ dynamics over the most recent sample period. Each frame FLIR data is arranged into a 64-dimensional measureme vector  $\underline{z}(t_i)$ , which is the input to the extended Kalm filter in the lower path of the figure. The extended Kalm filter uses the nonlinear and the linearized intensity fur tions ( $\underline{h}[\underline{x}(t_i^-), t_i]$  and  $\underline{H}[\underline{x}(t_i^-), t_i]$  respectively) to cc pute an updated estimate of the state variables,  $\underline{x}(t_i^-)$ from the measurement vector via the equation:

 $\underline{\hat{X}}(t_i^+) = \underline{\hat{X}}(t_i^-) + \underline{K}(t_i)[\underline{z}(t_i) - \underline{h}[\underline{\hat{X}}(t_i^-), t_i^-)] \quad (1-$ where  $\underline{\hat{X}}(t_i^+)$  = state estimate vector after measurement incorporation at time  $t_i$ 

- $\hat{\mathbf{X}}(t_i)$  = state estimate vector propagated from previous measurement update to time  $t_i$
- $\underline{K}(t_i) = Kalman filter gain$

$$\underline{z}(t_i) = \underline{h}[\underline{x}(t_i), t_i] + \underline{v}(t_i)$$



Ö

لمعتملته

Figure I-1. Data Processing Algorithm for the Extended Kalman Filter Tracker

71

- $\underline{h}[\underline{\hat{x}}(t_i), t_i] = \text{intensity shape function for measurements}$ at time  $t_i$  as a function of the state estimate
- $\underline{H}[\underline{x}(t_i^{-}), t_i^{-}] = \frac{\partial h}{\partial x} | \underbrace{x(t_i^{-})}_{\text{the Kalman filter gain } \underline{K}(t_i^{-})}$ 
  - v(t<sub>i</sub>) = measurement corruption noise vector which includes FLIR measurement and background noises

The Kalman filter will then propagate this updated state estimate to the next sample time,  $t_{i+1}$ , based on its internal target dynamics model. This information is passed to the controller which positions the FLIR so that the center of its field of view is once again pointing to the predicted target position. For this research it was assumed that the controller was capable of pointing to the filter indicated position in the time available between samples (1/30 second).

Returning to the upper path of Figure I-1, a Fourier Transform is taken of the FLIR measurement data so that subsequent operations can be performed in the frequency domain. The motivation for carrying out those calculations in the frequency domain will be covered in a later chapter. Unlike the  $\underline{z}(t_i)$  measurement vector formulated for the Kalman filter, the data is arranged in a 24 x 24 measurement array as opposed to an 8 x 8 array. This larger array was processed to reduce edge effects, aliasing, and leakage conditions involved when transforming finite sequences (15:18). In most engineering applications, the larger array is created by padding the original data with zeros. This is

a valid procedure as long as the image intensity is essentially zero at the edge of the 8 x 8 field of view. However, if this is not the case, padding with zeros will introduce artificial edge effects. In such cases, it is more appropriate to pad with data. Such padding is possible in this application because the field of view encompasses only a small portion of the entire FLIR measurement.

a de la companya de l

in the product of the frequence of the f

Generation of the nonlinear and linearized intensity functions requires interframe filtering to reduce the effects of noise. In order to perform this filtering, the target intensity profile must first be centered so that the noise can be averaged out. This centering of the target image is accomplished by multiplying the Fourier transform of the measurement data by a negating phase shift in the frequency domain. This negating phase shift is the complex conjugate of the linear phase shift corresponding to the (estimated) target image offset in the spatial domain. The causes of this offset are atmospheric jitter and imperfect propagation of the target dynamic states. This information is available from the updated state estimates of the extended Kalman filter.

Exponential smoothing is then performed on the centered data. Because the noise is expected to be changing far more rapidly than the target intensity pattern from one sample time to the next, this process will attenuate the noise by averaging the centered data of successive frames of data. The underlying target intensity function generated is used

as the reference image,  $\underline{h}[\underline{x}(t_i), t_i]$ . The smoothed data is then differentiated with respect to a change in the Kalman filter states by employing the derivative property of the Fourier Transform to provide the frequency domain representation of the linearized intensity function, H.

Because the nonlinear and linearized intensity functions will be used to update the Kalman filter estimates after the next measurement, both functions are evaluated at the state expected at the next sample time. This information is made available by propagating the updated Kalman filter states from the current measurement time to the next measurement time. Since it is assumed that the FLIR is centered on the position predicted due to estimated target dynamics, the intensity patterns are evaluated at the location of the predicted atmospheric states. Again, the shift theorem of the Fourier Transform is employed to perform the phase shift in the frequency domain. The inverse Fourier transform is then performed and  $h[\underline{x}(t_{i+1}^{-}), t_{i+1}]$  and  $\underline{H}[\underline{x}(t_{i+1}^{-}, t_{i+1}]$  are ready for the extended Kalman filter to use when processing the next frame of data.

1.2.2 Linear Kalman Filter/Correlator Tracker. The Linear Kalman Filter/Correlator Tracker was the tracker initially developed by Rogers (14) and later extended and implemented by Millner (13). As can be seen in Figure I-2, it is very similar in structure to the extended Kalman filter tracker described in Section 1.2.1. The derivation of the target reference image is accomplished as before.



0.5500

ACCOUNT ON AND A

الم روز م

Ö

· 31



Tr.

ł

The difference is that it is now used as the template of a correlator against which new data received from the sensor will be compared.

Correlation is performed in the frequency domain and the outputs of the correlator are the position offsets of the target centroid from the center of the reference image. Because these offsets are linear functions of the chosen filter states, a linear Kalman filter can be used in place of the extended Kalman filter operating on raw FLIR data, as in the previous section. This results in significant reduction in the number of operations required to process measurement information. In addition, because the "measurements" to the Kalman filter are only the position offsets, the measurement vector is now only a 2-dimensional vector.

Filter state propagation from one sample time to the next is accomplished as before, but now measurement updates for the linear Kalman filter are done using:

 $\underline{\hat{x}}(t_{i}^{\dagger}) = \underline{\hat{x}}(t_{i}^{-}) + \underline{K}(t_{i})[\underline{z}(t_{i}) - \underline{Hx}(t_{i}^{-})] \qquad (1-2)$ 

where  $\underline{X}(t_i^+)$  = state estimate vector after measurement incorporation at time  $t_i$ 

- $\hat{\mathbf{X}}(t_i)$  = state estimate vector propagated from previous measurement update to time  $t_i$ 
  - $K(t_i) = Kalman filter gain$
  - <u>z(ti)</u> = measurement vector of target centroid offsets from the center of the FLIR field of view, as generated by the correlator of Figure I-2
    - H = linear combination of the states which contribute to the respective measurements
Pointing of the sensor is accomplished as before under the assumption that the controller is capable of pointing to the filter indicated position within one sample time is as described in the previous section.

### 1.3 Overview

Chapters II, III, and IV describe in detail the mathematical models used in the computer simulation. More specifically, Chapter II presents the truth model, which was the environment from which measurements were taken. Chapter III describes the Kalman filters, extended and linear, that were used in the respective tracker configurations. Chapter IV discusses the multiple model filter algorithm; why it was chosen and how it was incorporated into the existing tracker structures.

Chapter V presents a performance analysis of the both tracker types, with the multiple model filter algorithm in place, against a variety of target trajectories. Chapter VI presents the conclusions and recommendations drawn from this research.

### 2.1 Introduction

The truth model is the representation of the real world implemented by the researcher. It may not be the most complete model available to him, but it should embody all important aspects of the problem and reproduce the real world environment with good fidelity. In this study, the following processes were included in the truth model: atmospheric jitter, target dynamics and shape effects, and background and FLIR noises. These processes are of importance to the tracking problem because they describe the motion of the true target in inertial space, and the distortion of the target intensity function as it passes through the atmosphere, as well as any background and FLIR measurement noises which combine to produce the observed target image.

This chapter describes the truth model used in this study. It includes discussions on the model for atmospheric jitter, the target trajectories used, and the development of the target measurement model.

### 2.2 Target Centroid Offset Model

During target tracking, there are a number of effects which can create apparent motion between the target and the sensor. These include: target dynamics, sensor boresight error, FLIR system vibrations, and atmospheric jitter. For this study of a ground-based laser system, it was assumed that dominant modes of apparent target motion are those associated with target dynamics and atmospheric disturbances. Therefore, a continuous-time target model which incorporates both target dynamics and atmospheric disturbances describes the apparent target motion.

The FLIR measurements are scalar quantities that represent the average intensity of the received image over each picture element (pixel). The tracking window used in this research consisted of an 8 x 8 array of pixels in the FLIR image plane. Although the measurements are passed to the extended Kalman filter as a 64-dimensional vector, target dynamics and atmospheric disturbances are described using an x-y coordinate frame in the two-dimensional FLIR image plane in units of pixels (where each pixel is 20  $\mu$ rads by 20  $\mu$ rads). Hence the x- and y- coordinates of the apparent target centroid in FLIR image plane coordinates are:

$$x_{C} = x_{D} + x_{A}$$
 (2-1a)  
 $Y_{C} = y_{D} + y_{A}$  (2-1b)

where  $x_C = x$ -coordinate of apparent target centroid  $y_C = y$ -coordinate of apparent target centroid  $x_D = x$ -offset due to target motion  $y_D = y$ -offset due to target motion  $x_A = x$ -offset due to atmospheric disturbances  $y_A = y$ -offset due to atmospheric disturbances

11-2

### 2.3 Atmospheric Disturbances

The target's intensity function will undergo translational motion on the FLIR image plane due to atmospheric disturbances which cause phase distortions in the radiated wavefronts from the target as they propagate through the atmosphere (14:27). The model used in this study was developed by The Analytical Sciences Corporation (16) and data analysis by Hogge and Butts (4). The power spectral density of this phenomenon in each of the two FLIR plane directions can be well approximated as the output of a third-order linear shaping filter driven by unit-strength, zero-mean, white Gaussian noise (12:12).



where w<sub>A</sub> = unit-strength, zero-mean, white Gaussian noise

- K = system gain
- a = break frequency, 14.14 rad/sec
- b = break frequency, 659.5 rad/sec
- $x_{\lambda}$  = output of the shaping filter

By adjusting the value of the system gain, K, the desired root-mean-squared (rms) atmospheric characteristic can be obtained (12). The effects of atmospheric jitter are assumed independent of the direction on the FLIR image plane, so two independent shaping filters of the above form can be used to model jitter; one for each coordinate direction of the FLIR image plane.

The developed mathematical model for atmospheric jitter will now be expressed in state space notation. Atmospheric jitter can be expressed in the time-invariant stochastic differential equation of the form:

 $\underline{\dot{x}}_{a}(t) = \underline{F}_{a}\underline{x}_{a}(t) + \underline{G}_{a}\underline{w}_{a}(t) \qquad (2-2)$ 

where  $\underline{F}_a$  = atmospheric plant matrix

- x<sub>a</sub>(t) = six atmospheric noise states
  - G<sub>a</sub> = atmospheric noise input matrix
- wa(t) = two-dimensional vector of white Gaussian noise inputs with statistics:

$$E\{\underline{w}_{a}(t)\} = \underline{0}$$
$$E\{\underline{w}_{a}(t)\underline{w}_{a}^{T}(t+\tau)\} = \underline{I}\delta(\tau)$$

The shift of the intensity funciton due to atmospheric jitter can be expressed with:

 $\underline{x}_{A}(t) = \underline{H}_{a}\underline{x}_{a}(t) \qquad (2-3)$ where  $\underline{x}_{A}(t)$  = shift in FLIR coordinates, with components  $x_{A}$  and  $y_{A}$  as in (2-1)

<u>H</u><sub>a</sub> = system output matrix

Because of the independence between the disturbances occurring in the horizontal and vertical directions of the FLIR image plane, they can be decoupled and separate models can be developed. In Jordan canonical form, the disturbances in the x- (horizontal) direction becomes (12:73-75):

$$\underbrace{\overset{\bullet}{x}}_{ax}(t) = \begin{bmatrix} -a & 0 & 0 \\ 0 & -b & 1 \\ 0 & 0 & -b \end{bmatrix} \underbrace{\underline{x}}_{ax}(t) + \begin{bmatrix} G1 \\ G2 \\ G3 \end{bmatrix} \underbrace{\underline{w}}_{ax}(t) \quad (2-4)$$

where a,b are the break frequencies described earlier and:

G1 =  $Kab^2/(a-b)^2$ G2 = -G1 G3 = (a-b)G1

The output equation becomes:

$$x_{Ax}(t) = [1 \ 1 \ 0] \underline{x}_{Ax}(t)$$
 (2-5)

The solution to the state differential equation (2-2) over one sample interval assumes the form:

$$\underline{\mathbf{x}}_{a}(t_{i+1}) = \underline{\Phi}_{a}(t_{i+1}, t_{i}) \underline{\mathbf{x}}_{a}(t_{i}) + \int_{t_{i}}^{t_{i+1}} \underline{\Phi}_{a}(t_{i+1}, \tau) \underline{G}_{a}(\tau) \underline{\mathbf{w}}_{a}(\tau) d\tau \qquad (2-6)$$

where  $\underline{\Phi}_a$  = the state transition matrix which is the solution to the matrix differential equation:

 $\underline{\Phi}_{a}(t,t_{i}) = \underline{F}_{a}\underline{\Phi}_{a}(t,t_{i})$ 

and the initial condition:

 $\Phi_a(t_i, t_i) = I$  (the identity matrix)

Since  $\underline{F}_a$  is time-invariant, the state transition matrix is solely a function of the sampling time,  $\Delta t = t_{i+1}-t_i$ . Therefore, for a constant sampling time,  $\Delta t$ , the statetransition matrix is itself a constant. Thus, for the system in (2-4) the state transition matrix for any time, t<sub>i</sub>, is:

$$\underline{\Phi}_{a}(t_{i+1},t_{i}) = \begin{bmatrix} \exp(-a\Delta t) & 0 & 0\\ 0 & \exp(-b\Delta t) & \Delta t \exp(-b\Delta t) \\ 0 & 0 & \exp(-b\Delta t) \end{bmatrix} (2-7)$$

As before, the distortion in the y- (vertical) direction can be represented with a model of the same form.

Since it was desired to use a digital computer program to test the developed algorithms, an equivalent discretetime system model of the continuous-time system was developed (6:42):

$$\underline{\mathbf{x}}_{\mathbf{a}}(\mathbf{t}_{i+1}) = \underline{\Phi}_{\mathbf{a}}(\mathbf{t}_{i+1}, \mathbf{t}_{i}) \underline{\mathbf{x}}_{\mathbf{a}}(\mathbf{t}_{i}) + \underline{\mathbf{w}}_{\mathbf{ad}}(\mathbf{t}_{i}) \quad (2-8)$$

where  $\underline{w}_{ad}(\cdot)$  is a discrete-time white Gaussian noise process with the identical statistics as the integral term of Equation (2-6) for all time:

 $E\{\underline{w}_{ad}(t_{i})\} = \underline{0} \qquad (2-9a)$   $E\{\underline{w}_{ad}(t_{i})\underline{w}_{ad}T(t_{i})\} = \underline{0}_{ad}(t_{i})$   $= \int_{t_{i}}^{t_{i+1}} \underline{\Phi}_{a}(t_{i+1},\tau)\underline{G}_{a}\underline{I}\underline{G}_{a}^{T}\underline{\Phi}_{a}^{T}(t_{i+1},\tau)d\tau \qquad (2-9b)$   $E\{\underline{w}_{ad}(t_{i})\underline{w}_{ad}^{T}(t_{i})\} = \underline{0} \quad (\text{for } t_{i} \neq t_{i}) \qquad (2-9c)$ 

Therefore the equivalent discrete-time system model can be expressed as (12:15):

$$\underline{\mathbf{x}}_{\mathbf{a}}(\mathtt{t}_{i+1}) = \underline{\Phi}_{\mathbf{a}}(\mathtt{t}_{i+1}, \mathtt{t}_{i}) \underline{\mathbf{x}}_{\mathbf{a}}(\mathtt{t}_{i}) + \sqrt[c]{\underline{Q}_{ad}} \underline{\mathbf{w}}_{\mathbf{n}}(\mathtt{t}_{i}) \quad (2-10)$$

where  $\sqrt[6]{Q_{ad}}$  is the lower Cholesky square root of  $\underline{Q}_{ad}$  (6:3 71) and:

$$E\{\underline{w}_n(t_i)\} = \underline{0}$$
 (2-1)

$$E\{\underline{w}_{n}(t_{i})\underline{w}_{n}^{T}(t_{j})\} = \underline{I}$$
 (2-1)

$$E\{\underline{w}_{n}(t_{i})\underline{w}_{n}^{T}(t_{j})\} = \underline{0} \text{ (for } t_{i} \neq t_{j}) \text{ (2-1)}$$

# 2.4 Target Dynamics Models

The model for target dynamics is a continuous-ti deterministic model which describes a highly maneuveral aircraft or missile. In order to test the algorithm in realistic environment as possible, a number of maneuv were generated (5:35):

(1) straight line propagation

(2) constant roll-rate maneuvers

(3) constant G, constant speed turns

These manuevers will be tested individually to det mine the algorithm's ability to follow a target perform highly dynamic maneuvers. Later these same maneuvers w be performed sequentially and/or simultaneously by the t get so the algorithm can be tested against a realis dynamic target whose intensity function is constanchanging.

Two previous studies, by Kozemchak (5) and Mill (13) respectively, assumed that the centroid of the tar intensity pattern coincides with the center of gravity the target. Furthermore, the center of gravity of target was assumed to be on the roll axis of the target, rolling maneuvers performed by a non-pitching aircraft have no effect on centroid dynamics. Conversely, a pull-up maneuver does affect centroid dynamics for multiple hot-spot targets (5:35,13:22).

As developed by Harnly and Jensen (3), the true target location in the two-dimensional FLIR image plane can be expressed in azimuth,  $\alpha(t)$ , and elevation,  $\beta(t)$ . This true location can be compared with the filter estimates to evaluate filter performance. As with Kozemchak (5) and Millner (13), the azimuth and elevation rate inputs are used for propagating the true position as well as determining the accuracy of the filter's acceleration estimate. Thus, the time history of the target location can be generated via:

$$\underbrace{\overset{\bullet}{\underline{x}}}_{D} = \underline{\underline{u}}(t) = \begin{bmatrix} \overset{\bullet}{\alpha}(t) \\ \overset{\bullet}{\beta}(t) \end{bmatrix}$$
(2-12)

where  $\dot{\alpha}(t)$  and  $\dot{\beta}(t)$  are the time varying azimuth and elevation velocities in inertial space. While it would have been easier to input time histories for  $\alpha(t)$  and  $\beta(t)$  directly for this deterministic case, the above form was used so a stochastic model could be readily implemented if desired at a later time (13:15).

At this point Kozemchak (5) and Millner (13) differed in their respective approximations of the dynamic offset. Kozemchak used a second-order model which assumed constant acceleration over each sample period:

$$\underline{\mathbf{x}}_{D}(t_{i+1}) = \underline{\mathbf{x}}_{D}(t_{i}) + \underline{\mathbf{x}}_{D}(t_{i})\Delta t + 0.5\underline{\mathbf{x}}_{D}(t_{i})\Delta t^{2} \qquad (2-13)$$

Millner used a first order model which used the values of the azimuth and elevation velocities which correspond to the midpoint of the sampling interval:

$$\underline{\mathbf{x}}_{\mathrm{D}}(\mathtt{t}_{i+1}) = \underline{\mathbf{x}}_{\mathrm{D}}(\mathtt{t}_{i}) + \underline{\underline{\mathbf{x}}}_{\mathrm{D}}(\mathtt{t}_{i}+\Delta\mathtt{t}/2)\Delta\mathtt{t}$$
(2-14)

The approximations of the dynamic offset were kept intact for each of the respective filters. This means that the extended Kalman filter tracker and the linear Kalman filter tracker used different truth models. If the assumptions made in the derivation of each model are correct (i.e. constant acceleration over each sample period for the extended Kalman filter tracker), then there is very little difference between the two models for a small sample period. However, the Millner form is a more correct model because it uses only those terms that are independently specifiable. The acceleration term in (2-13) is derived from the change in elevation and azimuth velocities,  $\dot{\alpha}(t)$  and  $\dot{\beta}(t)$  respectively, over a given sample period.

The azimuth and elevation velocities are initially calculated in the inertial frame and must then be projected into the FLIR image plane. The relationship between these two frames of reference is shown in Figure II-1 (5:37).

11-9



Figure II-1. Inertial Coordinate Frame

where  $\underline{x}_{I}$ ,  $\underline{y}_{I}$ ,  $\underline{z}_{I}$  = inertial axes  $\rho$  = range to target  $\underline{v}_{I}$  = target inertial velocity  $r_{h}$  = horizontal range  $\alpha$  = azimuth angular displacement  $\beta$  = elevation angular displacement

The geometry associated with azimuth direction is shown in Figure II-2 (13:19).

Ø





Thus, the displacement angle,  $\alpha(t)$ , is defined as:

$$\alpha(t) = tan^{-1}[z_{I}(t)/x_{I}(t)]$$
 (rad) (2-15)

and

$$\dot{a}(t) = [x_{I}(t)\dot{z}_{I}(t) - z_{I}(t)\dot{x}_{I}(t)] / [z_{I}^{2}(t) + x_{I}^{2}(t)]$$
  
(rad/sec) (2-16)

and

$$\ddot{a}(t) = \{ [x_{I}(t)\ddot{z}_{I}(t) - \ddot{x}_{I}(t)z_{I}(t)]r_{h}^{2} - 2[x_{I}(t)\dot{x}_{I}(t) + z_{I}(t)\dot{z}_{I}(t)] [x_{I}(t)\dot{z}_{I}(t) - \dot{x}_{I}(t)z_{I}(t)] \} / r_{h}^{4}$$
(rad/sec ) (2-17)

where  $r_h$  is determined by  $r_h = (x_I^2 + z_I^2)^{1/2}$ . To convert these values into FLIR image plane units, pixels/sec, divide the number of radians by 20 x 10<sup>-6</sup>. This conversion factor is derived from the fact that each pixel represents a region 20 µrads x 20 µrads (3:33).

Similarly, Figure II-3 displays the geometry involved in calculating elevation, elevation velocity, and elevation acceleration (13:20).





where 
$$\rho = (x_I^2 + y_I^2 + z_I^2)^{1/2}$$
. Therefore,  
 $\beta(t) = tan^{-1}[y_I(t)/r_h(t)]$  (rad) (2-18)

and

$$\dot{\beta}(t) = [r_{h}(t)\dot{y}_{I}(t) - y_{I}(t)\dot{r}_{h}(t)] / \rho^{2}(t) \text{ (rad/sec) (2-19)}$$
where  $\dot{r}_{h}(t) = [\dot{x}_{I}(t)x_{I}(t) + \dot{z}_{I}(t)z_{I}(t)] / r_{h}(t)$ 
(m/sec) (2-20)

and

$$\dot{\mathbf{y}}(t) = \{ \boldsymbol{\rho}^{2}(t) [\mathbf{r}_{h}(t) \ddot{\mathbf{y}}_{I}(t) - \mathbf{y}_{I}(t) \ddot{\mathbf{r}}_{h}(t) ] \\ - [\mathbf{r}_{h}(t) \dot{\mathbf{y}}_{I}(t) - \mathbf{y}_{I}(t) \dot{\mathbf{r}}_{h}(t) ] [2 \dot{\boldsymbol{\rho}}(t) \boldsymbol{\rho}(t) ] \} / \boldsymbol{\rho}^{4}(t) \\ (rad/sec) \qquad (2-21)$$

where 
$$\ddot{r}_{h}(t) = \{ [x_{I}(t)\ddot{x}_{I}(t) + \dot{x}_{I}^{2}(t) + z_{I}(t)\ddot{z}_{I}(t) + \dot{z}_{I}^{2}(t) ] \\ x r_{h}(t) - \dot{r}_{h}(t) [x_{I}(t)\dot{x}_{I}(t) + z_{I}(t)\dot{z}_{I}(t) ] \} \\ / r_{h}^{2}(t) (m/sec^{2})$$
(2-22)

$$\dot{\rho}(t) = [x_{I}(t)\dot{x}_{I}(t) + y_{I}(t)\dot{y}_{I}(t) + z_{I}(t)\dot{z}_{I}(t)]$$
  
/  $\rho(t)$  (m/sec) (2-23)

The values for  $\beta(t)$ ,  $\dot{\beta}(t)$ , and  $\ddot{\beta}(t)$  can be converted to FLIR image plane units by the same conversion factor procedure used for the azimuth terms.

The solutions to the dynamics differential equation (2-12) have the form:

$$\underline{\mathbf{x}}_{D}(t_{i+1}) = \underline{\Phi}_{D}(t_{i+1}, t_{i}) \underline{\mathbf{x}}_{D}(t_{i})$$

$$+ \int_{t_{i}}^{t_{i+1}} \underline{\Phi}_{D}(t_{i+1}, \tau) \underline{B}_{D}(\tau) \underline{u}_{D}(\tau) d\tau \qquad (2-24)$$

where 
$$\underline{\Phi}_{D}(t,\tau)$$
 = state transition matrix for vehicle  
dynamics  
$$\underline{B}_{D}(\tau) = \text{control} \text{ input matrix}$$
$$\underline{u}_{D}(\tau) = \text{control function for the truth model}$$
as defined in equation (2-12)

For digital computer implementation, accelerations approximated as piecewise constant between sampling times result in a piecewise linear function for  $\underline{u}_{D}(t)$ . Therefore Equation (2-24) can be expressed in discrete-time form as:

$$\underline{\mathbf{x}}_{\mathrm{D}}(\mathtt{t}_{i+1}) = \underline{\Phi}_{\mathrm{D}}(\mathtt{t}_{i+1}, \mathtt{t}_{i}) \underline{\mathbf{x}}_{\mathrm{D}}(\mathtt{t}_{i}) + \underline{B}_{\mathrm{d}}(\mathtt{t}_{i}) \underline{\mathbf{u}}_{\mathrm{d}}(\mathtt{t}_{i}) \quad (2-24)$$

In Kozemchak's thesis he defined  $\underline{B}_d(t_i)\underline{u}_d(t_i)$  as:

$$\underline{B}_{d}(t_{i})\underline{u}_{d}(t_{i}) = \begin{bmatrix} \Delta t & 0 & 0.5\Delta t^{2} & 0 \\ 0 & \Delta t & 0 & 0.5\Delta t^{2} \end{bmatrix} \begin{bmatrix} \dot{\alpha}(t_{i}) \\ \dot{\beta}(t_{i}) \\ \ddot{\alpha}(t_{i}) \\ \ddot{\beta}(t_{i}) \\ \ddot{\beta}(t_{i}) \end{bmatrix}$$
(2-26)

where, of course,  $\dot{\alpha}$  and  $\ddot{\alpha}$  are as given above and are not independently specifiable, and similarly for  $\dot{\beta}$  and  $\ddot{\beta}$ .

Millner defined the quantity somewhat differently, with (more appropriately) independently specifiable components only:

$$\underline{\underline{B}}_{d}(t_{i})\underline{\underline{u}}_{d}(t_{i}) = \begin{bmatrix} \Delta t & 0 \\ 0 & \Delta t \end{bmatrix} \begin{bmatrix} \alpha(t_{i} + \Delta t/2) \\ \beta(t_{i} + \Delta t/2) \end{bmatrix}$$
(2-27)

# 2.5 Overall State Space Model

Combining the truth models for target dynamics and atmospheric distortions yields a single eight state discrete-time model which consisting of two dynamics states and six atmospheric states.

$$\underline{\mathbf{x}}_{\mathbf{T}}(\mathbf{t}_{i+1}) = \underline{\Phi}_{\mathbf{T}}(\mathbf{t}_{i+1},\mathbf{t}_{i})\underline{\mathbf{x}}_{\mathbf{T}}(\mathbf{t}_{i}) + \underline{B}_{d}\underline{\mathbf{u}}_{d}(\mathbf{t}_{i}) + \underline{G}_{d}\underline{\mathbf{w}}_{d}(\mathbf{t}_{i})$$
(2-28)  
where  $\underline{\Phi}_{\mathbf{T}}(\mathbf{t}_{i+1},\mathbf{t}_{i}) = \underline{\Phi}_{\mathbf{T}}(\mathbf{t}_{i+1}-\mathbf{t}_{i}) = \underline{\Phi}_{\mathbf{T}}(\Delta \mathbf{t}) =$ 

| 1 | 0 | 0         | 0     | 0                   | 0                 | 0      | 0                         |
|---|---|-----------|-------|---------------------|-------------------|--------|---------------------------|
| 0 | 1 | 0         | 0     | 0                   | 0                 | 0      | 0                         |
| 0 | 0 | exp(-a∆t) | 0     | 0                   | 0                 | 0      | 0                         |
| 0 | 0 | 0 exp     | (-b∆t | $\Delta texp(\cdot$ | -b∆t) 0           | 0      | 0                         |
| 0 | 0 | 0         | 0     | exp(-b              | Δt) 0             | 0      | 0                         |
| 0 | 0 | 0         | 0     | 0                   | $exp(-a\Delta t)$ | ) ()   | 0                         |
| 0 | 0 | 0         | 0     | 0                   | 0 exp             | o(−b∆t | $\Delta texp(-b\Delta t)$ |
| 0 | 0 | 0         | 0     | 0                   | 0 -               | 0      | $exp(-b\Delta t)$         |
|   |   |           |       |                     |                   |        | _                         |

$$\underline{\mathbf{x}}_{D}(t_{i}) = [\mathbf{x}_{D}(t_{i}) \ \mathbf{y}_{D}(t_{i}) \ \mathbf{x}_{1A}(t_{i}) \ \mathbf{x}_{2A}(t_{i}) \ \mathbf{x}_{3A}(t_{i}) \ \mathbf{y}_{1A}(t_{i}) \mathbf{y}_{2A}(t_{i}) \ \mathbf{y}_{3A}(t_{i})]^{T}$$

For Kozemchak's study:

O

| $\underline{B}_{d}(t_{i}) =$ | Δt | 0  | $0.5\Delta t^2$ | 0               |
|------------------------------|----|----|-----------------|-----------------|
| - u +                        | 0  | Δt | 0               | $0.5\Delta t^2$ |
|                              | 0  | 0  | 0               | 0               |
|                              | 0  | 0  | 0               | 0               |
|                              | 0  | 0  | 0               | 0               |
|                              | 0  | 0  | 0               | 0               |
|                              | 0  | 0  | 0               | 0               |
|                              | 0  | 0  | 0               | 0               |

$$\underline{u}_{d}(t_{i}) = [\alpha(t_{i}) \ \beta(t_{i}) \ \alpha(t_{i}) \ \beta(t_{i})]^{T}$$

For Millner's study:

$$\underline{B}_{d}(t_{i}) = \begin{bmatrix} \Delta t & 0 \\ 0 & \Delta t \\ 0 & 0 \\ 0 & 0 \\ 0 & 0 \\ 0 & 0 \\ 0 & 0 \\ 0 & 0 \end{bmatrix}$$

 $\underline{\mathbf{u}}_{\mathbf{d}}(\mathbf{t}_{\mathbf{i}}) = \begin{bmatrix} \dot{\alpha}(\mathbf{t}_{\mathbf{i}} + \Delta \mathbf{t}/2) & \dot{\beta}(\mathbf{t}_{\mathbf{i}} + \Delta \mathbf{t}/2) \end{bmatrix}^{\mathrm{T}}$ 

والمراجع وال

### 2.6 Target Coordinate Frames

In order to describe the dynamics of various parts of the target relative to its center of mass, additional coordinate frames needed to be defined. This was necessary because a realistic image on the FLIR plane must be generated for instances when a multiple hot-spot target performs various maneuvers.

<u>Target frame</u> - This frame has its origin at the target center of mass. One of its axes is the inertial velocity vector of the target. Another axis is defined as being out the right side of the aircraft perpendicular to the velocity vector and in the plane of the radiating sources of the target. For example, for an aircraft with wing mounted engines, if the engines hang from pods, then the radiating sources are not in the same plane as the target center of mass. This model approximates many multi-engined aircraft. The third axis is defined by a vector perpendicular to the plane formed by the previous two vectors. This coordinate frame will be expressed with unit vectors  $\underline{e}_{v}$ ,  $\underline{e}_{pv}$ , and  $\underline{e}_{ppv}$ , respectively.

 $\alpha-\beta$  plane - This frame originates at the target center of gravity and is perpendicular to the true line of sight from the tracker (located at the origin of the inertial coordinate system) to the target. This plane can be shown to be defined by unit vectors  $\underline{e}_{\alpha}$  and  $\underline{e}_{\beta}$  which are misaligned from the inertial frame by the angles  $\alpha$  and  $\beta$  which were defined in Figure II-1. The third basis vector is aligned

along the line of sight to the target.

### 2.7 Target Trajectories

As mentioned previously, a number of deterministic target trajectories which incorporate a number of maneuvers were developed to provide as realistic targets for the tracking algorithm as possible. The basic trajectories used in this study are those used by Millner (13).

<u>Trajectory 1</u> - This trajectory is depicted in Figure II-4 and is a benign trajectory in which the target flies a constant-heading, straight-and-level course, either wings-level or performing a constant roll-rate maneuver.



Figure II-4. Trajectory 1

The inertial velocity,  $\underline{v}_{I}$ , for this maneuver is held constant and is parallel to the  $x_{I}-z_{I}$  plane. Roll maneuvers are performed with clockwise rotation of the target as seen from behind.

<u>Trajectory 2</u> - To evaluate filter performance against a more dynamic maneuver, a constant G pull-up maneuver was simulated. In this model the target begins with the same initial conditions as in the previous model. The maneuver is initiated at some pre-determined time after the simulation begins to allow the tracker to obtain good target position estimates before the maneuver begins. This trajectory can be seen in Figure II-5.



Figure II-5. Trajectory 2

It should be pointed out that this pull-up maneuver was started with a step change to the pitch-up rate, which is not a realistic model. However, this represents a more severe maneuver than any found in the real world, so the tracker should perform better against a more realistic target.

<u>Trajectory 3</u> - This trajectory was used to evaluate performance with a target that begins and terminates a maneuver during a simulation. As with trajectory 2, a constant G pull-up maneuver is executed but instead of continuing the maneuver to the end of the simulation, it is terminated (again with a step change in the pitch-up rate) at some earlier time so the target returns to straight-line propagation. The target assumes whatever inertial velocity it possessed at the termination of the pull-up maneuver. This velocity remains constant for the remainder of the simulation.

<u>Trajectory 4</u> - This trajectory was used to provide a target which displays motion in all inertial directions. As was true for trajectory 3, this trajectory is similar to trajectory 2, but instead of terminating the maneuver as done in trajectory 3, the target turns toward the FLIR plane. This out-of-plane change maneuver causes the projected target image on the FLIR image plane to change its appearance more dramatically than in the previous cases, with substantial changes in the separation between the individual hot-spots.

# 2.7 <u>Non-realistic Trajectories</u> and <u>Intensity Pattern</u> <u>Time</u> <u>Variations</u>

Other areas of interest include non-realistic trajectories and intensity pattern variations. The desire here is not to portray realistic targets, but to determine the algorithm's sensitivity to various parameter changes. This can include investigation of performance against targets executing maneuvers beyond the capability of current aircraft, such as instantaneous and dramatic heading changes. Such a test could be useful in determining the tracker's ability to re-acquire a target that has shifted out of its field of view. Other parameter changes that can be investigated include: varying hot-spot size, intensity, separation between hot-spots, and variation of these parameters with time.

### 2.8 <u>Measurement Model</u>

The measurements provided to the tracker algorithm represent the intensity function generated by the target projected onto the FLIR image plane. This function is also corrupted by any background and FLIR noises that may be present. For distant targets, it was found that these target patterns could be modelled with a bivariate Gaussian function with circular constant-intensity contours (12). However, close range targets were found to be better modeled with similarly distributed contours of elliptical shape (3). As the target gets closer to the tracker positon, individual and separate hot-spots can be identified on the target, each modelled with the following intensity function:

$$I = I_{max}exp[-0.5[(x-x_{peak}) (y-y_{peak})][\underline{P}]^{-1}$$
  
x [(x-x\_{peak}) (y-y\_{peak})]<sup>T</sup> (2-30)

where I<sub>max</sub> = maximum intensity of the hot spot

xpeak' ypeak = coordinates of the peak intensity of the hot-spot

 $\underline{P}$  = matrix whose eigenvalues are  $\sigma_v^2$  and  $\sigma_{pv}^2$ , which corresponds to the dispersion of the elliptical contour in the target plane defined earlier, and whose eigenvectors dictate the angular orientation of the principal axes of these ellipses

The x- and y- coordinates for Equation (2-30) are calculated relative to the center of the tracker field of view.

For single hot-spot targets the centroid of the intensity function is assumed to coincide with the target center of mass. For multiple hot-spot targets the hot-spots are distributed, for example, as shown in Figure II-6 (13:40)



Figure II-6. Distribution of Hot-Spots

The multiple hot-spot case requires that the distance each hot-spot from the target center of mass be kno Also, it is assumed that the target side slip angle a angle of attack are zero and that all the major axes of ellipsoids are parallel and aligned along the velocity v tor of the target, which extends out the nose of t vehicle.

For both the single and multiple hot-spot cases, 4 measurements are generated by taking the average intens of each pixel in an 8 x 8 pixel tracking window, which due to the combined effects of each of the hot-spots and corruptive background and FLIR noises:

$$z_{k1}(t_i) = \sum_{m=1}^{M} [1/A_p \int_{klth} I_m(x,y,x_{peakm}(t_i),y_{peakm}(t_i))dxdy]$$
  
pixel  
+  $y_{k1}(t_i)$ 

(2-

II-21

- - z<sub>kl</sub>(t<sub>i</sub>) = output of the klth pixel at time t<sub>i</sub>; the average intensity over that pixel as sensed by a detector in the FLIR image plane
    - $A_{p}$  = area of one pixel

- xpeakm(t<sub>i</sub>), ypeakm(t<sub>i</sub>) = location of the peak of the mth intensity function at time t<sub>i</sub>

### 2.9 Target Image

It was assumed that the major axes of the m hot-spots are all aligned parallel to the inertial velocity vector. Furthermore, by assumption, all m hot-spots lie in the plane formed by the wings of the target.

As discussed previously,  $\underline{v}_{I}$ , the inertial velocity vector of the target is assumed to be projected out the nose of the target. The  $\alpha - \beta$  plane is perpendicular to the true line of sight from the tracker to the target as defined by the basis vector  $\underline{e}_{r}$ . The origin of the target coordinate frame, the  $\alpha - \beta$  plane, is the target center of mass. Figure II-7 illustrates the geometry involved.



Figure II-7. Image Projection

From the figure it can be seen that

$$\cos\theta = \dot{\alpha}(t) / [\underline{v} | \text{LOS}] \qquad (2-32)$$

 $\sin\theta = \dot{\beta}(t) / [\underline{v} \perp \text{LOS}] \qquad (2-33)$ 

where  $(\underline{v} \perp LOS)$  is the magnitude of the velocity perpendicular to the tracker line of sight, i.e., the projection of  $\underline{v}_{I}$  onto the  $\alpha$ - $\beta$  plane, defined by  $[\underline{v} \perp LOS] = [\dot{\alpha}(t)^{2} + \dot{\beta}(t)^{2}]^{1/2}$ .

The size of the target and the distances between the hot-spots are fixed and do not vary with time. On the other hand, the image of the target as seen by the sensor will change as the target moves closer or farther away from the tracker or changes its orientation relative to the FLIR. This image at any time during the simulation is expressed in terms of some previously defined image, which has been specified at an initial range and size, and the current target range and velocity. The reference image is defined with the target lying flat in a plane perpendicular to the line of sight to the target. This produces an image with the largest possible size. For example, if the reference image was flat and circular in shape, any other orientation will project an elliptical image of much smaller area onto the FLIR image plane. The following expressions relate the current image size with respect to the reference image:

and a fait of the second

PRODUCT ADDRESS BARRAR

$$\sigma_{pv} = \sigma_{pvo}\rho_{o}/\rho \qquad (2-34)$$

$$\sigma_{v} = (\rho_{o}/\rho)[\sigma_{pvo} + \cos \{ (\sigma_{vo}-\sigma_{pv})]$$

$$= \sigma_{pv}[1+[(\underline{v} \perp LOS)/\underline{v}_{I}][AR-1]] \qquad (2-35)$$
where  $\sigma_{vo}, \sigma_{pvo}$  = the disperison of the target along the major and minor axes of the radiating ellipsoid, i.e., axes along and perpendicular to the velocity vector, respectively, for the reference image
$$\sigma_{v}, \sigma_{pv} = \text{the current dispersions of the target image}$$

$$\rho_{o} = \text{reference range from sensor to the target}$$

$$\rho = \text{current range from sensor to the target}$$

$$\underline{v}_{I} = \text{inertial velocity vector}$$

$$(\underline{v} \perp LOS) = \text{projection of } \underline{v}_{I} \text{ onto the } \alpha - \beta \text{ plane, the plane perpendicular to the line of sight to the target}$$

- vector,  $\underline{v}_{I}$ , and the  $\alpha \beta$  plane, as shown in Figure II-7
- AR =  $\sigma_{vo}/\sigma_{pvo}$  = maximum aspect ratio of the hot-spot reference image

As stated at the beginning of this section, the objective is to define the target image in terms of FLIR image plane coordinates. The relative distance between the hotspots is known in the target frame defined in Section 2.6.

The coordinates of the centers of the hot-spots must be converted to  $\alpha - \beta$  plane coordinates for use in (2-30) and (2-31). The transformation from the target frame coordinates to  $\alpha - \beta$  coordinates is accomplished via:

$$\underline{\mathbf{x}}_{\alpha\beta} = \begin{bmatrix} \mathbf{x} \\ \mathbf{y} \end{bmatrix}_{\alpha\beta} = \begin{bmatrix} \cos\theta & -\sin\theta \\ \sin\theta & \cos\theta \end{bmatrix} \begin{bmatrix} \mathbf{x} \\ \mathbf{y} \end{bmatrix} \text{ target frame } = \underline{\mathbf{A}} \ \underline{\mathbf{x}}$$
(2-36)

The dispersion matrix is transformed using:

$$\underline{\mathbf{P}}_{\boldsymbol{\alpha}\boldsymbol{\beta}} = \underline{\mathbf{A}} \ \underline{\mathbf{P}} \ \underline{\mathbf{A}}^{\mathrm{T}} \tag{2-37}$$

Since it is desired to have the inverse of  $\underline{P}_{\alpha\beta}$  for use in (2-30), a more convenient yet equivalent transformation can be generated by inverting this expression and using the orthogonal nature of A to yield

$$\underline{\mathbf{P}}^{-1} = \underline{\mathbf{A}} \left( \underline{\mathbf{P}}^{-1} \right) \underline{\mathbf{A}}^{\mathrm{T}}$$
(2-38)

### 2.10 Spatially Correlated Background Noise

The noise term,  $v_{kl}(t_i)$ , in Equation (2-31) associated with real data was found by Harnly and Jensen (3:19) to contain spatial correlation of the background noise with a correlation distance of about 2 pixels; this was modelled by maintaining non-zero correlation between each pixel and its two closest neighbors symmetrically in all directions. The 64 measurements (8 x 8 pixel array tracking window) are arranged as a 64-dimensional vector and so the spatially correlated noise can be modeled with:

$$\underline{\mathbf{v}}(t_i) = \sqrt[\mathbf{c}]{\underline{\mathbf{R}}} \ \underline{\mathbf{v}}'(t_i) \tag{2-39}$$

$$E\{\underline{v}'(t_{i})\} = \underline{0}$$
$$E\{\underline{v}'(t_{i})\underline{v}'^{T}(t_{j})\} = \underline{I}\delta_{ij}$$

The resulting noise process,  $\underline{v}(t_i)$ , has strength,  $E\{\underline{v}(t_i)\underline{v}^T(t_j)\} = \underline{R}\delta_{ij}$ , where <u>R</u> is the 64 x 64 matrix which describes the spatial correlation between pixels and is discussed in detail in the studies by Harnly and Jensen (3) and Kozemchak (5).

In order to generate the spatially correlated noise, the pixel numbering scheme illustrated in Figure II-8 was adopted. This array, which corresponds to the 8 x 8 FLIR data array, was numbered from 1 to 64 starting from the upper left hand corner of the array and proceeding across the rows as done by Harnly and Jensen (3:19).

| 1  | 2  | 3  | 4  | 5  | 6  | 7  | 8  |  |
|----|----|----|----|----|----|----|----|--|
| 9  | 10 | 11 | 12 | 13 | 14 | 15 | 16 |  |
| 17 | 18 | 19 | 20 | 21 | 22 | 23 | 24 |  |
| 25 | 26 | 27 | 28 | 29 | 30 | 31 | 32 |  |
| 33 | 34 | 35 | 36 | 37 | 38 | 39 | 40 |  |
| 41 | 42 | 43 | 44 | 45 | 46 | 47 | 48 |  |
| 49 | 50 | 51 | 52 | 53 | 54 | 55 | 56 |  |
| 57 | 58 | 59 | 60 | 61 | 62 | 63 | 64 |  |
|    |    |    |    |    |    |    |    |  |

Figure II-8. Pixel Numbering Scheme

11-26

Harnly and Jensen (3) determined that the correlation coefficient matrix derived from using the above numbering scheme is:

Harnly and Jensen (3) also determined that the correlation terms not associated with the first and second nearest neighbors of a given pixel, could be approximated as essentially zero. For example, pixel 28 in Figure II-8, would have non-zero correlation terms only for pixels 10, 11, 12, 13, 14, 18, 19, 20, 21, 22, 26, 27, 28, 29, 30, 34, 35, 36, 37, 38, 42, 43, 44, 45, and 46. The measurement noise covariance matrix is obtained by multiplying the derived correlation coefficient matrix by the variance of the background noise.

Harnly and Jenson (3) also found that the effect of time correlation of the background noises is negligible at the anticipated signal-to-noise ratios. At this point, all the necessary terms for the measurement equation (2-31) have been developed in full.

### 2.11 Summary

This chapter has presented a number of models which have been used to model the real world. The processes that were modeled include: atmospheric jitter, target dynamics and shape effects, and background and FLIR noises. Deterministic target trajectories were introduced to provide baseline and realistic tracking scenarios to test the tracker algorithm. Finally, a number of coordinate frames were defined to aid in generating the appropriate realizations of the target intensity function in the FLIR image plane, and the entire truth model simulation was established.

# 3.1 Introduction

The two tracker configurations, which served as the foundation of this research, were developed in studies by Kozemchak (5) and Millner (13). These trackers are sufficiently different that each must be presented separately in this chapter. Both of these trackers were first developed by Rogers (14) for multiple hot-spot targets performing very benign maneuvers. The later trackers extended the previous work to include targets performing much more highly dynamic and realistic trajectories.

The first tracker to be presented in this chapter is the one developed by Kozemchak (5). As shown in Figure I-1, it uses an extended Kalman filter to provide estimates of target position, velocity, and acceleration, as well as estimates of atmospheric disturbances. The need to include the estimate of target acceleration with the position and velocity estimates in a tracking scenario against highly maneuverable targets was shown by Harnly and Jensen (3). In order to maintain a reasonable computational burden for filter operation while still having the best possible target model, two different target models were used. The first used a simple first-order Gauss-Markov acceleration model (3,5,12,13,14,15,17), while the second used a constant speed, constant turn-rate model for target acceleration (5,17). While non-linear, the latter model has been shown to provide a much better description of real world target

dynamics when operating at short ranges, as in the case for air-to-air combat (2,17). Development of the equations needed to propagate the filter estimates and to update these estimates with FLIR measurement data will be presented later in the chapter.

a sa manané a sa manéné na kané na kané

While the above algorithm has tracked simulated targets adequately, the non-linearity of the problem requires a large number of equations to be processed on-line in realtime. If some sort of linear relationship in estimating the target parameters could be established, then many of the quantities needed to propagate and update the target estimates could be pre-computed and the on-line computational burden could be reduced. This desire for reducing the computational loading led to the development of an alternative tracker design to the extended Kalman filter tracker.

This alternative tracker uses the estimated target shape as a template in a correlator to provide pseudomeasurements to a linear Kalman filter, as shown in Figure I-2. These pseudo-measurements allow the use of the linear Kalman filter because they are in reality offset distances from the center of the field of view in the FLIR image plane. These distances are linear functions of the chosen state variables described in Chapter II (13). This tracker uses a first-order Gauss-Markov target acceleration model because the linear measurement model motivates use of a linear target descriptor, in order to yield a linear Kalman filter as the optimal data processor. Development of the

necessary equations to propagate and update these target estimates as well as a description of the correlation algorithm used will also be presented in this chapter.

# 3.2 Extended Kalman Filter Tracker

As stated earlier, this tracker uses an extended Kalman filter to provide estimates of target dynamics and atmospheric disturbances. Implicit in this use of the extended Kalman filter is the assumption that the non-linear target intensity function can be adequately linearized about the estimated states by using a Taylor series truncated to first order. Due to the relatively high measurment rate used in this research, such an approximation is considered valid.

3.2.1 Gauss-Markov Target Acceleration Model. The first of the filter models represents target dynamics acceleration and atmospheric jitter position as stationary, first-order Gauss-Markov processes. Such processes can be generated as the output of a first-order lag driven by zeromean white Gaussian noise (6). A third-order model for atmospheric jitter position was presented in Chapter II, but because the double pole appearing in that model is sufficiently far away from the single dominant pole, it has little effect on the low frequency characteristics of the jitter (5). The filter vector states can be defined as target position, velocity, and acceleration, and jitter position, in each direction of the FLIR image plane:

 $\underline{\mathbf{x}}_{\mathbf{F}} = [\mathbf{x}_{\mathbf{D}} \quad \mathbf{y}_{\mathbf{D}} \quad \mathbf{v}_{\mathbf{x}} \quad \mathbf{v}_{\mathbf{y}} \quad \mathbf{a}_{\mathbf{x}} \quad \mathbf{a}_{\mathbf{y}} \quad \mathbf{x}_{\mathbf{A}} \quad \mathbf{y}_{\mathbf{A}}]^{\mathrm{T}}$ (3-1)

The relationship between the states are as follows:

$$\mathbf{\dot{x}}_{\mathrm{D}} = \mathbf{v}_{\mathrm{X}}$$
 (3-2a)

$$\dot{\mathbf{y}}_{\mathrm{D}} = \mathbf{v}_{\mathrm{y}}$$
 (3-2b)

$$\mathbf{\dot{v}}_{\mathbf{x}} = \mathbf{a}_{\mathbf{x}}$$
 (3-2c)

$$\dot{\mathbf{v}}_{\mathbf{y}} = \mathbf{a}_{\mathbf{y}}$$
 (3-2d)

$$\dot{a}_{x} = (-1/T_{DF})a_{x} + w_{Dx}$$
 (3-2e)

$$\ddot{a}_{y} = (-1/T_{DF})a_{y} + w_{Dy}$$
 (3-2f)

$$\dot{x}_{A} = (-1/T_{AF})x_{A} + w_{AX}$$
 (3-2g)

$$\ddot{y}_{A} = (-1/T_{AF})y_{A} + w_{AV}$$
 (3-2h)

where  $T_{DF}$  = correlation time for target acceleration  $T_{AF}$  = correlation time for atmospheric jitter  $w_{Dx}$ ,  $w_{Dy}$ ,  $w_{Ax}$ ,  $w_{Ay}$  = white Gaussian noise processes of zero mean and strength depending on the effect being modelled.

Note that identical independent models are used to represent effects in x- and y- directions of the FLIR image plane.

 $\mathbf{O}$ 

From the above relationships a state vector differential equation can be written in standard form:

$$\underline{\mathbf{x}}_{\mathbf{F}}(t) = \underline{\mathbf{F}}_{\mathbf{F}}(t) \underline{\mathbf{x}}_{\mathbf{F}}(t) + \underline{\mathbf{G}}_{\mathbf{F}}(t) \underline{\mathbf{w}}_{\mathbf{F}}(t)$$
(3-3)

where  $\underline{F}_{F}(t)$  = the system plant matrix which is constant for this application and can be written as:

$$\underline{\mathbf{F}}_{\mathbf{F}} = \begin{bmatrix} 0 & 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & -1/T_{\text{DF}} & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 0 & -1/T_{\text{AF}} & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 0 & 0 & -1/T_{\text{AF}} \end{bmatrix}$$
(3-4)

 $\underline{G}_{\mathbf{F}}(t)$  = the system noise input matrix which also is constant for this application

$$\underline{G}_{F} = \begin{bmatrix} (4 \ \frac{0}{x} \ 4) \\ (4 \ \frac{1}{x} \ 4) \end{bmatrix}$$
(3-5)

 $\underline{w}_{F}(t)$  = noise vector containing mutually independent white Gaussian noise processes  $w_{Dx}$ ,  $w_{Dy}$ ,  $w_{Ax}$ ,  $w_{Ay}$  with statistics:

$$E\{\underline{w}_{F}(t)\} = \underline{0}$$

$$E\{\underline{w}_{F}(t)\underline{w}_{F}^{T}(t+)\} = \underline{Q}_{F}\delta()$$

$$\underline{Q}_{\mathbf{F}} = \begin{bmatrix} 2\sigma_{\mathbf{DF}}^{2}/T_{\mathbf{DF}} & 0 & 0 & 0 \\ 0 & 2\sigma_{\mathbf{DF}}^{2}/T_{\mathbf{DF}} & 0 & 0 \\ 0 & 0 & 2\sigma_{\mathbf{AF}}^{2}/T_{\mathbf{AF}} & 0 \\ 0 & 0 & 0 & 2\sigma_{\mathbf{AF}}^{2}/T_{\mathbf{AF}} \end{bmatrix}$$
(3-6)

 $\sigma_{\rm DF}^2$  = assumed target dynamics noise variance (or rms value)

 $\sigma_{AF}^{2}$  = assumed atmospheric jitter noise variance (or rms value)

3.2.2 State Propagation of the Gauss-Markov Model. Due to the linear nature of the filter state model equations above, the state estimates can be propagated using standard Kalman filter propagation equations.

$$\underline{\hat{\mathbf{X}}}(t_{i+1}) = \underline{\Phi}_{\mathbf{F}}(t_{i+1}, t_i) \underline{\hat{\mathbf{X}}}(t_i^{+})$$
(3-7a)

$$\underline{P}(t_{i+1}^{-}) = \underline{\Phi}_{F}(t_{i+1}, t_{i}) \underline{P}(t_{i}^{+}) \underline{\Phi}_{F}^{T}(t_{i+1}, t_{i})$$

$$+ \int_{t_{i}}^{t_{i+1}} \underline{\Phi}_{F}(t_{i+1}, \tau) \underline{G}_{F} \underline{Q}_{F} \underline{G}_{F}^{T} \underline{\Phi}_{F}^{T}(t_{i+1}, \tau) d\tau$$

$$(3-7b)$$

- where  $\underline{\Phi}_{\mathbf{F}}(t_{i+1}, t_i) =$  filter state transition matrix for propagating from time  $t_i$  to time  $t_{i+1}$ 
  - P(ti<sup>+</sup>) = conditional state covariance matrix after measurement update at time ti
  - $\underline{P}(t_{i+1}) = \text{conditional state covariance}$ matrix before measurement update at time  $t_{i+1}$ .

The filter state transition matrix  $\underline{\Phi}(t_{i+1}, t_i)$  must satisfy the following differential equation:

$$\underline{\mathbf{\Phi}}_{\mathbf{F}}(\mathsf{t},\mathsf{t}_{\underline{i}}) = \underline{\mathbf{F}}_{\mathbf{F}} \underline{\mathbf{\Phi}}_{\mathbf{F}}(\mathsf{t},\mathsf{t}_{\underline{i}}) \tag{3-8}$$

over the interval  $(t_i, t_{i+1})$ , given the initial condition  $\underline{\bullet}_F(t_i, t_i) = \underline{I}$ . Because of the time invariant  $\underline{F}_F$  matrix, the state transition matrix,  $\underline{\bullet}(t_{i+1}, t_i)$ , is only a function of the sampling time  $\Delta t$  ( =  $[t_{i+1} - t_i]$ ) and can be evaluated via Laplace domain techniques or via matrix exponentials as:

$$\mathbf{P}(t_{i+1}, t_i) = \begin{vmatrix} 1 & 0 & \Delta t & 0 & J1 & 0 & 0 & 0 \\ 0 & 1 & 0 & \Delta t & 0 & J1 & 0 & 0 \\ 0 & 0 & 1 & 0 & J2 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & J2 & 0 & 0 \\ 0 & 0 & 0 & 0 & J3 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & J3 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 0 & J4 & 0 \\ 0 & 0 & 0 & 0 & 0 & 0 & 0 & J4 \end{vmatrix}$$
(3-9)

where J1 =  $T_{DF} [\Delta t - T_{DF} (1 - \exp(-\Delta t / T_{DF}))]$ 

 $J2 = T_{DF} [1 - \exp(-\Delta t/T_{DF})]$   $J3 = \exp(-\Delta t/T_{DF})$  $J4 = \exp(-\Delta t/T_{AF})$ 

()

The solution to the integral in Equation (3-7) becomes:

$$\underline{Q}_{FD} = \begin{bmatrix}
A1 & 0 & A2 & 0 & A3 & 0 & 0 & 0 \\
0 & A1 & 0 & A2 & 0 & A3 & 0 & 0 \\
A2 & 0 & A4 & 0 & A5 & 0 & 0 & 0 \\
0 & A2 & 0 & A4 & 0 & A5 & 0 & 0 \\
0 & A2 & 0 & A4 & 0 & A5 & 0 & 0 \\
0 & A3 & 0 & A5 & 0 & A6 & 0 & 0 \\
0 & A3 & 0 & A5 & 0 & A6 & 0 & 0 \\
0 & 0 & 0 & 0 & 0 & 0 & A7 & 0 \\
0 & 0 & 0 & 0 & 0 & 0 & A7 & 0 \\
0 & 0 & 0 & 0 & 0 & 0 & A7 & 0
\end{bmatrix}$$
(3-1)
where A1 =  $\sigma_{DF}^{2} [2T_{DF}\Delta t^{3}/3 - 2T_{DF}^{2}\Delta t^{2} - 4T_{DF}^{3}\Delta texp(-\Delta t/T_{DF})]$ 

$$\mathbf{M} = \mathbf{\sigma}_{\mathrm{DF}}^{2} (12_{\mathrm{DF}}^{2} \Delta t^{-1})^{2} (12_{\mathrm{DF}}^{4} \exp(-2\Delta t/T_{\mathrm{DF}}) + T_{\mathrm{DF}}^{4})$$

$$+ 2T_{\mathrm{DF}}^{3} \Delta t - T_{\mathrm{DF}}^{4} \exp(-2\Delta t/T_{\mathrm{DF}}) + T_{\mathrm{DF}}^{4})$$

$$+ 2T_{\mathrm{DF}}^{2} (T_{\mathrm{DF}}^{2} \Delta t^{2} + 2T_{\mathrm{DF}}^{2} \Delta t \exp(-\Delta t/T_{\mathrm{DF}}) + T_{\mathrm{DF}}^{3})$$

$$- 2T_{\mathrm{DF}}^{3} \exp(-\Delta t/T_{\mathrm{DF}}) - 2T_{\mathrm{DF}}^{2} \Delta t$$

$$+ T_{\mathrm{DF}}^{3} \exp(-2\Delta t/T_{\mathrm{DF}}) - 2T_{\mathrm{DF}}^{2} \Delta t$$

$$+ T_{\mathrm{DF}}^{3} \exp(-2\Delta t/T_{\mathrm{DF}}) + T_{\mathrm{DF}}^{2}$$

$$- T_{\mathrm{DF}}^{2} (-2\Delta t/T_{\mathrm{DF}}) + T_{\mathrm{DF}}^{2} + 4T_{\mathrm{DF}}^{2} \exp(-\Delta t/T_{\mathrm{DF}})$$

$$- T_{\mathrm{DF}}^{2} \exp(-2\Delta t/T_{\mathrm{DF}}) ]$$

$$+ T_{\mathrm{DF}}^{2} (T_{\mathrm{DF}} - 2T_{\mathrm{DF}} \exp(-\Delta t/T_{\mathrm{DF}}) + T_{\mathrm{DF}} \exp(-2\Delta t/T_{\mathrm{DF}}) ]$$

$$+ T_{\mathrm{DF}}^{2} (T_{\mathrm{DF}} - 2T_{\mathrm{DF}} \exp(-\Delta t/T_{\mathrm{DF}}) + T_{\mathrm{DF}} \exp(-2\Delta t/T_{\mathrm{DF}}) ]$$

$$+ T_{\mathrm{DF}}^{2} (1 - \exp(-2\Delta t/T_{\mathrm{DF}}) ]$$

$$+ T_{\mathrm{DF}}^{2} (1 - \exp(-2\Delta t/T_{\mathrm{DF}}) ]$$

3.2.3 Constant Turn-Rate Target Acceleration Mode The constant turn rate model has been shown to model t dynamics of real world airborne targets at close rar better than the Gauss-Markov acceleration model (2,17). tradeoff of using this improved model is the introduction
non-linearities into the propagation equations. The state differential equation becomes:

$$\underbrace{\mathbf{x}_{F}(t) = \underline{f}[\underline{x}_{F}(t)] + \underline{G}_{F}\underline{w}_{F}(t) = \begin{bmatrix} v_{X} & & \\ v_{Y} & & \\ a_{X} & & \\ a_{Y} & & \\ -\omega^{2}v_{X} + w_{DX}(t) & \\ -\omega^{2}v_{Y} + w_{DY}(t) & \\ (-1/T_{AF})x_{A} + w_{AX}(t) & \\ (-1/T_{AF})y_{A} + w_{AY}(t) \end{bmatrix} }$$

$$(3-11)$$

where  $\boldsymbol{\omega}$  = magnitude of the target's turn rate on the FLIR image plane derived via the relationship,

$$\boldsymbol{\omega} = |\underline{\mathbf{v}} \times \underline{\mathbf{a}}| / |\underline{\mathbf{v}}|^2 = (\mathbf{v}_{\mathbf{x}} \mathbf{a}_{\mathbf{y}} - \mathbf{v}_{\mathbf{y}} \mathbf{a}_{\mathbf{x}}) / (\mathbf{v}_{\mathbf{x}}^2 + \mathbf{v}_{\mathbf{y}}^2)$$

All other variables are as defined in the development of the Gauss-Markov target acceleration model. With the statistics of the noise vector being:

$$\underline{Q}_{F} = \begin{bmatrix} \sigma_{DF}^{2} & 0 & 0 & 0 \\ 0 & \sigma_{DF}^{2} & 0 & 0 \\ 0 & 0 & 2\sigma_{AF}^{2}/T_{AF} & 0 \\ 0 & 0 & 0 & 2\sigma_{AF}^{2}/T_{AF} \end{bmatrix}$$
(3-12)

The filter estimates are propagated forward in time by integrating:

$$\underline{\hat{\mathbf{x}}}(t/t_i) = \underline{f}[\underline{\hat{\mathbf{x}}}(t/t_i)] \quad (3-13)$$

$$\underline{\hat{\mathbf{x}}}(t/t_i) = \underline{F}[t_i \underline{\hat{\mathbf{x}}}(t/t_i)] \underline{P}(t/t_i) + \underline{P}(t/t_i) \underline{F}^{\mathrm{T}}[t_i \underline{\hat{\mathbf{x}}}(t/t_i)] + \underline{G}(t) \underline{Q}(t) \underline{G}^{\mathrm{T}}(t) \quad (3-14)$$

where  $(t/t_i)$  means at time t, given measurements through time  $t_i$ , starting from the initial conditions:

$$\underline{\hat{\mathbf{X}}}(t_i/t_i) = \underline{\hat{\mathbf{X}}}(t_i^+) \qquad (3-15)$$

$$\underline{P}(t_{i}/t_{i}) = \underline{P}(t_{i}^{+})$$
 (3-16)

The non-linear function  $\underline{f}[\underline{x}(t/t_i)]$  can be shown to be equivalent to the time rate of change of the estimate of the state at time t. Furthermore, since there is a desire to keep the computational burden as low as possible, the integral of Equation (3-13) will be approximated with first order Euler integration, and so the state propagation equation becomes:

$$\underline{\hat{\mathbf{X}}}(t_{i+1}) = \underline{\hat{\mathbf{X}}}(t_i^+) + \underline{\hat{\hat{\mathbf{X}}}}(t_i^-/t_i^-)\Delta t \qquad (3-17)$$

The relationship assumes that the time rate of change of the state vector is piecewise constant during the time interval,  $\Delta t$ . This approximation is valid when the propagation time,  $\Delta t$ , is small compared to the natural transient times of the system.

To solve Equation (3-14) in real time would be both computationally burdensome and time consuming because of a time-variant  $\underline{F}_{F}$  matrix which requires continuous re-calculation of the state transition matrix. For practical implementation, a first order Euler integration approximation was made here as well. First,  $\underline{F}[t; \underline{\hat{X}}(t/t_i)]$  is assumed to be piecewise constant during each time interval,  $\Delta t$ , and can be derived as a function of the current state estimate.

$$\underline{F}(t_{i}) = \partial \underline{f}[\underline{x}] / \partial \underline{x} | \underline{x} = \underline{\hat{x}}(t_{i}^{+})$$
(3-18)

Given an invariant system plant matrix for each sample period, the state transition matrix can be determined. However, even this approximation requires significant numbers of operations to calculate  $\Phi_F(t_{i+1},t_i)$  and to determine the integral of Equation (3-14). To resolve this problem the upper-left 6 x 6 portion of the state transition matrix was truncated to first order terms.

$$\underline{\Phi}_{\mathbf{F}}(\mathsf{t}_{i+1},\mathsf{t}_i) = \underline{I} + \underline{F}(\mathsf{t}_i)\Delta \mathsf{t} \qquad (3-19)$$

The remaining portion of the state transition matrix is associated with the atmospheric jitter model and since it is time invariant, can be determined exactly, as J4 in Equation (3-9). The integral of Equation (3-15) can be similarly approximated with:

$$\underline{Q}_{FD} = \underline{G}_{F} \underline{Q}_{F} \underline{G}_{F}^{T} \Delta t \qquad (3-20)$$

3.2.4 Measurement Update Equations. The measurement equation presented in Chapter II, (2-31), can be written in general form as:

$$z_{k1}(t_i) = h_{k1}[\underline{x}(t_i), t_i] + v_{k1}(t_i)$$
 (3-21)

where  $z_{k1}(t_i)$  represents the average intensity in the FLIR

image plane at the kl-th pixel at time ti, hkl represents the average intensity over that pixel due to the true target intensity, and v<sub>k1</sub> represents the summed effects of background corruption and internal FLIR noise. The average intensities over each pixel in the FLIR image plane together form the target intensity shape function in terms of the most recent target measurement. The measurement information, provided at each of the 64 pixels in the tracking window, is used to update the filter and to produce a new estimate of the target centroid location in the FLIR image plane. The extended Kalman filter was used to incorporate the measurement information because of its ability to handle the nonlinearities of the problem and because it is less computationally burdensome than other non-linear filters  $(7:Ch \ 12)$ .

The need to minimize computational loading also motivated the use of the inverse covariance form of the measurement update equations. Use of this form eliminated the need to perform a 64 x 64 matrix inversion at every update time (12). Thus, the update relations are:

$$\underline{P}^{-1}(t_{i}^{+}) = \underline{P}^{-1}(t_{i}^{-}) + \underline{H}^{T}(t_{i})\underline{R}^{-1}(t_{i})\underline{H}(t_{i})$$
(3-22)

$$\underline{P}(t_i^+) = [\underline{P}^{-1}(t_i^+)]^{-1}$$
(3-23)

$$\underline{K}(t_{i}) = \underline{P}(t_{i}^{+})\underline{H}^{T}(t_{i})\underline{R}^{-1}(t_{i})$$
(3-24)

$$\underline{\mathbf{x}}(\mathtt{t_i}^+) = \underline{\mathbf{x}}(\mathtt{t_i}^-) + \underline{\mathbf{K}}(\mathtt{t_i}) \{\underline{\mathbf{z}}(\mathtt{t_i}) - \underline{\mathbf{h}}[\underline{\mathbf{x}}(\mathtt{t_i}^-), \mathtt{t_i}]\} \quad (3-25)$$

where  $\underline{H}(t_i) = \partial \underline{h}[\underline{x}, t_i] / \partial \underline{x} \mid \underline{x} = \underline{R}(t_i)$ 

the first partial of the average intensity function evaluated at the most

#### recent state estimate

 $P(t_i) = propagated conditional covariance matrix$  $before measurement update at time <math>t_i$ .

- $\underline{P}(t_i^+)$  = conditional covariance matrix after measurement update at time  $t_i$ .
  - $K(t_i) = Kalman filter gain$

- $\underline{x}(t_i)$  = propagated state estimate before measurement update at time  $t_i$ .
- $\underline{X}(t_i^+)$  = state estimate after measurement update at time  $t_i$ .
- $\underline{h}[\underline{\hat{x}}(t_i^{-}), t_i] = \text{non-linear measurement function of} \\ average intensities at time t_i as a function of the most recent state estimate.}$ 
  - $\underline{z}(t_i) = actual realization of the measurement vector at time <math>t_i$ .

The method used to derive the non-linear and linearized intensity functions will now be presented.

3.2.5 Derivation of Non-linear and Linearized Intensity Functions. The extended Kalman filter tracker uses the nonlinear intensity function  $\underline{h}[\underline{x}(t_i), t_i]$ , and the linearized intensity function,  $\underline{H}(t_i)$ , to update the filter state estimates after each measurement, as shown in Equations (3-22) to (3-25). The method for deriving the nonlinear and linearized intensity functions will be outlined here.

All of the information of a two-dimensional intensity pattern can be represented by a set of eigenvalues and eigenfunctions. To obtain all the information contained in such a pattern may require an infinite number of such functions and values. Such a representation is unattractive because it cannot be practically implemented.

#### III-12

Ideally, it is desired to have a transformation which is not burdensome yet provides an accurate representation of patterns in an x-y coordinate system. In keeping with the truth models developed in Chapter II, it should also provide decoupling of the components in the new coordinate space.

The Karhunen-Loeve transformation is one such transformation. It generates a new coordinate space with perfectly uncorrelated elements. Ine major disadvantages of this technique are that it produces a correlation matrix of dimension  $N^2 \ge N^2$  for an N  $\ge N$  input matrix, and it is very difficult to perform in its exact form (14:15).

Such disadvantages encourage use of the Fourier transform. While this does not provide perfect decorrelation of the components, it is computationally attractive and possesses a property of separability which allows a two-dimensional transform to be obtained via one-dimensional operations (14:15).

3.2.6 Two-Dimensional Fourier Transform. In Section 1.2.1, it was explained that the target image had to be centered before averaging over successive frames of data could be performed to attenuate the noise. The derivative of the nonlinear intensity function was also needed to update the extended Kalman filter (see Figure I-1). The Fourier transform was used because it allows us to perform the centering (shifting) and derivative operations in the frequency domain where they are easily done. In the Fourier transform, complex exponentials are used as eigenfunctions and the image is projected along the basis vectors associated with those exponentials.

- **1** - 1

The Fourier transform of a complex-valued function of two independent variables,  $\tilde{g}(x,y)$ , is a decomposition of  $\tilde{g}(x,y)$  into a linear combination of functions of the form  $\exp[j2\pi(f_xx + f_yy)]$  (15:8). The Fourier transform is defined by:

$$\widetilde{G}(f_{X}, f_{Y}) = F(\widetilde{g}(x, y)) = \iint_{-\infty}^{\infty} \widetilde{g}(x, y) \exp[-j2\pi(f_{X}x + f_{Y}y)] dxdy$$
(3-26)

where  $\widetilde{G}(f_x, f_y)$  = frequency spectrum, transformed function in spatial frequency domain  $\tilde{g}(x, y)$  = function in spatial domain  $f_x, f_y$  = spatial frequencies x, y = spatial variables F() = Fourier Transform operation

The inverse of this transform also exists and is defined by:

$$\tilde{g}(\mathbf{x},\mathbf{y}) = \mathbf{F}^{-1}(\widetilde{G}(\mathbf{f}_{\mathbf{x}},\mathbf{f}_{\mathbf{y}}))$$

$$= \int_{-\infty}^{\infty} \widetilde{G}(\mathbf{f}_{\mathbf{x}},\mathbf{f}_{\mathbf{y}}) \exp[+j2\pi(\mathbf{f}_{\mathbf{x}}\mathbf{x} + \mathbf{f}_{\mathbf{y}}\mathbf{y})]d\mathbf{f}_{\mathbf{x}}d\mathbf{f}_{\mathbf{y}} (3-27)$$

where the terms are as defined above.

Because the FLIR provides target information as the average intensities over its exposed area, a two-dimensional discrete Fourier transform (DFT), is used. Due to of the separability of the Fourier transform, the two-dimensional transformation can be accomplished via a series of onedimensional transformations, so the double integral can be resolved into a double summation for the discrete case. The equations for the DFT and its inverse are:

$$H(f_{x}, f_{y}) = \sum_{x=0}^{N-1} \sum_{y=0}^{N-1} h(x, y) \exp[-j2\pi(f_{x}x + f_{y}y)]$$
(3-28)

$$h(x,y) = 1/N^2 \sum_{f_x=0}^{N-1} \sum_{f_y=0}^{N-1} H(f_x, f_y) \exp[+j2\pi(f_x x + f_y y)]$$
(3-29)

where  $H(f_x, f_y) =$ frequency spectrum, transformed function in spatial frequency domain h(x,y) =function in spatial domain  $f_x, f_y =$ spatial frequencies x, y =spatial variables

The variable N refers to the period of the assumed recurring sequence in both directions. The assumption of a periodic sequence is essential in formulation of the DFT. Thus the complex sequence of intensity values is discretized into an N x N pixel array.

Although the tracking window is dimensioned to 8 x 8 pixels, the array that is processed by the DFT is dimensioned to 24 x 24. This is achieved by padding the data with a border of 8 zeros on each side of the tracking window. The purpose of this padding is to reduce the edge effects, aliasing, and leakage conditions involved when transforming finite sequences of numbers (15:18). Due to small area of the tracking window, it might not be appropriate to pad with zeros since the image intensity at the edges of the window might not be essentially zero. To pad such an image with zeros would introduce artificial edge effects (5:10). However, since the tracking window is actually part of a much larger field of data from the FLIR, it is possible to pad the data of the tracking window with real data instead of zeros and thus minimize the introduced edge effects.

3.2.7 Shifting Property of the Fourier Transform. Since the target intensity pattern must be generated from noise corrupted FLIR data, interframe smoothing is necessary to attenuate the noise. This smoothing requires the target intensity profile to be centered from frame to frame since each pattern experiences different shifts from the center of the field of view. Successive centered frames of data can then be averaged to attenuate the noise, while at the same time accentuating the true target intensity function. Centering of each frame utilizes the shifting property of the Fourier transform as well as the filter's estimated location of the intensity profile.

The shift theorem for the Fourier transform states that a linear phase shift in the frequency domain corresponds to translation in the spatial domain. Because of the assumed periodic nature of the sampled data, such a phase shift can be thought of as a cylindrical shift. That is, rotation of the samples out one side of the interval results in rotating them into the other side of the interval. This property can be used to show that the only difference between the centered image and a translated image is a linear phase shift proportional to the spatial displacement in the x- and y-

directions as in (3-30), if 
$$F(\tilde{g}(x,y)) = \tilde{G}(f_x, f_y)$$
, then

 $F(\tilde{g}(x-a,y-b)) = \tilde{G}(f_x,f_y)exp[-j2\pi(f_xa + f_yb)]$  (3-30) where a = shift of the spatial function in the xdirection

> b = shift of the spatial function in the ydirection

The filter's updated estimate of the location of the centroid of the intensity profile with respect to the center of the FLIR field of view can be used to determine the negating phase shift required to obtain the centered image needed for interframe smoothing.

3.2.8 Exponential Smoothing. The intensity profile of the target is neither known at any given time, nor can it be measured directly. Furthermore, the measurements that are available are corrupted by FLIR measurement noises as well as background noise. It is assumed that for most sampling rates, these corruptive noises tend to change significantly faster than the target intensity pattern from sample period to sample period (12).

A memory efficient, exponential smoothing algorithm was used to exploit this property. It captures the essence of a true finite memory averager without the need for storage of all the previous frames of data. This algorithm can be expressed as (1):

 $\underline{\hat{y}}(t) = \alpha \underline{y}(t) + (1-\alpha)\underline{\hat{y}}(t-1) \qquad (3-31)$ where  $\underline{\hat{y}}(t)$  = current averaged value y(t) = current data frame

 $\hat{\mathbf{Y}}(t-1)$  = previous averaged data frame

 $\alpha$  = smoothing constant,  $0 \leq \alpha < 1$ 

The smoothing constant,  $\alpha$ , can be adjusted to account for the dynamics of the image. For slowly changing images, a smaller  $\alpha$  would be appropriate, while a rapidly changing image requires that the most recent frames of data be weighted heavily. An  $\alpha$  should be chosen that gives the best performance characteristics for all expected image variations. Appropriate values for  $\alpha$  are:  $0 \leq \alpha < 1$ .

The necessary operations required to perform interframe smoothing have now been defined. A method for centering the intensity profiles and smoothing the data to attenuate the effects of noise has also been presented. The following operations are performed to get a centered intensity profile in the spatial domain.

- 1) The Fourier transform of the raw FLIR measurements is calculated
- 2) The appropriate negating phase shift is applied to center the image in the frequency domain based on the extended Kalman filter's estimate of the location of the centroid of the image
- 3) Interframe smoothing of the centered data is performed
- 4) The predicted target centroid position at the next sample time,  $\Re(t_{i+1})$ , is the sum of the predicted position due to target dynamics and the predicted position due to atmospheric disturbances. In equation form,

$$\underline{\hat{x}}(t_{i+1}) = \underline{\hat{x}}_{dvn}(t_{i+1}) + \underline{\hat{x}}_{atm}(t_{i+1})$$

However, as stated in Chapter I, control is applied at each sample time to zero the predicted dynamic states. That is, the FLIR is pointed so the center of the FLIR field of view points toward the predicted target centroid position due to target dynamics. So now, the intensity shape function is evaluated at the states after control has been applied,

 $\underline{\hat{\mathbf{x}}}(\mathbf{t}_{i+1}^{-\mathbf{C}}) = \underline{\hat{\mathbf{x}}}_{atm}(\mathbf{t}_{i+1}^{-})$ 

where the superscript c denotes after controller application

5) The inverse Fourier transform is performed to obtain the intensity profile

3.2.9 Derivative Property of the Fourier Transform. In Section 3.2.4, the derivative of the intensity function with respect to the states was shown to be necessary to perform filter updates. This can be easily accomplished in the frequency domain, where differentiation in the spatial domain becomes simple multiplication by j2 ( $f_x + f_y$ ). This process can be described by:

$$\mathbf{F}[\partial \underline{\mathbf{h}}(\mathbf{x}, \mathbf{y}) / \partial \underline{\mathbf{x}}] = \mathbf{j} 2\pi \mathbf{f}_{\mathbf{x}} \mathbf{F}[\underline{\mathbf{h}}(\mathbf{x}, \mathbf{y})]$$
(3-32)

$$F[\partial h(x,y) / \partial y] = j 2\pi f_v F[h(x,y)] \qquad (3-33)$$

The necessary variables to propagate and update the estimates of the extended Kalman filter tracker have now been presented. The next sections of this chapter will cover the linear Kalman filter/correlator tracker configuration shown in Figure I-2, which shares many of the same processes described above.

# 3.3 Linear Kalman Filter/Correlator Tracker

As stated in the introduction to this chapter, a desire to obtain an algorithm that was less computationally burdensome to implement than a high-measurement-dimensioned extended Kalman filter led to the investigation of the feasibility of using an enhanced correlator and Kalman filter in the same tracker. The correlator can be considered enhanced because it uses the estimated intensity function as a template against which to compare the new target information. Also, thresholding is performed on the cross-correlation to reduce the likelihood that false peaks will skew the estimate of the point of maximum correlation. This allows the correlator to incorporate a priori information about the target into the algorithm instead of operating solely on the collected data, as well as providing a better target template than the previous frame of raw FLIR data. Correlation of the template and the target information is used to estimate the relative position offsets from one sample period to the next. These pseudo-measurements are provided to the linear Kalman filter, which uses them to generate a new estimate of the target intensity function and target centroid location. Although Millner (13) investigated many different correlation techniques, the FFT method exhibited the best performance characteristics and was the method employed for this research.

3.3.1 FFT Correlator. The correlator used in this study computes the cross correlation of the template, which is the estimated target intensity function positioned at the best estimate of centroid offset  $(\underline{h}[\underline{x}(t_i^{-c}]))$ , and the raw data from the FLIR. The FFT can be used to perform the cross-correlation as illustrated below:

III-20

$$\mathbf{F}[\underline{g}(\mathbf{x},\mathbf{y})] = \underline{G}(\mathbf{f}_{\mathbf{x}},\mathbf{f}_{\mathbf{y}}) \tag{3}$$

$$F[\underline{1}(x,y)] = \underline{L}(f_x, f_y)$$
(3)

$$F[\underline{g}(\mathbf{x},\mathbf{y})*\underline{\mathbf{l}}(\mathbf{x},\mathbf{y})] = \underline{G}(\mathbf{f}_{\mathbf{x}},\mathbf{f}_{\mathbf{y}}) \cdot \underline{\mathbf{L}}^{*}(\mathbf{f}_{\mathbf{x}},\mathbf{f}_{\mathbf{y}})$$
(3)

where  $\underline{g}(x,y) \star \underline{1}(x,y) = \text{cross-correlation of the two}$ dimensional spatial sequences  $\underline{g}(x,y)$  and  $\underline{1}(x,y)$ 

$$\underline{L}^{*}(f_{x}, f_{y}) = \text{complex conjugate of the Four:}$$
  
transform of the sequence 1(x

By taking the inverse FFT, or IFFT, of Equation (3-36), cross-correlation is obtained:

$$\underline{R}(x,y) = \underline{q}(x,y) * \underline{1}(x,y) = F^{-1}[\underline{G}(f_x,f_y) \cdot \underline{L}^*(f_x,f_y)]$$
(3)

Once the cross-correlation,  $\underline{R}(x,y)$ , has b determined, it may be necessary to process it wit threshholding function. If any one element of  $\underline{R}(x,y)$ less than some pre-selected fraction of the element maximum correlation, then it will be considered as hav poor correlation information and be set to zero. I should reduce the likelihood of false peaks biasing estimated offset between the template and the target da While using a true maximum correlation finder wo eliminate this source of error, it is not attractive implement due to such problems as ambiguity of multi peaks and heavier computational loading, so a correlat peak was "found" by a center of mass correlation.

After threshholding, a centroid summation was use locate the center of mass of  $\underline{R}(x,y)$ . This centroid assumed to be a good indication of the peak location. location of the autocorrelation centroid is calculated in either the x- or y- direction using the following equation:

The calculated position of the centroid of  $\underline{R}(x,y)$  is the correlator's estimate of the offset of the target from the center of the data frame. This information is now the "measurement" passed to the Kalman filter. The appropriate measurement equation is:

$$\underline{z}(t_i) = \underline{H}_{F} \underline{x}_{F}(t_i) + \underline{v}_{F}(t_i)$$
(3-39)

where 
$$\underline{z}(t_1) = \begin{bmatrix} x_{DC} \\ y_{DC} \end{bmatrix} + \begin{bmatrix} x_{AC} \\ y_{AC} \end{bmatrix}$$

= the estimate x- and y- coordinates of the centroid of the target intensity function as estimated by the correlation algorithm. The estimate of the centroid location is based on filter predicted centroid location due to dynamics and atmospherics.

$$\underline{\mathbf{H}}_{\mathbf{F}} = \begin{bmatrix} 1 & 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 & 1 \end{bmatrix}$$

= the linear combination of the state variables which contribute to the measurement elements

$$\underline{\mathbf{x}}_{\mathbf{F}}(\mathbf{t}_{\mathbf{i}}) = [\mathbf{x}_{\mathbf{D}} \quad \mathbf{y}_{\mathbf{D}} \quad \mathbf{v}_{\mathbf{x}} \quad \mathbf{v}_{\mathbf{y}} \quad \mathbf{a}_{\mathbf{x}} \quad \mathbf{a}_{\mathbf{y}} \quad \mathbf{x}_{\mathbf{A}} \quad \mathbf{y}_{\mathbf{A}}]^{\mathrm{T}}$$

III-22

 $\underline{v}_{F}(t_{i})$  = corruptive noise assumed to be a white Gaussian process with statistics

$$E\{\underline{v}_{F}(t_{i})\} = \underline{0}$$
$$E\{\underline{v}_{F}(t_{i})\underline{v}_{F}^{T}(t_{j})\} = \underline{R}_{F}(t_{i})\delta_{ij}$$

Recall that each measurement is considered to be the sum total of the position offsets from the center of the field of view due to target dynamics and atmospheric jitter, and the corruptive noise.

The appropriate propagation equations for this filter are the same as those developed earlier for the Gauss-Markov target acceleration model. This linear model was used because it is linear and, together with the linear measurement model, will yield a final filter that is totally linear. Because of the linearity of the equations in this tracker, standard Kalman filter update equations can be used:

$$\underline{K}(t_{i}) = \underline{P}_{F}(t_{i}) \underline{H}_{F}^{T} [\underline{H}_{F} \underline{P}_{F}(t_{i})] \underline{H}_{F}^{T} + \underline{R}_{F}(t_{i})]^{-1}$$
(3-40)

$$\underline{\mathbf{R}}_{\mathbf{F}}(\mathbf{t_{i}}^{+}) = \underline{\mathbf{R}}_{\mathbf{F}}(\mathbf{t_{i}}^{-}) + \underline{\mathbf{K}}(\mathbf{t_{i}})[\underline{\mathbf{z}}(\mathbf{t_{i}}) - \underline{\mathbf{H}}_{\mathbf{F}}\underline{\mathbf{R}}_{\mathbf{F}}(\mathbf{t_{i}}^{-})]$$
(3-41)

$$\underline{\underline{P}}_{F}(t_{i}^{+}) = \underline{\underline{P}}_{F}(t_{i}^{-}) + \underline{\underline{K}}(t_{i}^{-})\underline{\underline{H}}_{F}\underline{\underline{P}}_{F}(t_{i}^{-})$$
(3-42)

where all quantities have been defined previously.

## 3.4 Estimation of Dynamic Driving Noise

The dynamic driving noise matrix,  $\underline{Q}_{FD}$ , as in Equation (3-6), is used to model the expected effects of target motion. Because this motion is rarely constant, a constant  $\underline{Q}_{FD}$  would be less than optimal most of the time for a target exhibiting a wide range of manuevers. Filter performance

would be improved if this matrix could be adaptively set in response to the observed target behavior. This adaptive estimation of the noise matrix can be employed for either tracker.

An estimator for  $\underline{Q}_{FD}$  is determined as follows (71:Ch 10):

$$\underline{P}_{F}(t_{i}^{-}) = \underline{\Phi}_{F}(t_{i}, t_{i-1}) \underline{P}_{F}(t_{i-1}^{+}) \underline{\Phi}_{F}^{T}(t_{i}, t_{i-1}) + \underline{Q}_{FD}(t_{i-1})$$
(3-43)

which is a restatement of Equation (3-7b), substituting  $\underline{Q}_{FD}(t_i)$  for the integral;  $t_i$  for  $t_{i+1}$ ; and  $t_{i-1}$  for  $t_i$ .

$$\underline{P}(t_i^+) = \underline{P}_F(t_i^-) - \underline{K}(t_i)\underline{H}(t_i)\underline{P}_F(t_i^-)$$
(3-44)

Solving for  $\underline{Q}_{FD}$  using Equations (3-43) and (3-44) yields:

$$\underline{Q}_{FD}(t_{i-1}) = \underline{K}(t_i)\underline{H}(t_i)\underline{P}_F(t_i^{-}) + \underline{P}_F(t_i^{+}) - \underline{\Phi}_F(t_i,t_{i-1})\underline{P}_F(t_{i-1}^{+})\underline{\Phi}_F^T(t_i,t_{i-1}) (3-45)$$

Only the first term of Equation (3-45) is not readily available because it is desired to have the Kalman filter gain reflect the observed target behavior. To incorporate this information into Equation (3-45) using the filter residual,  $r_F$ , let

$$\underline{\mathbf{x}}_{\mathbf{F}}(\mathbf{t}_{\mathbf{i}}^{+}) - \underline{\mathbf{x}}_{\mathbf{F}}(\mathbf{t}_{\mathbf{i}}^{-}) = \underline{\mathbf{K}}(\mathbf{t}_{\mathbf{i}})\underline{\mathbf{r}}_{\mathbf{F}}(\mathbf{t}_{\mathbf{i}}) = \underline{\mathbf{\Delta}}\underline{\mathbf{x}}(\mathbf{t}_{\mathbf{i}}) \quad (3-46)$$

If the ergodic assumption is made, the ensemble average can be replaced with a time averager on a single sample, as

$$E\{\Delta \underline{x}(t_{i}) \Delta \underline{x}^{T}(t_{i})\} = 1/N \sum_{j=i-N+1}^{i} [\Delta \underline{x}(t_{j}) \Delta \underline{x}^{T}(t_{j})]$$
(3-47)

Additionally, the residual sequence has been shown to be a white Gaussian sequence of zero mean and covariance  $[\underline{H}(t_i)\underline{P}(t_i)\underline{H}^{T}(t_i) + \underline{R}(t_i)]$  (6:229).

$$E\{\underline{r}_{F}(t_{i})\underline{r}_{F}^{T}(t_{i})\} = \underline{H}(t_{i})\underline{P}(t_{i}^{-})\underline{H}^{T}(t_{i}) + \underline{R}(t_{i}) \quad (3-48)$$

so that

$$E\{\Delta \underline{x}(t_{i})\Delta \underline{x}^{T}(t_{i})\} = \underline{K}(t_{i})E\{\underline{r}_{F}(t_{i})\underline{r}_{F}^{T}(t_{i})\}\underline{K}^{T}(t_{i})$$
$$= \underline{K}(t_{i})\underline{H}(t_{i})\underline{P}(t_{i}^{-}) \qquad (3-49)$$

Combining Equations (3-45), (3-46), and (3-49) gives:

$$\frac{\Delta}{\underline{O}_{FD}}(t_{i}) = \{1/N \sum_{j=i-N+1}^{i} [\Delta \underline{x}(t_{j}) \Delta \underline{x}^{T}(t_{j})] \} + \underline{P}_{F}(t_{i}^{+})$$

$$- \frac{1}{F}(t_{i}, t_{i-1}) \underline{P}_{F}(t_{i-1}^{+}) F^{T}(t_{i}, t_{i-1})$$
(3-50)

But rather than averaging just for the first term of the above equation, averaging was performed over all terms for the N most recent sample periods:

$$\hat{\underline{Q}}_{FD}(t_{j}) = 1/N \sum_{j=i-N+1}^{i} [\Delta \underline{x}(t_{j}) \Delta \underline{x}^{T}(t_{j}) + \underline{P}_{F}(t_{j}^{+}) - \underline{\Phi}_{F}(t_{j}, t_{j-1}) \underline{P}_{F}(t_{j-1}^{+}) \underline{\Phi}_{F}^{T}(t_{j}, t_{j-1})]$$
(3-51)

This is also a closed form approximation to the maximum likelihood estimate of  $\underline{Q}_{FD}$  to be obtained simultaneously with a state estimate (7:123).

III-25

To reduce the data storage requirements, a fading memory approximation to this average was used:

$$\underline{Q}_{FD}(t_i) = k\underline{Q}_{FD}(t_{i-1}) + (1-k)\underline{Q}_{FD1}(t_i) \qquad (3-52)$$

where  $\underline{Q}_{FD1}(t_i)$  = is a single term in the summation of Equation (3-51) when j=i

k = parameter which controls the length of time old estimates of  $Q_{FD}$  are maintained, 0 < k < 1

## 3.5 Summary

This chapter presented the two tracker configurations which served as the foundations of this research effort. The developemnt of how state estimates were propagated and updated for both trackers was presented. Additionally, the use of the Fourier transform to derive the target intensity function was demonstrated.

The next chapter will discuss addition of a multiple model adaptive filter structure to the algoritms presented here.

#### 4.1 Introduction

The studies by Kozemchak (5) and Millner (13) cited in earlier chapters used a tracking window much smaller than the entire area covered by each frame of FLIR data. Only an 8 x 8 array tracking window for measurement updates and a 24 x 24 array for data processing (out of an available 500 x 400 pixel measurement array) were used in an effort to minimize the computational and memory storage requirements of the tracker (3:4). Unfortunately, limiting the dimensions of the tracker field of view increases the likelihood that the image of a highly dynamic, close-range target will be outside the tracking window during a given sample period. Such a condition causes the tracker to lose track of the target. Any effort to increase the aperture of the single filter tracker without simultaneously increasing the array dimensions will decrease tracker resolution. A reduction of resolution results in poor tracker performance for benign target trajectories.

One approach which allows the tracker field of view to be expanded without increasing the dimensionality of the data processing arrays uses a multiple model filtering algorithm. Under this approach, a second identically dimensioned filter is processed in parallel with the original filter. The differences between the two filters are that the added filter possesses a different model of target dynamics and larger field of view. By optimally combining the estimates produced by the two filters at each sample time, an overall estimate can be created. This structure will maintain the desirable high resolution for benign target trajectories while allowing the tracker to maintain lock on highly dynamic targets.

This chapter will describe the multiple model filter algorithm and how this algorithm was implemented in both the Kozemchak and Millner form of trackers.

# 4.2 Multiple Model Adaptive Filter Algorithm

Lange Distant

In any tracking scenario there are many parameters related to target motion which are highly uncertain. Such parameters may be related to target capabilities and/or target commanded maneuvers. Furthermore, these uncertain parameters can be grouped together to form, a; a vector which represents these uncertain parameters. The parameter vector a lies in a continuous parameter space, as each parameter composing a can generally assume any value over a continuous range of values. To make the estimation of a a more manageable task, this continuous space was discretized into K distinct models for the uncertain parameter vector. These models are well distributed over the range of expected values for  $\underline{a}$  (7:130). The multiple model algorithm consists of K independent (Kalman) filters, each with its own estimate of the value of a. That is, each filter's model for a contains values for each uncertain parameter, i.e. component of a, that falls within the range of possible realizations of that parameter. Given that the filter whose a vector

IV-2

most closely resembles the true values for the uncertain parameters should produce the best state estimates, one would, in a multiple filter environment, want to weight the estimates from that filter most heavily, while weighting those from the progressively less accurate models increasingly lightly.

The multiple model filter structure is shown in Figure IV-1 (7:132).



Figure IV-1. Multiple Model Filter Algorithm

1V-3

As the figure illustrates, the multiple model filter consists of a bank of K independent Kalman filters. The filters are processed in parallel, and they independently produce their own states estimates based on the incoming measurements. When a measurement update is performed at each sample time, the residuals of all filters are used to calculate conditional probabilities which are used to assign the appropriate weighting factors to the estimates of each filter.

وأجوادها والمقار فالأعلاك فستعلأ ملاما الال

These conditional probabilities, called hypothesis conditional probabilities, are the probabilities that the uncertain parameters, <u>a</u>, have assumed the same values as those modeled by the kth Kalman filter, <u>ak</u>, given the measurements received up to that time, for K Kalman filters.

 $p_{k}(t_{i}) = prob \{ \underline{a} = \underline{a}_{k} \mid \underline{Z}(t_{i}) = \underline{Z}_{i} \}$  (4-1)

- where  $\underline{a} = a$  random vector which can take on values  $\underline{a_1}$ to  $\underline{a_k}$ 
  - <u>ak</u> = values for the uncertain parameters of the kth Kalman filter where k = 1,2,...,K for K Kalman filters in the filter structure

 $Z_i$  = measurement time history up to time  $t_i$ 

 $P_k(t_i)$  = weighting factor for the estimates of the kth Kalman filter at time  $t_i$ 

The conditional probabilities at time t<sub>i</sub> can be expressed as (7:131):

IV-4

$$p_{k}(t_{i}) = f_{\underline{z}}(t_{i}) | \underline{a}, \underline{z}(t_{i-1}) \quad (\underline{z}_{i} | \underline{a}_{k}, \underline{z}_{i-1}) \quad p_{k}(t_{i-1})$$

$$/ \sum_{j=1}^{K} f_{\underline{z}}(t_{i}) | \underline{a}, \underline{z}(t_{i-1}) \quad (\underline{z}_{i} | \underline{a}_{j}, \underline{z}_{i-1}) \quad p_{j}(t_{i-1})$$

$$k = 1, 2, \dots, K \quad (4-2)$$

where the denominator is simply the sum of all computed numerator terms and thus is the scale factor required to insure that the sum of all conditional probabilities is one. This permits the use of the conditional probabilities as the weighting factors of the filters' estimates.

The conditional density functions of (4-2) may be evaluated as:

$$f_{z(ti)}|\underline{a},\underline{z}(ti-1) \quad (\underline{z}_{i}|\underline{a}_{k},\underline{z}_{i-1})$$

$$= (2\pi)^{m/2} |\underline{A}_{k}(t_{i})|^{1/2} \exp\{ \cdot \} \quad (4-3)$$
where {  $\cdot \} = \{-0.5\underline{r}_{k}^{T}(t_{i})\underline{A}_{k}^{-1}(t_{i})\underline{r}_{k}(t_{i})\}$ 

$$\underline{r}_{k} = \operatorname{residuals of the kth Kalman filter}_{k=1,2,\ldots,K}$$

$$\underline{A}_{k} = [\underline{H}_{k}\underline{P}_{k}(t_{i}^{-})\underline{H}_{k}^{T} + \underline{R}_{k}]$$

As stated earlier, one would expect the filter whose model for <u>a</u> most closely resembles the true values for the uncertain parameters to produce the best state es<sup>+</sup>imates. That is, it should have the best behaved residuals, i.e. the most zero mean, white, Gaussian, of covariance equal to the computed value of <u>A</u><sub>k</sub> (7:133). It is the size of the residuals relative to the filter's computed estimates of the variance of the residuals errors (via <u>A</u> = [<u>HPH</u><sup>T</sup> + <u>R</u>]) that indicates which is the "correct" model for <u>a</u>. As illustrated in Figure IV-1, all the filter estimates are combined to form

$$\underline{\hat{x}}(t_{i}^{+}) = \sum_{k=1}^{K} \underline{\hat{x}}_{k}(t_{k}^{+}) p_{k}(t_{i})$$
(4-4)

The conditional covariance of  $\underline{X}(t_i)$  can be expressed as:

$$\underline{P}(t_{i}^{+}) = \sum_{k=1}^{K} p_{k}(t_{i}) \left\{ \underline{P}_{k}(t_{i}^{+}) + \left[ \underline{\hat{X}}_{k}(t_{k}^{+}) - \underline{\hat{X}}(t_{k}^{+}) \right] \right\}$$

$$x \left[ \underline{\hat{X}}_{k}(t_{i}^{+}) - \underline{\hat{X}}_{k}(t_{i}^{+}) \right]^{T} \right\}$$
(4-5)

It is not absolutely necessary to compute (4-5) in the online algorithm.

As stated earlier, the filter which consistently produces the smallest residuals relative to its own estimates of its errors that is weighted most heavily. Therefore, it is important that there be significant differences between the residuals from this "best" filter and the residuals from the other mismatched filters. Failure to obtain such differences could cause the algorithm to assign inappropriately large probabilites to incorrect models of the uncertain parameter values which will result in poor performance. In terms of implementation, this means that each filter in the algorithm should be tuned for optimal performance when the true values of the uncertain parameters are identical to its model for those parameters. When tuning these filters, one should avoid a "conservative" philosophy, that is, adding large magnitudes of pseudonoise to dynamics. This would "hide" the inadequacies of the target dynamics model.

Finally, care should be taken when calculating the hypothesis conditional probability values. As can be seen in Equation (4-2), the current value of the conditional probability is the product of the conditional density of Equation (4-3) and the values of the conditional probability at the previous time, divided by the sum of all such products for the K filters. This means that if the conditional probability is allowed to go to zero at any one time, all subsequent values of the conditional probability for that filter will be zero. This will effectively shut off all future contributions from that filter's estimates into the overall multiple model filter estimate. This could reduce the overall filter's ability to respond to future changes in the true parameter values. Consequently, each conditional probability was artificially bounded to keep it from converging to zero (7:135). Another factor which motivated using a lower bound on the conditional probabilities was natural damping effect on the conditional probabilities imposed by the structure of Equation (4-2). Because the current conditional density function is multiplied by the previous value of the conditional probability, there is a certain amount of lag before the filter responds to changes in the true parameter values. By setting the lower bound to a value of 0.01, it was possible to keep the conditional probabilities from converging to zero while simultaneously improving the multiple model filter's ability to respond to

sudden changes in the true parameter values. This value for the bound was chosen based on actual test results of the program. After lower bounding each probability, the resultant probabilities are rescaled so that their sum is kept at one.

## 4.3 Implementation of the Multiple Model Algorithm

The multiple model algorithm just described was implemented in both the Kozemchak and Millner tracker formulations. While the means of implementation differed because of the inherent differences in the structures of both trackers, certain characteristics are shared by both.

In both filter models, the structure of the original filter and of the truth model as described in Chapters II and III remains unchanged. Also, with the exception of the inclusion of the second filter and the associated algorithms needed to combine the estimates via the multiple model algorithm just described, the structure of each tracker remains as outlined in Chapter I.

FLIR measurements for the original small field of view are generated as before. Measurements for the larger field of view are generated by taking the 24 x 24 array with the same field of view center as used for the original smaller field of view and averaging the intensities within each 3 x 3 pixel area within that original region, to create an 8 x 8 measurement array for the larger field of view. This averaging of the intensities over each 3 x 3 pixel was done to keep the dimensionality of the filter for the larger

| ſ | AID-814 | 1 144 | ENH   | ENHANCED TRACKING OF AIRBORNE TARGETS USING MULTIPLI<br>MODEL FILTERING TECH. (U) AIR FORCE INST OF TECH |                                                              |  |  |  |  |  |  |         | 2/4 |  |
|---|---------|-------|-------|----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------|--|--|--|--|--|--|---------|-----|--|
| ļ | UNCLAS  | SIFIE | D DEC | B3 AF                                                                                                    | HI-PHITERSON HEB OH SCHOOL OF ENGL. R<br>83 AFIT/GE/EE/84M-4 |  |  |  |  |  |  | 12/1 NL |     |  |
| - |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
|   |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
|   |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
|   |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
|   |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
|   |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
| ſ |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
|   |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
|   |         |       |       |                                                                                                          |                                                              |  |  |  |  |  |  |         |     |  |
|   |         | -     |       |                                                                                                          |                                                              |  |  |  |  |  |  | -       |     |  |

4 .

. •



field of view at the same level as that of the original filter; it's pixels are now 60  $\mu$ rad-by-60  $\mu$ rad instead of 20  $\mu$ rad-by-20  $\mu$ rad, as originally.

For the larger field of view, the measurement noise of the FLIR was assumed to be spatially uncorrelated because the distance between its pixel centers is now greater than two small (20  $\mu$ rad) pixels in length. Recall from Chapter II that spatial correlation was modeled as being non-zero only for each pixel and its two closest neighbors in all directions. In addition, the larger field of view makes it reasonable to assume that the target image intensity is essentially zero at the borders. Therefore, the larger field of view was padded with zeros, as opposed to data. Also, all noise contributions to the larger field of view were calculated as the average of the noise terms of the equivalent area for the smaller field of view.

ANALY ANALY ANALY ANALY ANALY

Finally, each filter was tuned for best performance at different target trajectories. The smaller filter was tuned for best performance at very benign trajectories (trajectory 1 as described in Chapter II). The larger filter was tuned to accommodate a highly dynamic target trajectory (trajectory 2 as described in Chapter II with a 20g commanded pull-up maneuver). As explained earlier, this tuning was performed to create significant differences between the residuals of the two filters so that the weighting factors assigned to the estimates of each filter would accurately reflect the suitability of that filter's target dynamics model to the

والمرد ويراجر والمراجر والمراجر والمراجر والمراجر والمتراج والمتراج والمتراج والمتراج والمتراج والمراجر والمراجر

maneuver being performed at that time.

4.3.1 Implementation in the Kozemchak Tracker Configuration. In this tracker model, the multiple model algorithm was implemented by rescaling the dimensions of the pixels in the second filter. The pixel sizes for the second filter were set to be 60 microradians by 60 microradians, which is a threefold increase in pixel size. Recall from Chapter I that the extended Kalman filter tracker processes the FLIR measurements directly, so the increase in pixel size corresponds to the way the FLIR measurement array for the larger field of view was created in Section 4.3. This rescaling of the pixel size also necessitated that the dynamics model for the second filter also had to be rescaled.

The measurements from the FLIR did not require rescaling except for the aforementioned averaging to account for the larger pixel size. Because of this averaging, the extended Kalman filter measurement vector,  $\underline{z}$ , remains a 64 x 1 vector, but its components are each the average intensities over a 3 x 3 block of the smaller pixels.

Because  $\underline{z}$  is a 64-dimensioned vector, the matrix  $\underline{A}$  in Equation (4-3) will be a 64 x 64 matrix. This would require that a full 64 x 64 matrix be performed at every sample time to determine each of the hypothesis conditional probabilities. For a multiple model filter with K Kalman filters, this would correspond to K 64 x 64 matrix inversions at every sample time, assuming each filter is identically dimensioned. As explained in Chapter II, a desire to avoid

doing one such inversion at each sample time required using the inverse covariance method for measurement update of the Kalman filter for the single filter tracker. However, calculation of the <u>A</u> matrix is explicitly required in the multiple model filter formulation.

Two approximations were made to alleviate this problem. First, the inverse of the <u>A</u> matrix was derived using only the diagonals of <u>A</u> (2:24). This approximation was easy to implement and requires only 64 divisions instead of the 32 million operations needed to perform a full inversion. Similarly, the calculation of the determinant of <u>A</u>, also required in Equation (4-3), requires over 8000 multiplies so it also required an approximation. Since the magnitude of the determinant is independent of the "correctness" of the filter models and it was anticipated that the major differences between the two filters would be in the residuals, the scalar term  $(2\pi)m/2 |\underline{A}_{\mathbf{k}}|^{1/2}$  terms of Equation (4-3) were ignored (2:24,7:133).

Another approach to providing a value of  $\underline{A}^{-1}$  would have been to perform full inversion of those elements of  $\underline{A}$  that are associated with the center 4 x 4 region of the field of view. This central region was chosen because it is the area where the centroid of the target is expected to be located (2:24). The  $\underline{A}$  matrix would be treated as diagonal or ignored altogether, outside of this region.

Due to the size of the matrices involved, the exponential argument of Equation (4-3) would often exceed the bounds for the exponential function as implemented on the computer (2:25). A scale factor of .01 was used to bring the argument's magnitude down to acceptable levels. While this scaling reduces the relative ratios between the two filter conditional probabilities, it was deemed acceptable until another means of scaling could be found (2:25). The implemented form of Equation (4-3) was:

 $f_{\underline{z}(ti)|\underline{a},\underline{z}(ti-1)} (\underline{z}_{i}|\underline{a}_{k},\underline{z}_{i-1}) = \exp\{-0.005\underline{r}_{k}^{T}(t_{i}) \\ \times \underline{A}_{k}^{-1}(t_{i})\underline{r}_{k}(t_{i})\} (4-6)$ 

For the approximation which uses the 4 x 4 foveal approximation, because of the reduced dimensions of the <u>A</u> matrix, no such scaling factor was introduced.

Finally, in Chapter I it was explained that for the extended Kalman filter, the filter state estimates were used to derive the non-linear intensity shape function and the linearized intensity measurement function. This assumed that the appropriate control was applied to FLIR so that the center of the tracking window was located at the predicted target centroid position due to target dynamics as shown in Equation (4-7):

 $\underline{\hat{\mathbf{x}}}_{cen}(\mathbf{t}_{i+1}^{-}) = \underline{\hat{\mathbf{x}}}_{dyn}(\mathbf{t}_{i+1}^{-}) + \underline{\hat{\mathbf{x}}}_{atm}(\mathbf{t}_{i+1}^{-})$ (4-7) where  $\underline{\hat{\mathbf{x}}}_{cen}(\mathbf{t}_{i+1}^{-}) =$  predicted target centroid location  $\underline{\hat{\mathbf{x}}}_{dyn}(\mathbf{t}_{i+1}^{-}) =$  predicted target centroid location due to dynamics. Control is applied to point the sensor toward this spot so it is effectively zeroed out; that is,  $\underline{\hat{\mathbf{x}}}_{dyn}(\mathbf{t}_{i+1}^{-})^{-}$  (dynamic location after control is applied) is zero

i al a la calandar de la calandar de

IV-12

N,

 $\underline{\mathbf{x}}_{atm}(t_{i+1})$  = predicted target centroid location due to atmospheric disturbances

Because the predicted target centroid location due to dynamics is zeroed out, the nonlinear and linearized shape functions are evaluated at the filter's predicted centroid location solely due to atmospherics.

 $(\underline{\mathbf{x}}_{dyn}(t_{i+1}) + \underline{\mathbf{x}}_{atm}(t_{i+1}) - \underline{\mathbf{x}}_{dyn}(t_{i+1}) = \underline{\mathbf{x}}_{atm}(t_{i+1})$ (4-8)

However, in the multiple model filter, the FLIR is pointed such that the predicated target centroid position due to target dynamics of the <u>combined</u> estimate is at the center of the field of view. Therefore the intensity shape function and the intensity measurement function for both filters must now be calculated based on the fact that the foveal center is now at some offset distance from the filter's estimate of the location of the target centroid. The nonlinear and linearized intensity functions are now evaluated at:

 $\underline{\mathbf{x}}_{dynk}(t_{i+1}) + \underline{\mathbf{x}}_{atmk}(t_{i+1}) - \underline{\mathbf{x}}_{dyn}(adaptive)^{(t_{i+1})}$ (4-9)

for the kth filter.

For very severe target maneuvers, the difference between the position estimates of the small field of view filter and those of the multiple model filter become so large that cylindrical shift of the image approaches a complete cycle. At this point, the small field of view filter will diverge. Because of the lower bounding on the

᠉ᡏᡧ᠈᠉ᠫ᠉᠋ᡗᢦᡸᢦᢄᡧᢄᡧᡬ᠕ᡬ᠕ᡬ᠖ᡬᡧᡬ᠕ᡬ᠖ᡸᡧᢞ᠘ᠻ᠘ᢉᡬᡭᠻᡭᡭᡭᡭᡭᡭᡭᡭᡭᡭᡭᡭᡭᡭᡭᡭᢤᡭ᠕ᡬᡧ᠘ᢣ᠘ᢢ᠘ᡀ᠔ᢣ᠘ᢣ᠘ᢣ᠘ᢣ᠘ᢣ᠘ᢣ᠘ᢣ᠘ᢣ᠘ᢣ᠘

conditional probabilities of each of the filters, this will eventually cause the whole algorithm to diverge. To avoid this condition, the states of the small field of view filter were set to the appropriately rescaled values of the large field of view filter when the magnitude of the shift exceeded 3.0 pixels in magnitude. Its error covariances were set at levels so the filter would undergo another acquisition cycle. The conditional probabilities were kept at their current values to indicate low confidence in the small field of view filter's state estimates.

4.3.2 Implementation in the Millner Tracker Configuration. In the Millner tracker model the rescaling needed to accommodate the larger field of view was accomplished within the correlator, and was transparent to the linear Kalman filter. Recall from Chapter I that the correlator produces "measurements" for the Kalman filter by calculating the offsets from the center of the field of view. Since the correlator provides these estimates in units of pixels, the output of the correlation algorithm for the larger field of view was multiplied by a factor of three prior to being provided to the associated Kalman filter.

Unlike the implementation of the multiple model algorithm in the Kozemchak tracker model, the conditional probabilities can be calculated exactly as outlined in Equations (4-2) and (4-3). This is because <u>A</u> is now a 2 x 2 matrix since the measurements to the Kalman filters are offsets from the center of the tracking window.

IV-14

a faith a start and a start a faith a f
4.4 Summary

This chapter described the reasons for adopting the multiple model filter algorithm and the structure of the algorithm. It also explained how this algorithm was implemented with the existing tracker configurations.

### 5.1 Introduction

This chapter presents the tracking performance of the two tracker formulations against the target trajectories descirbed in Chapter II, after incorporation of the multiple model filter algorithm. The first section of the chapter discusses the figures of merit used to evaluate tracker performance. The next section describes the performance plots generated by each set of computer simulations. The third section of the chapter lists the parameter values assigned to the truth and filter models used in this research. These values were chosen on the basis that they provided the best model of true target behavior and/or they were shown to result in best tracker performance in the studies by Kozemchak (5) and Millner (13). The final section of this chapter discusses the results of the computer simulations using both the figures of merit and the performance plots.

### 5.2 Derivation of Tracker Statistics

Statistics on tracker performance were gathered using a Monte Carlo analysis technique. Previous studies by Harnly and Jensen (3) and Flynn (2) have shown that a total of 10 Monte Carlo runs will exhibit reasonable convergence of the error statistics to the actual error statistics of an infinite number of runs. Each Monte Carlo run simulated 5 seconds in real time for a total of 150 frames of data at the 30 Hz sampling rate.

In the tracking problem, one of the quantities of interest is the error committed when estimating the target's true position due to its own dynamics. This error reflects how well the internal filter dynamics model performs against a target performing a wide variety of manuevers. Statistics were also kept on the accuracy of the tracker's estimates of the target centroid. These statistics are of interest because errors in the estimation of the target centroid will affect the accuracy of the estimated target intensity shape function in the extended Kalman filter tracker, and the target reference image in the linear Kalman filter/correlator tracker. Errors in the estimated intensity function are important because this function is used when updating filter estimates each time new information is received from the FLIR. Errors in the target reference image will affect the correlation process in the linear Kalman filter/correlator tracker. This will produce less accurate offset quantities (the pseudo-measurements for the linear Kalman filter), thereby affecting the accuracy of the tracker.

All of the above statistics were kept for instances before and after measurement incorporation. By comparing the errors committed before and after the filter estimates are updated, it is possible to evaluate how well estimates of target position are improved each time information from a frame of data is received.

The sample mean errors of the filter state estimates are calculated as follows:

$$\overline{E_{xd}}(t_i) = 1/N \sum_{k=1}^{N} [x_{dk}(t_i) - \hat{x}_{dfk}(t_i)] = 1/N \sum_{k=1}^{N} e_{xdk}(t_i)$$
(5-1)

- where  $\overline{E_{xd}}(t_i)$  = sample mean error (i.e. ensemble average error over all simulations) in x-dynamics position at time  $t_i$ 

  - $x_{dk}(t_i)$  = truth model x-dynamics value at time  $t_i$  for simulation k
  - e<sub>xdk</sub>(t<sub>i</sub>) = error in x-dynamics position at time t<sub>i</sub>
    for simulation k

N = number of Monte Carlo runs

and the sample variance of the error is given by:

$$\sigma_{xd}^{2}(t_{i}) = 1/(N-1) \sum_{k=1}^{N} e_{xdk}^{2}(t_{i}) - [N/(N-1)] \overline{E_{xd}^{2}}(t_{i})$$
(5-2)

where the quantities are as defined above. The two equations, (5-1) and (5-2), may be generalized to perform sample mean error and variance calculations for the errors committed when estimating the y-dynamics position, and the x- and y- centroid location coordinates. These errors are expressed in FLIR image plane coordinates and describe offset from the center of the sensor field of view. The units of the errors are pixels, with each pixel being 20 rads in length.

**V-3** 

In addition to averaging the errors over all Monte Carlo runs, time averaging was performed on the mean errors and standard deviations. This temporal averaging was conducted from t=0.5 seconds to t=2.0 seconds for trajectory 1 evaluations, and from t=3.5 seconds to t=5.0 seconds for all other trajectories. The earlier time frame for trajectory 1 evaluations was selected because it allowed the filter transients to die out while avoiding the minimum range/maximum passing rate condition that occurs near the end of the simulation. In this condition, any filter tuned for a benign target trajectory begins to exhibit markedly degraded performance. The later time frame chosen for the other trajectories allows the filter transients due to manuevers initiated at t=2.0 seconds to die out completely before time averaging begins.

This time averaging allows presentation of the data in a compact, tabular form. However, temporal averaging can also result in misleading figures of merit. For instance, if the errors should follow a ramp function from negative to positive values over the period of temporal averaging, a misleading sample mean error of approximately zero will be the result. This figure could lead to an erroneous assumption that an unbiased estimate was being generated by the filter. Therefore, care should be taken before making sweeping generalizations based only on the figures resented in the tables. If this reason, the performance results of the simulations is eincluded in Appendices A, B, and C.

These plots are grouped according to trajectory type and according to severity of the maneuver within those The performance plots for trajectory 1 evaluations groups. are followed by those for trajectory 2, trajectory 3, and son on. Within those groups, the plots for a 2g pull-up maneuver are followed by those for the 10g pull-up maneuver, and then the 20g maneuver. Still further, the plots of the multiple model filter are followed by those of the small field of view filter, and then those of the large field of view filter. Appendix A contains the plots for the linear Kalman filter/correlator tracker. Appendix B contains the plots for the extended Kalman filter tracker using the Gauss-Markov target acceleration model. Appendix C contains the plots for the same tracker as Appendix B, but these results are for the constant turn rate target acceleration model.

### 5.3 Performance Plots

As mentioned in the previous section, performance plots of the simulations are generated to prevent possible misinterpretations of the results caused by the temporal averaging of the statistics. These performance plots are of the x- and y- dynamics mean errors, and the x- and y- centroid mean errors; plus and minus the standard deviation of the respective errors. Because of the large number of cases run during this research, only the x- and y- dynamics mean error (plus and minus one standard deviation) plots at both before and after measurement incorporation are included in this document. This is due to the fact that: (1) estima true target states  $x_d$  and  $y_d$  are of primary importantracking; and (2) it is in fact easier to estimat position of the apparent target centroid than to ide the individual components of Equation (2-1). The num plots was further pared down by including only those which illustrated important trends. For example, most involving target maneuver will have plots only for t axis, which was the direction of the maneuver.

Examples of the types of performance plots gene are Figures V-1 and V-2. These are plots of the y-dy mean error plus and minus one standard deviation. these figures, it is evident that the maneuver was ini at t=2.0 seconds. At that point, a dramatic increa mean tracker error is exhibited. It can also be see it takes the filter some finite amount of time to re-Figure V-1 is a plot of the y-dynamics error statist the time prior to measurement incorporation, or the minus" error. Figure V-2 is a plot of the same qua but at the time after measurement incorporation, or t plus" error. To illustrate how well the filter improv target position estimates each time measurement infor is received from the FLIR, note the mean peak erre Figures V-1 and V-2. The mean peak error is approxim -7.0 pixels for the minus time, and is only -3.8 pixe the plus time. Pixels are defined as before, with pixel measuring 20  $\mu$ rad in length. The values of th

V-6



a a contrata de la co



error and standard deviation are generated by applying Equations (5-1) and (5-2) to the data at each sample time for each Monte Carlo run.

### 5.4 Parameter Values

In order to allow direct comparison of the results obtained for the extended Kalman filter tracker and the linear Kalman filter/correlator tracker, both trackers were presented with as identical as possible truth model representations of the target. Therefore, the truth models of both the extended Kalman filter and the linear Kalman filter/correlator trackers were set so they possessed identical descriptions of target shape and intensity, as well as identical models for background and FLIR measurement noises, atmospheric jitter, etc. On the other hand, the parameter values used in the filter models for the respective trackers will vary somewhat with the tuning done for each tracker.

5.4.1 Truth Model Parameters. For all simulations, the initial inertial parameters of the target in the inertial reference frame were:

Inertial position: x = 5000 m500 m 20000 m -1000 m/s Inertial velocity:  $v_X$ Inertial acceleration: a<sub>x</sub>

V-9

All trajectories described in Chapter II start from these initial conditions. The input measurement variance, which includes both the background and FLIR noises, was set to a value of one. This parameter value was expressed in terms of (intensity)<sup>2</sup> units, which are arbitrarily chosen units used to indicate the strength of the image received by the FLIR. The maximum intensity of each target hot-spot was given a value of 20. The resulting signal-to-noise (SNR) of all simulations is 20, which is defined by the following relationship for this application:

(5-3)

This value for the SNR is representative of realistic tracking scenarios (3). The variance of the atmospheric jitter was set to 0.2 pixels<sup>2</sup>. This value may be somewhat low compared to the true level of atmospheric jitter in the real world so it may be advisable to investigate the sensitivity of the algorithm to different levels of atmospheric jitter in future studies. Finally, the multiple hot -spot target were defined using circular intensity contours with a glint dispersion parameter of 2.0 pixels<sup>2</sup>.

5.4.2 Data Processing Parameters. In the data processing algorithms of both tracker types, a number of parameters could be varied to alter tracker performance. One such parameter determines the nature of the padding done to the 8 x 8 tracking window discussed in Chapter I. Recall that a 24 x 24 array is processed instead of the 8 x 8 field of view to generate the FFT's. A common engineering practice would be to pad the field of view by surrounding it with a border of 8 rows and columns of zeros. However, as stated earlier, this would introduce artificial edge effects if the target intensity function is not very close to zero at the borders of the 8 x 8 tracking window. Since measurement data of the regions outside the tracking window was available, there was the luxury of being able to pad the tracking window of the smaller field of view with real, but noise-corrupted data, instead of zeros. On the other hand, the increased size of the larger field of view motivated padded it with zeros at all times since it is better to pad with zeros versus the noise value, if the signal has gone to zero at the borders of the tracking window.

High frequency spatial frequency filtering could also be performed for both tracker types when deriving the target intensity profile. This can be accomplished by zeroing out a specified number of high frequency components within the Fourier transform of the image. For a target whose intensity is slowly changing in the spatial domain, this type of filtering will generally enhance tracker performance. Conversely, for a rapidly changing target intensity function, this type of filtering will produce errors in the target reference image and degrade tracker performance (13:106). In both tracker configurations, it was inappropriate to perform high frequency filtering for the larger field of view since, with pixel scaling, the large field of view signal spatial frequency content goes to three times the highest frequency as seen in the original field of view.

Finally, the relative weighting parameter used in performing exponential smoothing can be varied to account for a rapidly changing target intensity profile. This parameter was set to 0.1 for both tracker formulations since it was found to yield best performance in previous studies (5,13). This indicates that the target intensity function is expected to be slowly varying relative to the 30 Hz sampling rate.

5.4.3 Filter Parameter Values. As stated in Chapter IV, each filter in both tracker types was tuned for optimum performance at a specified degree of target motion. The smaller field of view filters were tuned to achieve best performance for constant velocity trajectories. The large field of view filters were tuned for best performance for the 20g pull-up maneuver. During this research, it was found that explicitly tuning the larger field of view for a 20g pull-up manuever made it incapable of maintaining lock on a constant velocity target. Therefore, this filter was tuned for best performance for a 20g pull-up maneuver while still being able to track a constant velocity target. Because this study was intended to be a feasibility study for the multiple model adaptive filter algorithm, no investigations were made to determine the increase in rms error caused by the need to maintain lock on a constant velocity

target. In a tracking scenario, this philosophy makes sense since the wider aperture would also be used for target acquisition, where the targets would be at long-range and exhibiting very benign behavior. The performance achieved from the individual filters under these tuning conditions will be presented in Table V-1 and associated figures.

For all filters, atmospheric jitter was modelled with an assumed correlation time of 0.07 seconds. This value was chosen based on the results of previous research efforts (3,12,14).

For the extended Kalman filter tracker, the values for the parameters for the smaller field of view filter were selected on the basis of those values which gave best performance for the benign target trajectories in Kozemchak's (5) study. A filter time correlation constant for filter acceleration dynamics, using the first-order Gauss-Markov model, was set to 1.5 seconds, with the assumed target acceleration white noise strength set to  $300 \text{ pixels}^2/\text{sec}^5$ . As before, pixels are defined to be 20  $\mu$ rads in length. Except where explicitly stated, this definition of a pixel will hold at all times. Parameters for the larger field of view will also be expressed in these pixel units even though the actual picture elements of the larger field of view measure 60  $\mu$ rads in length. On the basis of tuning runs performed, the time correlation constant for filter acceleration dynamics was set to 1.3 seconds for the Gauss-Markov model. The target acceleration white noise strength was set to 5000 pixels<sup>2</sup>/sec<sup>5</sup> for both the constant turn rate and Gauss-Markov target acceleration models for the larger field of view filter. The similarity of the values for the two disimilar acceleration models is due in part to the need to maintain lock on a constant velocity target.

For the linear Kalman filter/correlator tracker, Millner's (13) results indicated that a dynamic correlation time of 3.5 seconds and an assumed target acceleration white noise strength of 150 pixels<sup>2</sup>/sec<sup>5</sup> achieved best performance for the small field of view filter. The large difference in the dynamics correlation time for the two tracker types indicates that perhaps the larger correlation time for the linear Kalman filter/correlator tracker could be reduced to be more consistent with the value of the same parameter for the extended Kalman filter tracker. This was not investigated during this research and should be explored in any future study. Test runs of the larger field of view filter indicated it performed best for the 20g pull-up maneuver when these parameters were set to 1.5 seconds and 2000 pixels<sup>2</sup>/sec<sup>5</sup> respectively.

Before presenting the performance capabilities of the individual filters at their tuned-for conditions, a narrative on the organization and presentation of the information would be helpful. First, each case, or simulation is uniquely identified using a mnemonic code. This code is described in Figure V-3. The figures of merit presented in the tables are defined as:

- $\bar{x}_e$  = time-average of the mean error for the true position in the x-direction from time t=3.5 seconds to t=5.0 seconds (t=0.5 seconds to t=2.0 seconds for trajectory 1 targets) at time minus and plus (similarly for  $\bar{y}_e$ )
- $\overline{\sigma x}_e$  = time-average of the standard deviation of the error for the true position in the x-direction from the time t=3.5 seconds to t=5.0 seconds (t=0.5 seconds to t=2.0 seconds for trajectory 1 targets) at times minus and plus (similarly for  $\overline{\sigma y}_e$ )
  - $\overline{cx}_e$  and  $\overline{\sigma cx}_e$  = errors as degined above for the centroid position (similarly for  $cy_e$  and  $cy_e$ )

The performance capabilities of the respective filters are presented in Table V-1. The first three cases presented in the table are those of the small field of view filters. Recall that this filter was tuned for the constant velocity trajectory. The performance plots of the x-dynamics error for these cases are Figures A-2, B-2, and C-2, respectively.

The Millner linear filter/enhanced correlator tracker exhibits lower sample mean errors but with larger standard deviations. From the performance plots for the extended Kalman filter tracker cases, the poor tuning of the small field of view filter is clearly evident. The estimates show a drifting behavior that indicates that the filter dynamic driving noise was set too low. Unfortunately, this poor tuning condition (based on tuning values established in reference [5]) was not noticed until late in this research. At that point, while it would have still been possible to perform a total retuning of the filter, for this feasibility study, there was some concern that increasing the level of the filter dynamics driving noise would reduce the differ-

-Xer

# INDIVIDUAL FILTER EVALUATIONS

| CASE    | <u>x</u> (-)/               | <u>x</u> (+)/                   | <u>cx</u> e (-)/ | <u>cx</u> (+)/ | <u>y</u> e (-)/ | <u>y</u> e(+)/  | <u>cy</u> e (-)  | $\overline{cy}_{e}(+)$       |
|---------|-----------------------------|---------------------------------|------------------|----------------|-----------------|-----------------|------------------|------------------------------|
|         | <b>o</b> x <sub>e</sub> (-) | <del>o</del> x <sub>e</sub> (+) | <u>(-)</u>       | 0 CX e (+)     | <u>oy</u> e (-) | <u>oy</u> e (+) | <u>ocy</u> e (-) | <b>o</b> cy <sub>e</sub> (+) |
| LTIFI   | .046/<br>.180               | .064/<br>.160                   | .080/<br>.177    | .108/<br>.062  | 015/<br>.170    | 020/<br>.152    | 012/<br>.181     | 020/<br>.057                 |
| GTIFI   | 236/                        | 195/<br>.124                    | 180/<br>.133     | 107/<br>.079   | 266/<br>.118    | 270/<br>.106    | 263/<br>.126     | 271/<br>.053                 |
| CTIF1   | .208/<br>.141               | .184/<br>.129                   | .185<br>.138     | .145/<br>.084  | 209/<br>.123    | 210/<br>.112    | 203/<br>.131     | 206/<br>.067                 |
| L20T2F2 | .583/<br>.237               | .355/<br>.193                   | .386/<br>.239    | 053/<br>.113   | 264/<br>.250    | .036/<br>.214   | 040/<br>.232     | .398/<br>.129                |
| G20T2F2 | .354/<br>.198               | .186/                           | .241/.171        | .015/          | 884/<br>.204    | 674/<br>.181    | 766/<br>.186     | 484/<br>.127                 |
| C20T2F2 | .084/<br>.203               | .073/                           | .060/<br>.174    | .045/<br>.087  | 117/<br>.194    | 123/<br>.162    | 127/<br>.175     | 135/<br>.097                 |

V-16

Each simulation is uniquely identified with a mnemonic code. This code will consist of up to eight characters and describes the tracker formulation used, target acceleration model, trajectory type, and whether the multiple model, the small field of view filter, or the large field of view filter was used in the simulation.

The code generally follows this pattern:

MF - multiple model adaptive filter = F1 - small field of view filter only = F2 - large field of view filter only = T1 - trajectory 1 = T2 - trajectory 2 = T3 - trajectory 3 = T4 - trajectory 4 g level of maneuver (i.e. 10 g's) = L - linear Kalman filter/correlator tracker, Gauss-Markov target acceleration model = G - extended Kalman filter tracker, Gauss-Markov target acceleration model

= C - extended Kalman filter tracker, constant turn rate target acceleration model

### Special initial codes:

en de la calenda de la cale

L 10 T2 MF

| LA<br>GA<br>CA | tracker with ad hoc changes to conditional probabilites |
|----------------|---------------------------------------------------------|
| LS             | linear Kalman filter/correlator tracker, SNR = 10       |

Figure V-3. Simulation Mnemonic Codes

ences between the large and small field of view filter models, thus making it difficult for the multiple model adaptive filter algorithm to select the correct filter model to match current target behavior. This lack of significant differences hampered the early study on the multiple model filter algorithm by Flynn(2). Due to this poor tuning of the smaller field of view filter for the extended Kalman filter tracker, the Millner tracker formulation also exhibits smaller sample rms errors for the benign trajectory. However, in a laser weapon system, it may be more important to minimize the area painted by the laser so it may be more desirable to have smaller standard deviations than small offset errors; i.e. small standard deviations may be more essential than small mean errors. One final characteristic evident in these performance plots is the increase in the mean errors during the last 0.5 seconds of each simulation. This increase is due to the minimum range/maximum passing rate condition alluded to earlier in this chapter.

The second group of cases cited in Table V-1 are those of the large field of view filters for the 20g pull-up maneuver. The performance plots for these cases are Figure A-10, B-10, and C-10. Only the y-dynamics error plots are presented since the maneuver took place in that direction. As before, the extended Kalman filter tracker has much smaller sample standard deviations. However, that tracker formulation, using the contant turn rate acceleration model, demonstrates far superior performance. As shown in the

performance plots, its peak mean errors are smaller and the bias of the estimate once the filter transients have settled down is very small. The performance plots of the Millner tracker formulation (Figure A-10) show similar mean peak errors with and unbiased estimate after the transients have died out. But comparing the length of time it takes the filter to recover from the manuever shows that the extended Kalman filter tracker using the constant turn rate acceleration model has a much shorter transient time. Figure B-10, which is of the same tracker formulation but with the Gauss-Markov model shows the same good transient behavior but with larger peak mean errors. The figure also graphically illustrates the large bias that appears in Table V-1. This large rms error indicates the inadequacies of using the firstorder Gauss-Markov target acceleration model for tracking highly dynamic targets.

### 5.5 Tracker Performance Against Target Trajectories

Both multiple model tracker formulations were evaluated against the target trajectories described in Chapter II. For convenience, they are summarized here. Trajectory 1 is a constant velocity trajectory which maintains the initial velocity throughout the simulation. Trajectory 2 is a constant g pull-up maneuver. It begins the same as trajectory 1, but the target initiates a constant-speed, constant-g pull-up maneuver two seconds into the simulation and maintains this maneuver until the end of the simulation. The cases studied included 2, 10, and 20g pull-up maneuvers.

Trajectory 3 is similar to trajectory 2, but inste continuing the pull-up maneuver until the end of the s tion, it terminates the pull-up maneuver 3.5 second the simulation. At this point, the inertial velocity existing at that time are propagated until the end simulation at 5 seconds. Trajectory 4 is begun a previous maneuvers, but instead of performing a pu maneuver, it executes a constant-g turn toward the t until the end of the simulation.

5.5.1 Evaluations Using Trajectory 1. For this t tory, Monte Carlo runs were performed to evaluate to performance using both filters in the multiple model tive algorithm; only the small field of view filte: only the large field of view filter. Recall from Chap that a claim was made that simply enlarging the to field of view for all applications would result in p tracker performance at benign trajectories relative t achieved using the small field of view filter.

Table V-2 presents the results of these simul: using the multiple model adaptive algorithm and on large field of view filter for both tracker types. Re that the results of the simulations using only the field of view filter were presented in Table V-1. corresponding performance plots are Figures A-1 and Aand B-3; and C-1 and C-3.

The first two cases in the table are for the Mi tracker formulation. The first of these is for the mu

model adpative algorithm, the second is for the larg of view filter only. As expected, the larger field filter, which was tuned to be able to handle a 20g maneuver, performed poorly against a benign target tory relative to the original filter presented in Ta which was expressly tuned for good performance agai type of target behavior. Comparing the figures of n the multiple model adaptive filter and those of t field of view filter reveals values that are very indicating that the multiple model algorithm is c weighting the estimates of the small field of vie heavily, thus maintaining the desired high resol benign target trajectories. This heavy weightin small field of view filter also manifests itsel similarity of the performance plots for the multip and small field of view filters (Figures A-1 and A-: tively).

Results for the extended Kalman filter track rithm are similar though less conclusive due to tuning of the small field of view filter. In this ( larger field of view filter exhibits much lower san and rms errors than the small field of view filte ever, its standard deviations are much higher. The acteristics are evident in the performance plots, F. 3 and C-3. It should also be noted that the multip filter figures of merit resemble those of the small view filter (as with the linear Kalman filter/co

**V-20** 

TRAJECTORY 1 EVALUATIONS

| CASE  | $\overline{x}_{e}(-)/$ | $\overline{\mathbf{x}}_{e}(+)/$ | <u>cx</u> e(-)/  | <u>cx</u> e (+)      | <u>y</u> e (-)/ | $\overline{Y}_{e}(+)/$ | <u>cy</u> e (-)/ | <u>cy</u> e (+) /            |
|-------|------------------------|---------------------------------|------------------|----------------------|-----------------|------------------------|------------------|------------------------------|
|       | <u>ox</u> e (-)        | <b>6</b> X <sub>e</sub> (+)     | <u>ocx</u> e (-) | σcx <sub>e</sub> (+) | <u> </u>        | <u>ر</u> +(+)          | <u> </u>         | <b>o</b> cy <sub>e</sub> (+) |
| LTIMF | .052/<br>.194          | .067/<br>.172                   | .082/<br>.179    | .110<br>.063         | 001/<br>.063    | 005/                   | .003/<br>.188    | 004/<br>.078                 |
| LT1F2 | .098/<br>.297          | .120/<br>.245                   | .125/<br>.295    | .157/<br>.218        | 017/<br>.289    | 020/<br>.246           | 011/<br>.299     | 014/<br>.228                 |
| GTIMF | 223/<br>.134           | 185/<br>.121                    | 170/<br>.129     | 101/<br>.069         | 241/<br>.114    | 246/<br>.102           | 239/<br>.122     | 247/<br>.046                 |
| GT1F2 | 006/<br>.191           | .008/                           | .008/<br>.168    | .028/<br>.092        | .015/           | .013/<br>.128          | .020/<br>.158    | .018<br>.075                 |
| CTIMF | .167/<br>.141          | .145/<br>.128                   | .147/<br>.135    | .110/<br>.076        | 208/<br>.121    | 210/<br>.109           | 202/<br>.127     | 205/<br>.058                 |
| CT1F2 | .037/<br>.194          | .026/<br>.156                   | .036/            | .022/                | .013/<br>.166   | .012/<br>.129          | .019/<br>.161    | .018/                        |

<u>.</u>....

tracker) despite the fact the larger field of view filter has smaller rms error values. This characteristic is confirmed by comparing Figures B-1 and B-2, and Figures C-1 and C-3. This indicates that the multiple model algorithm was still able to choose the correct filter model even in this pporly tuned case. those of the larger field of view filter even though it has smaller sample rms error values. This indicates that the multiple model algorithm was able to choose the correct filter model even in this poorly tuned case.

5.5.2 Evaluations Using Trajectory 2. Tables V-3, V-4, and V-5 present the results of the simulations for 2, 10, and 20g constant speed, constant-g pull-up maneuvers respectively. The corresponding performance plots are Figures A-4 thru A-10, B-4 thru B-10, and C-4 thru C-10.

Table V-3 presents the results for a 2g pull-up maneuver. Along with the results for the multiple model adaptive filter algorithm, are those for the individual filters. Generally speaking, the linear Kalman filter/correlator tracker produces lower sample mean errors with much larger standard deviations than the extended Kalman filter tracker configuration. In this instance, the sample rms errors of the multiple model filter in the Millner tracker formulation are also much smaller. This is again largely due to the poor tuning of the smaller field of view filter for the Kozemchak tracker formulation. As seen in the performance plots and the figures of merit, the performance of the

£

¢

### **TABLE V-3**

<u> Maria Maria Maria Maria Maria Maria M</u>aria Maria Mar

## TRAJECTORY 2 EVALUATION (2G)

| CASE   | x (-)/          | <u>x</u> (+)/               | <u>cx</u> e (-)/ | <u>cx</u> (+)/       | 7 <sub>e</sub> (-)/ | $\overline{y}_{e}^{(+)/}$ | <u>cy</u> e (-)/              | <u>cy</u> e(+)/ |
|--------|-----------------|-----------------------------|------------------|----------------------|---------------------|---------------------------|-------------------------------|-----------------|
|        | <u>ox</u> e (-) | <b>J</b> x <sub>e</sub> (+) | <u>(-)</u>       | σcX <sub>e</sub> (+) | <del>øy</del> e (-) | <u> </u>                  | $\overline{\sigma cy}_{e}(-)$ | <u> </u>        |
| L2T2MF | .226/<br>.182   | .197/<br>.164               | .150/<br>.189    | -089/<br>-069        | 157/<br>.225        | 118/<br>.209              | 099/<br>.228                  | 020/<br>.151    |
| L2T2F1 | .230/           | .197/<br>.159               | .146/<br>.185    | .078/<br>.059        | 187/<br>.166        | 146/<br>.152              | 123/<br>.180                  | 038/<br>.069    |
| L2T2F2 | .209/<br>.303   | .203/<br>.251               | .180/<br>.307    | .171/<br>.218        | .118/<br>.317       | .152/<br>.274             | .135/<br>.326                 | .185/<br>.248   |
| G2T2MF | .175/           | .165/                       | .146/<br>.151    | .129<br>.099         | 802/<br>.129        | 737/<br>.119              | 725/<br>.135                  | 609/<br>.083    |
| G2T2F1 | .198/           | .189/<br>.112               | .169/<br>.154    | .152/<br>.103        | 796/<br>.129        | 731/<br>.119              | 718/<br>.136                  | 601/<br>.086    |
| G2T2F2 | .051/<br>.196   | .046/<br>.159               | .030/<br>.171    | .024/<br>.083        | 006/<br>.174        | .017/<br>.139             | .001/<br>.164                 | .033/<br>.076   |
| C2T2MF | .409/<br>.128   | .384/                       | .359/<br>.157    | .315/<br>.105        | 435/<br>.145        | 432/<br>.134              | 439/<br>.149                  | 433/<br>.105    |
| C2T2F1 | .436/           | .410/<br>.116               | .386/<br>.158    | .341/<br>.108        | 422/<br>.149        | 419/<br>.138              | 427/<br>.153                  | 421/<br>.111    |
| C2T2F2 | .070/<br>.201   | .060/<br>.160               | .047/<br>.173    | .034/<br>.085        | .039/<br>.178       | .039/                     | .033/<br>.167                 | .032/<br>.077   |

multiple model adpative filter more closely resembles that of the small field of view filter since its constant velocity model more closely resembles the 2g pull-up maneuver than does the 20g pull-up maneuver model. The performance plots also show that the bias for the extended Kalman filter tracker using the Gauss-Markov model is very large compared to the biases for the other cases. Finally, the performance plots for the large field of view filter (Figures A-6, B-6, and C-6 respectively) show that, because of the large filter target acceleration white noise strength, the maneuver is not as evident as in the plots of the small field of view filter.

Tables V-4 and V-5 respectively, present the results for 10 and 20g pull-up maneuvers. The performance plots for these cases are Figures A-7 thru A-10, B-7 thru B-10, and C-7 thru C-10. Only plots for the y-axis errors are included since that is the direction of target motion. Furthermore, only cases using the multiple model filter algorithm and only the large field of view filter are presented here as the small field of view filter was completely unable to maintain lock on these targets. For these highly dynamic cases, the extended Kalman filter tracker exhibits much smaller sample rms errors than the linear Kalman filter/correlator tracker. For both tracker types the multiple model filter has slightly worse figures of merit than those of the large field of view filter by itself. This is due to the lower bounding of the conditional probabilities discussed in

TRAJECTORY 2 EVALUATIONS (10G)

| 6  | <u>x</u> (-)/       | <u>x</u> (+)/   | <u>cx</u> e (-) / | <u>cx</u> e (+)/ | <u>y</u> e (-)/                        | <u>y</u> e (+)/ | <u>cy</u> e (-)/ | <u>cy</u> e (+)/ |
|----|---------------------|-----------------|-------------------|------------------|----------------------------------------|-----------------|------------------|------------------|
|    | σx <sub>e</sub> (-) | <u>ox</u> e (+) | 0 -)              | <u>øcx</u> e (+) | <u><math> \sigma y_{e} (-) </math></u> | <u> </u>        | <u>ocy</u> e (-) | <u> (+)</u>      |
| Č4 | .338/<br>.458       | .269/<br>.432   | .250/<br>.466     | .144/<br>.419    | 089/<br>.747                           | .084/<br>.740   | .046/<br>.729    | .307/<br>.706    |
| 1  | .148/<br>.282       | .085/<br>.232   | .077/<br>.271     | 014/<br>.177     | .337/<br>.308                          | .508/           | .461/<br>.298    | .711/<br>.234    |
|    | .131/<br>.197       | .084/<br>.160   | .086/             | .023<br>.086     | 232/<br>.171                           | 107/<br>.138    | 164/             | .073             |
| 2  | .129/<br>.196       | .083/<br>.159   | .085/<br>.171     | .023/<br>.084    | 175/<br>.172                           | 052/<br>.138    | .111/            | .055/            |
|    | .109/               | .097/<br>.147   | .078/<br>.155     | .058/<br>.063    | 203/<br>.167                           | 221/<br>.139    | 211/<br>.158     | 232/<br>.081     |
| 2  | .201                | .160            | .050/<br>.173     | .037/<br>.085    | .055/<br>.178                          | .052/           | .047/<br>.167    | .044/<br>.078    |

Ø

ł.

TRAJECTORY 2 EVALUATIONS (20G)

| CASE    | ¥ <sub>e</sub> (-)/ | ¥ <sub>e</sub> (+) /                  | <u>cx</u> e (-) / | <u>cx</u> e(+)/ | $\overline{y}_{e}(-)/$ | <u>Y</u> e(+)/  | <u>cy</u> e (-)/                         | <u>cy</u> e (+) /    |
|---------|---------------------|---------------------------------------|-------------------|-----------------|------------------------|-----------------|------------------------------------------|----------------------|
|         | <del>õx</del> e (-) | <u><del>o</del>x</u> <sub>e</sub> (+) | <u>ocx</u> e (-)  | <u> (+)</u>     | <u>oy</u> e (-)        | <u>øy</u> e (+) | <u> <del>(</del></u> cy <sub>e</sub> (-) | 6cy <sub>e</sub> (+) |
| L20T2MF | .537/<br>.441       | .310/<br>.421                         | .330/<br>.454     | 007/<br>.412    | 243/<br>.450           | 026/<br>.436    | 032/<br>.436                             | .367                 |
| L20T2F1 | .583/<br>.237       | .355/<br>.193                         | .386/<br>.239     | 053/            | 264/<br>.250           | .036/<br>.214   | 040/<br>.232                             | .398/<br>.129        |
| G20T2MF | .356/<br>.200       | .186/<br>.164                         | .243/             | .015/           | 923/<br>.181           | 710<br>.155     | 805/<br>.166                             | 519/<br>.097         |
| G20T2F2 | .354/<br>.198       | .186/                                 | .241/<br>.171     | .015/<br>.086   | 884/<br>.204           | 674<br>.181     | 766/<br>.186                             | 484/<br>.127         |
| C20T2MF | .116/<br>.187       | .104/                                 | .086/<br>.159     | .067/<br>.069   | 285/<br>.160           | 300/<br>.129    | 297/<br>.150                             | 316/<br>.064         |
| C20T2F2 | .084/<br>.203       | .073/<br>.162                         | .060/             | .045/<br>.087   | 117/<br>.194           | 123/<br>.162    | 127/<br>.175                             | 135/<br>.097         |

Chapter IV. This results in having at least part of the state estimates being based on the estimates of a filter with a totally inappropriate model for the current target behavior. The lower bounding could cause a problem if at any time either of the filters in the bank should diverge. During the course of this research, it was found that a higher setting of the lower bound allowed the multiple model filter to apply heavy weight to the correct filter much more quickly, at the expense of a heavier weight on a totally inappropriate model after the transients have died out.

From the performance plots of these cases, it can be seen that the extended Kalman filter tracker exhibits better transient behavior than the other tracker formulation. This includes a shorter transient in response to a step change in the target truth model acceleration and lower mean peak errors as well. This is due to the non-linear nature of the algorithm as well as the fact that the extended Kalman filter tracker operates directly with the raw FLIR data while the other tracker type receives only position offset information from the enhanced correlator.

In the Millner tracker formulation, the multiple model as well as the single filter cases performed more poorly for a 10g pull-up maneuver than the more severe 20g pull-up maneuver. This is because the large field of view filter was tuned for best performance for a 20g pull-up maneuver while no equivalent filter was tuned for a 10g pull-up maneuver. In other words, the 10g maneuver was a case for which no filter model existed. However, no such behavior was observed for the extended Kalman filter tracker using either target acceleration model. This could be due to the poor tuning of the small field of view filter or perhaps the nonlinearity of the filter itself. These issues should be investigated in future studies.

At the 20g pull-up maneuver, use of the constant turn rate target acceleration model results in superior performance over all other cases. This follows since this acceleration model has been shown to be a better model of realistic, highly-dynamic, target maneuvers than the Gauss-Markov acceleration model.

5.5.3 Evaluations Using Trajectory 3. Tables V-6, V-7, and V-8 present the results of the simulations for trajectory 3 targets performing 2, 10, and 20g pull-up and continuation maneuvers. The plots for these cases are Figures A-11 thru A-17, B-11 thru B-17, and C-11 thru C-17. As with trajectory 2 evaluations, the results of the multiple model adaptive filter algorithm and the individual filters are presented. All of the trends observed in the trajectory 2 evaluations are repeated here. The sample mean errors tend to be higher than those for the trajectory 2 evaluations since the period over which temporal averaging is performed includes the transients created by the termination of the pull-up maneuver; thus the plots are particularly important here for appropriate insights.

Once again, the performance plots shown are only of the

and the second

ASSAULTERED MARKENEL EVERY TURNER RECEIL RECEIL RECEIL REPORT RECEIL RECEIL

C

**X** ~

وریکر اور ا

TRAJECTORY 3 EVALUATIONS (2G)

| CASE   | $\overline{x}_{e}(-)/$                 | <u>x</u> (+)/                   | <u>cx</u> e (-)/ | <u>cx</u> e(+)/ | <u>Y</u> e(-)/ | <u>Y<sub>e</sub>(+)/</u>     | σy <sub>e</sub> (-) / | <del>در</del> (+) / |
|--------|----------------------------------------|---------------------------------|------------------|-----------------|----------------|------------------------------|-----------------------|---------------------|
|        | <u> <del>0</del>x</u> <sub>e</sub> (-) | <del>đ</del> x <sub>e</sub> (+) | <u>σcx</u> e (-) | 000 (+)         | <u> </u>       | $\overline{\sigma y}_{e}(+)$ | <u>ocy</u> e (-)      | <u> </u>            |
| L2T3MF | 428/<br>.459                           | 399/<br>.428                    | 434/<br>.436     | 392/<br>.379    | .049/<br>.588  | .032/<br>.582                | .027/<br>.586         | .001/               |
| L2T3F1 | 323/<br>.314                           | 260/<br>.273                    | 236/<br>.259     | 105/<br>.122    | .418/<br>.208  | .304/                        | .209/                 | 028/<br>.078        |
| L2T3F2 | 012/<br>.337                           | .009/<br>.28 <b>4</b>           | 022/<br>.311     | .009/           | .116/<br>.229  | .266                         | .299                  | .061/<br>.235       |
| G2T3MF | .084/                                  | .080/                           | .062/<br>.143    | .055/<br>.082   | 246/<br>.126   | 296/<br>.114                 | 316/<br>.130          | 406/<br>.070        |
| G2T3F1 | .102/<br>.122                          | .098/                           | .080/            | .073/           | 230/           | 28-/<br>.115                 | 300/                  | 390/<br>.074        |
| G2T3F2 | .047/<br>.197                          | .045/                           | .028/            | .025/<br>.084   | .068/<br>.174  | .139                         | .050/                 | .022/               |
| C2T3MF | .332/<br>.126                          | .309/                           | .285/<br>.149    | .245/<br>.058   | .043/<br>.133  | 018/<br>.122                 | 041/<br>.144          | 146/<br>.099        |
| C2T3F1 | .349/<br>.129                          | .326/<br>.113                   | .302/<br>.151    | .262/           | .075/          | .014/                        | 009/<br>.147          | 115/<br>.103        |
| C2T3F2 | .200                                   | .060/<br>.160                   | .046/<br>.173    | .340<br>.085    | .078/<br>.178  | .053/<br>.141                | .059/<br>.167         | .027/               |

Anter Service Color Color

۲

TRAJECTORY 3 EVALUATIONS (10G)

| CASE    | <u>x</u> e (-)/ | $\overline{\mathbf{x}}_{e}(+)/$ | <u>cx</u> e(-)/  | <u>cx</u> e (+)/ | $\overline{Y}_{e}(-)/$             | <u>Y</u> e(+)/ | <u>σy<sub>e</sub> (-) /</u> | टy <sub>e</sub> (+) / |
|---------|-----------------|---------------------------------|------------------|------------------|------------------------------------|----------------|-----------------------------|-----------------------|
|         | <u>ox</u> e (-) | <u>o</u> x <sub>e</sub> (+)     | <u>ocx</u> e (-) | <u>dcx</u> e (+) | <b>đ</b> <u>y</u> <sub>e</sub> (-) | <u> </u>       | <u>ocy</u> e (-)            | <u>ocy</u> e (+)      |
| LIOT3MF | 469/<br>.415    | 430/<br>.383                    | 473/<br>.384     | 419/<br>.326     | .043/<br>.419                      | 078/<br>.402   | 055/<br>.404                | 236/<br>.362          |
| L10T3F2 | 088/<br>.298    | 053/<br>.249                    | 091/<br>.272     | 039/<br>.180     | .337/<br>.281                      | .219/<br>.244  | .248/<br>.278               | .076/<br>.196         |
| GlOT3MF | .037/           | ,045/<br>.161                   | .022/<br>.170    | .034/<br>.084    | .145/<br>.171                      | .048/<br>.137  | 088/<br>.161                | 043/<br>.073          |
| G10T3F2 | .038/           | .047/<br>.159                   | .024/<br>.171    | .035/<br>.083    | .185/                              | .089/<br>.138  | .129/<br>.163               | 001/<br>.075          |
| CIOT3MF | .037/<br>.200   | .044/                           | .022/<br>.172    | .031/<br>.084    | .175                               | .096/<br>.138  | .146/<br>.163               | 013/<br>.073          |
| C10T3F2 | .044/<br>.201   | .052/                           | .029/<br>.173    | .039/<br>.084    | .237/                              | .118/          | .168/                       | .012/<br>.075         |

Õ

TRAJECTORY 3 EVALUATIONS (20G)

| CASE    | x <sub>e</sub> (-)/ | <u>x</u> (+)/   | <u>cx</u> e (-)/ | <u>cx</u> e (+)/     | $\overline{Y}_{e}(-)/$ | <u>Y</u> e (+)/              | <u>cy</u> e (-)/ | <u>cy</u> e (+)/             |
|---------|---------------------|-----------------|------------------|----------------------|------------------------|------------------------------|------------------|------------------------------|
|         | <u>ox</u> e (-)     | <u>ox</u> e (+) | <u>ocx</u> e (-) | ocx <sub>e</sub> (+) | <u>مy</u> و (-)        | $\overline{\sigma Y_{e}}(+)$ | <u>σcy</u> e (-) | <b>o</b> cy <sub>e</sub> (+) |
| L20T3MF | 508/<br>.419        | 432/<br>.388    | 485/<br>.397     | 375/<br>.333         | .724/<br>.352          | .488/<br>.328                | .538/<br>.345    | .188/<br>.289                |
| L20T3F2 | 100/<br>.280        | 025/<br>.227    | 078/<br>.257     | .032/<br>.159        | .807/<br>.268          | .576/<br>.228                | .639/<br>.277    | .301/<br>.184                |
| G20T3MF | 037/<br>.200        | .006/<br>.164   | 034/<br>.171     | .024/<br>.084        | .094/<br>.166          | 093/<br>.135                 | 012/<br>.155     | 265/<br>.069                 |
| G20T3F2 | 033/<br>.200        | .009/           | 031/<br>.172     | .026/<br>.084        | .080/                  | 106/<br>.148                 | 025/<br>.162     | 275/<br>.082                 |
| C20T3MF | 083/<br>.203        | 025/<br>.165    | 073/<br>.173     | .002/<br>.084        | .332/                  | .101/.143                    | .204/<br>.168    | 101/<br>.087                 |
| C20T3F2 | 071/<br>.202        | 014/<br>.164    | 060/             | .014/<br>.083        | .384/                  | .154/.165                    | .255/            | 048/<br>.105                 |

.

y-dynamics mean errors since that is the direction of the target maneuver. It is easy to see the termination of the maneuver at t=3.5 seconds. In the 2g case, the heavy weighting of the small field of view filter is still evident in comparisons of the figures of merit and the plots for the multiple model filter and the small field of view filter. The performance of the multiple model filter appears to be much worse than the performance of either of the individual filters in the Millner tracker formulation. The plots of this case, Figure A-11, show that while the times of the pull-up and continuation maneuver are clearly evident, the standard deviations become very large, especially after termination of the maneuver. During this time period, the multiple filter is having a difficult time selecting the correct filter model. The observed target behavior is that of a constant velocity target buth the approach of the minimum range/maximum passing rate condition indicates that the field of view should be larger to handle a more dynamic trajectory. No such behavior is noted in the plots for the extended Kalman filter tracker, where again the poor tuning of the small field of view filter may be hindering the filter from adaptively reducing the size of the effective tracking window. As before, the extended Kalman filter tracker, using the Gauss-Markov model is showing much larger biases. Finally, as with trajectory 2 evaluations, the high setting of the strength of the target acceleration white noise tends to mask out low g maneuvers.

5.5.4 Evaluations Using Trajectory 4. Table V sents the results of the simulations for trajectory gets performing 2, 10, and 20g constant speed turns ever, because early results indicated that perform the trackers did not vary greatly from the perfo observed in trajectory 2 evaluations, simulations we: performed for the linear Kalman filter/correlator t The reason for the similarity is that, even with target, the out-of-plane component of the target path to remain very small due to the short duration of th lation. At the same time, because the target turns the tracker during this maneuver, the target intensit function changes considerably over the simulatio This demonstrates the tracker's ability to mainta: estimates of a time varying shape function, and the tracking performance.

The performance plots of these cases are Figu. thru A-24. The plots of the x-dynamics mean error included for the 2g case to demonstrate that the majo the maneuver still takes place in the y- directio plots of the y-dynamics error statistics can be comp those for the trajectory 2 evaluation (Figures A-4 10) to see the similarity of tracker performance for types of trajectories.

5.5.5 Changes in Signal-to-Noise Ratio. As earlier in this chapter, the nominal SNR for all simu was set 20. To test the multiple filter's robustne



## TRAJECTORY 4 EVALUATIONS

| CASE            | <u>x</u> (-)/                   | <u>x</u> (+)/                   | <u>cx</u> (-)/   | <u>cx</u> (+)/                          | $\overline{Y}_{e}(-)/$             | <u>Y</u> e(+)/              | <u>cy</u> e (-)/ | <u>cy</u> (+)) |
|-----------------|---------------------------------|---------------------------------|------------------|-----------------------------------------|------------------------------------|-----------------------------|------------------|----------------|
|                 | <del>0</del> x <sub>e</sub> (-) | <del>o</del> x <sub>e</sub> (+) | <u>ocx</u> e (-) | $\overline{\sigma_{\rm CX}}_{\rm e}(+)$ | <b>o</b> <u>y</u> <sub>e</sub> (-) | <b>o</b> Y <sub>e</sub> (+) | <u>ocy</u> e (-) | <u>(+)</u>     |
| L2T4MF          | .208/                           | .183/<br>.172                   | .142/<br>.195    | .090/<br>.087                           | 172/<br>.343                       | 132/<br>.332                | 119/<br>.345     | 040/<br>.301   |
| L2T4F1          | .226/                           | .194/<br>.159                   | .145/<br>.184    | .079/<br>.057                           | 182/<br>.166                       | 142/<br>.152                | ~.118/<br>.181   | 033/<br>.069   |
| L2T4F2          | .202/<br>.301                   | .197/<br>.250                   | .174/<br>.302    | .166/<br>.214                           | .120/<br>.323                      | .154/<br>.280               | .137/<br>.330    | .187/<br>.253  |
| L10T4MF         | .369/<br>.459                   | .307/<br>.433                   | .271/<br>.460    | .165/<br>.408                           | 141/<br>.730                       | .019/<br>.719               | .004/<br>.695    | .257/<br>.656  |
| L10T4F2         | .154/<br>.295                   | .106/<br>.249                   | .092/<br>.290    | .022/<br>.204                           | .345/<br>.286                      | .501/<br>.254               | .458/<br>.280    | .687/<br>.206  |
| L20T4MF         | .386/<br>.418                   | .202/<br>.395                   | .214/<br>.428    | 057/<br>.381                            | .022/<br>.526                      | .201/                       | .159/<br>.518    | .419/<br>.488  |
| L20 <b>T4F2</b> | .359/<br>.264                   | .176/                           | .193/<br>.264    | 073/<br>.159                            | .014/<br>.256                      | .195/<br>.222               | .152/<br>.252    | .416/<br>.162  |

.
change in the expected SNR, a few test cases were performed using the Millner tracker formulation at a SNR=10. No retuning of the filter was done for this lower SNR. This value also corresponds to a reasonable SNR for a tracking scenario (3). The results are presented in Table V-10 for a trajectory 2 target performing 2, 10, and 20g pull-up maneu-The associated plots are Figures A-25 thru A-27. As vers. with trajecotry 4 evaluations, these plots can be compared to the original plots for the trajectory 2 evaluations (Figures A-4 thru A-10), to see the similarity in performance. While there is some increase in the tracker sample rms errors, it still does a reasonable job of tracking even the highly dynamic 20g target. Attempts to reduce the SNR further to a value of unity resulted in completely divergent filter behavior, as seen previously in the study by Harnly and Jenson (3) for single hot-spot targets.

5.5.6 Ad Hoc Changes to Filter Conditional Probabilities. The final test performed during this research was done to establish some upper bound on the performance of the multiple model filter's ability to switch from one filter model to the other when a step change occurs in the truth model description of the target. This was accomplished by artificially providing the filter with perfect knowledge of when a maneuver was initiated. The result is a step change in the conditional probabilities of the single filters to correspond with the step change in the truth model. Because the filter transients have already died out by the time temporal averaging of the statistics takes place, this ad hoc change in the conditional probabilities had little or no effect on the figures of merit. However, the plots (Figures A-28 and A-29; B-18 and B-19; and C-18 and C-19) show that, as compared to the trajectory 2 evaluation plots, there was little effect on the time it took the filter transients to die out after the maneuver was initiated. What was changed, was the mean peak error of each tracker type, which was reduced by a factory of two. Since this was the only significant deviation in the performances of the multiple filter and the ad hoc cases, it was decided that the multiple filter performs reasonably well as compared to this theoretical upper bound.

: **X** 

An entre presses and a free and a fact

TABLE V-10

Ö

1

. . .

1.

•

TRAJECTORY 2 EVALUATIONS (SNR=10)

| CASE     | <u>x</u> (-)/                   | ¥e (+)/                                | <u>cx</u> e (-)/ | <u>сх<sub>е</sub> (+) /</u>  | Y <sub>e</sub> (-)/ | $\overline{\mathbf{Y}}_{\mathbf{e}}^{(+)/}$ | टY <sub>e</sub> (-) / | σy <sub>e</sub> (+)/ |
|----------|---------------------------------|----------------------------------------|------------------|------------------------------|---------------------|---------------------------------------------|-----------------------|----------------------|
|          | <del>0</del> x <sub>e</sub> (-) | <u> <del>0</del>x</u> <sub>e</sub> (+) | <u>(-)</u>       | <b>o</b> cx <sub>e</sub> (+) | <u> </u>            | $\overline{\sigma Y}_{e}(+)$                | <u> (-)</u>           | ocy <sub>e</sub> (+) |
| LS2T2MF  | .256/<br>.182                   | .231/<br>.162                          | .187/<br>.197    | .135/<br>.085                | 139/<br>.331        | 098/<br>.314                                | 085/<br>.334          | 004/<br>.283         |
| LS10T2MF | .312/<br>.378                   | .244/<br>.353                          | .228/<br>.379    | .125/<br>.329                | .051/               | .224/                                       | .184/                 | .440/<br>.747        |
| LS20T2MF | .595/                           | .364/<br>.272                          | .390/            | .049/                        | 223/<br>.423        | .075/                                       | .003/                 | .441/<br>.393        |

## VI. Conclusions and Recommendations

The multiple model filter algorithm has been shown to perform well against a wide dynamic range of targets. While not capable of performing as well as the original tracker formulations against relatively benign target trajectories, it does offer the advantage of being able to maintain lock onto highly dynamic targets that the original trackers were unable to follow. This increase in tracker capability would seem to offset the additional memory storage requirements and computational burden imposed by the multiple model filter algorithm. The additional time required to implement the multiple model filter can be tempered by using parallel processing techniques to process each individual filter, instead of processing them sequentially, as was done in this research.

From an implementation standpoint, it is easier to incorporate the multiple model algorithm into the linear Kalman filter/correlator tracker formulation than into the extended Kalman filter tracker. The linearity of the filter as well as the low dimensionality of the filter measurement vector for the Millner tracker formulation made it possible to implement the multiple model algorithm without any of the approximations and ad hoc changes necessary to implement the same algorithm in the extended Kalman filter tracker. The high dimensionality of the filter measurement vector for the extended Kalman filter tracker also drastically increased the memory storage requirements and computational loading of

the multiple model filter. Using identical FLIR measurement algorithms for the computer simulations of both tracker types, the simulation for the multiple model filter for the extended Kalman filter tracker occupied twice as much memory space and took approximately 30 percent more time to execute than the same simulation incorporating the linear Kalman filter/correlator structure. For this feasibility study, no attempts were made to minimize the memory requirements of each simulation other than what would be the result of structured programming techniques. Therefore, the memory required to run either simulation is subject to change and the differences between the simulations may also change when more efficient programming techniques are applied.

The results of the computer simulations show that the extended Kalman filter tracker significantly outperforms the Millner tracker formulation when evaluating tracker performance against highly dynamic targets. Furthermore, the constant turn rate acceleration model is superior to the Gauss-Markov model as a model of true target behavior. Even at very benign trajectories, where the poor tuning of the small field of view filter was a problem, the Kozemchak tracker formulation possessed much smaller sample standard deviations of tracker errors. For a laser weapon system, this may be more desirable than the lower sample mean errors achieved by the Millner tracker formulation, as long as the rms errors of both tracker types are about the same. The filter transients and mean peak errors were also smaller for

the extended Kalman filter tracker. Overall, for this configuration of the multiple model adaptive algorithm, with two independent Kalman filters, the improved performance of the extended Kalman filter tracker is worth the additional memory storage and computational burden.

## Recommendations

Further study is recommended in order to investigate problems encountered or tasks not accomplished in this research. These areas include:

- Improved filter tuning of the small field of view
  filter for the extended Kalman filter tracker
- Addition of more filters in the bank of independent filters, to imclude additional target dynamics levels, different fields of view, etc.
- Implementation of the constant turn rate target acceleration model in the linear Kalman filter/correlator tracker
- Realistic changes to the truth model target dynamics. A more realistic model than the step change of the truth model at the inititation of a maneuver may show that the dynamic range achieved by the multiple model filter is not necessary for realistic tracking scenarios.

Test trackers against targets whose shape functions are rapidly changing. This includes targets performing roll maneuvers as well as non-realistic scenarios where the size, shape, intensity, and number of hot-spots is changing.

- Improved ways of handling the approximation made to implement the multiple model algorithm in the extended Kalman filter tracker
- \* Extending the length of the simulation to investigate the minimum range/maximum passing rate condition and the response of the multiple model adaptive filter to it
- Determine how changes in the sampling rate of the simulation affects tracker performance
- Determine the algorithm's sensitivity to different settings for the standard deviation of atmospheric jitter
- Determine the algorithm's sensitivity to the value of the lower bound on conditional probabilities (requiring filter retuning for best performance)
- Other robustness studies should be performed to determine the algorithm's sensitivity to target shape, separation of the hot-spots, etc.

## <u>Bibliography</u>

- Bedworth, David D. <u>Industrial Systems Planning, Analy-</u> sis, <u>Control</u>. New York: The Ronald Press Company, 1973.
- Flynn, Patrick M., "Alternative Dynamics Models and Multiple Model Filtering for a Short Range Tracker," M.S. Thesis, Air Force Institute of Technology, Wright-Patterson AFB, Ohio, December 1981.
- 3. Harnly, Douglas A. and Robert L. Jensen, "An Adaptive Distributed-Measurement Extended Kalman Filter for a Short Range Tracker," M.S. Thesis, Air Force Institute of Technology, Wright-Patterson AFB, Ohio, December 1979.
- 4. Hogge, C.B. and R.R. Butts, "Frequency Spectra for the Geometric Representation of Wavefront Distortions Due to Atmospheric Turbulence," IEEE Transactions on Antennas and Propagation, Vol. AP-24, No. 2, March 1976.
- 5. Kozemchak, Mark R., "Enhanced Image Tracking: Analysis of Two Acceleration Models in Tracking Multiple Hot-Spot Images," M.S. Thesis, Air Force Institute of Technology, Wright-Patterson AFB, Ohio, December 1982.
- 6. Maybeck, Peter S. <u>Stochastic Models</u>, <u>Estimation and</u> <u>Control Volume I</u>. New York: Academic Press Incorporated, 1979.
- 7. Maybeck, Peter S. <u>Stochastic Models, Estimation and</u> <u>Control Volume II</u>. New York: Academic Press Incorporated, 1982.
- 8. Maybeck, Peter S., Douglas A. Harnly and Robert L. Jensen, "Robustness of a New Infrared Target Tracker," <u>Proceedings of IEEE National Aerospace and Electronics</u> <u>Conference, Dayton, Ohio, pp. 639-644, May 1980.</u>
- 9. Maybeck, Peter S. and Daniel E. Mercier, "A Target Tracker Using Spatially Distributed Infrared Measurements," <u>IEEE Transactions on Automatic Control</u>, Vol. AC-25, No.2, pp. 222-225, April 1980.
- 10. Maybeck, Peter S. and Steven K. Rogers, "Adaptive Tracking of Dynamic Multiple Hot-Spot Target IR Images," <u>Proceedings of IEEE Conference on Decision and Control</u>, Orlando, Florida, pp. 1145-1151, December 1982.

BIB-1

- 11. Maybeck, Peter S., William H. Worsley and Patrick M. Flynn, "Investigation of Constant Turn-Rate Dynamics Models in Filters for Airborne Vehicle Tracking," <u>Proceedings of IEEE National Aerospace and Electronics</u> <u>Conference, Dayton, Ohio, pp. 896-903, May 1982.</u>
- 12. Mercier, Daniel E., "An Extended Kalman Filter for Use in a Shared Aperture Medium Range Tracker," M.S. Thesis, Air Force Institute of Technology, Wright-Patterson AFB, Ohio, December 1978.
- 13. Millner, Paul P., "Enhanced Tracking of Airborne Targets Using a Correlator/Kalman Filter," M.S. Thesis, Air Force Institute of Technology, Wright-Patterson AFB, Ohio, December 1982.
- Rogers, Steven K., "Enhanced Tracking of Airborne Targets Using Forward Looking Infrared Measurements," M.S. Thesis, Air Force Institute of Technology, Wright-Patterson AFB, Ohio, December 1981.
- 15. Singletery, James Jr., "Adaptive Laser Pointing and Tracking Problem," M.S. Thesis, Air Force Institute of Technology, Wright-Patterson AFB, Ohio, December 1980.
- 16. The Analytic Sciences Corporation. <u>Advanced Adaptive</u> <u>Optics Control Techniques</u>. TR-966-1. Prepared for the Air Force Weapons Laboratory, Kirtland AFB, New Mexico, January 6, 1978.
- Worsley, William H., "Comparison of Three Extended Kalman Filters for Air-to-Air Tracking," M.S. Thesis, Air Force Institute of Technology, Wright-Patterson AFB, Ohio, December 1980.

## Appendix A

This appendix contains the performance plots for the linear Kalman filter/enhanced correlator tracker configuration using a Gauss-Markov target acceleration model. These cases are identified according to the mnemonic code described in Figure V-3. The values for the parameters of these simulations are also found in Chapter V.



















a state of a state state of a

 $\sim 7.5$ 



A-10





Figure A-6a. Performance Plot for L2T2F2



•

()

Figure A-6b. Performance Plot for L2T2F2





























Figure A-13b. Performance Plot for L2T3F2








· · · •

Figure A-15b. Performance Plot for L10T3F2









Figure A-17b. Performance Plot for L20T3F2















Ŵ



Figure A-20b. Performance Plot for L2T4F2





|    | AD-R14       | 1 144 | ENH | ENHANCED TRACKING OF AIRBORNE TARGETS USING<br>MODEL FILTERING TECH. (U) AIR FORCE INST OF |  |  |  |  |  |  | MULTIPLE 3/4<br>TECH |             | 4 |  |   |
|----|--------------|-------|-----|--------------------------------------------------------------------------------------------|--|--|--|--|--|--|----------------------|-------------|---|--|---|
| IJ | UNCLASSIFIED |       |     | DEC 83 AFIT/GE/EE/84M-4                                                                    |  |  |  |  |  |  | F/G 1                | F/G 12/1 NL |   |  |   |
|    |              |       |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  |   |
|    |              |       |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  |   |
|    |              |       |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  |   |
|    |              |       |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  |   |
|    |              |       |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  |   |
|    |              |       |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  |   |
|    |              |       |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  | = |
|    |              |       |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  |   |
| J  |              | -     |     |                                                                                            |  |  |  |  |  |  |                      |             |   |  |   |



The states of the second states and the second states of the second stat

1000 A

and have well that with how and a series

3

ŧ

ł

. .

Τ.  .....



SAVES & ALCELES









LANCA I CALATA TALE







.















Call Same


## Appendix B

This appendix contains the performance plots for the extended Kalman filter tracker configuration using a Gauss-Markov target acceleration model. Thes cases are identified according to the mnemonic code described in Figure V-3. The values for the parameters of these simulations are also found in Chapter V.

22222222



CALLENAL AND A LEVEL











و که ده د د د



Sec.







B-11

STATATA A



Figure B-6a. Performance Plot for G2T2F2



R. 777 . 73

. ز. زر زر







<u>ĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸ</u>

Section of the sectio









Called at a called a contract.





B-22









-20















B-33

5614141414141414141414141

No.S








Figure B-18b. Performance Plot for GA10T2MF



B-38

Inter a second and a second second



B-39

## Appendix C

This appendix contains the performance plots for the extended Kalman filter tracker configuration using a constant turn rate target acceleration model. Thes cases are identified according to the mnemonic code described in Figure V-3. The values for the parameters of these simulations are also found in Chapter V.





Figure C-1b. Performance Plot for CT1MF







والمراجعة المحاصي والمراجع

1.1.1











Figure C-5b. Performance Plot for C2T2F1





Figure C-6b. Performance Plot for C2T2F2





C-15



Figure C-8a. Performance Plot for C10T2F2





**. Y** 







 $\sim$ 



Figure C-11a. Performance Plot for C2T3MF









Figure C-13a. Performance Plot for C2T3F2



Figure C-13b. Performance Plot for C2T3F2



Figure C-14a. Performance Plot for C10T3MF












C-34



C-35





C-37





Robert I. Suizu was born on January 14, 1959 in Honolulu, Hawaii. He graduated from Maui High School in May 1976. In May 1980, he graduated from the United States Air Force Academy with a Bachelor of Science degree in Electrical Engineering. From June 1980 to June 1982, he served as an electrical engineer at the Aeronautical Systems Division Computer Center. In June 1982, Lt. Suizu entered the Air Force Institute of Technology to pursue a Master of Science Degree in Electrical Engineering. Lt. Suizu is a member of Tau Beta Pi.

## Permanent address: 176 Ehilani St Pukalani, HI 96788

| U | IN | С | LA | SS | I | $\mathbf{F}$ | [] | ED |  |
|---|----|---|----|----|---|--------------|----|----|--|
|   |    |   |    |    |   |              |    |    |  |

UNCLASSIFIED SECURITY CLASSIFICATION OF THIS PAGE

## AD-A141144

|             |                       |                          |                          | REPORT DUCUM                                    | ENTATION PAGE                                | E                                   |                                                |            |
|-------------|-----------------------|--------------------------|--------------------------|-------------------------------------------------|----------------------------------------------|-------------------------------------|------------------------------------------------|------------|
| UNCI        | ASSIFIE               | LASSIFICA                | ATION                    |                                                 | 15. RESTRICTIVE M                            | ARKINGS                             |                                                |            |
| 2a. SECURI  | TY CLASSIFI           | CATION AU                | THORITY                  | <u> </u>                                        | 3. DISTRIBUTION/A                            | VAILABILITY OF                      | REPURT                                         |            |
|             |                       |                          |                          |                                                 | Approved                                     | for public                          | ic releas                                      | е;         |
| 26. DECLAS  | SIFICATION/           | DOWNGRA                  | DING SCHE                | DULE                                            | distribu                                     | tion unlig                          | nited.                                         |            |
| 4. PERFOR   | MING ORGAN            | ZATION                   | REPORT NUN               | ABER(S)                                         | 5. MONITORING OR                             | GANIZATION REP                      | ORT NUMBER(S                                   | )          |
| AF          | 'IT/GE/E              | E/84M-                   | - 4                      |                                                 |                                              |                                     |                                                |            |
| 6a. NAME C  |                       | NG ORGAN                 | NIZATION                 | 6b. OFFICE SYMBOL<br>(If applicable)            | 78. NAME OF MONIT                            | TORING ORGANIZ                      | ATION                                          |            |
| SCH         | JUT OI I              | nstue.                   | ering                    |                                                 |                                              |                                     |                                                |            |
| 6c. ADDRE   | SS (City, State       | and ZIP Co               | de)                      |                                                 | 76. ADDRESS (City                            | State and ZIP Code                  |                                                |            |
| Air<br>Wri∈ | Force I<br>ht-Patt    | institu<br>terson        | ate of<br>AFB, O         | Technology<br>hio 45433                         |                                              |                                     |                                                |            |
| 8. NAME (   | F FUNDING             | SPONSORI                 | NG                       | 86. OFFICE SYMBOL                               | 9. PROCUREMENT                               | INSTRUMENT IDE                      | NTIFICATION N                                  | JMBER      |
| Air Fo      | rce Wea               | ipons I                  | Lab                      | AFWL/ARAA                                       | 1                                            |                                     |                                                |            |
| 8c. ADDRE   | SS (City, State       | and ZIP Co               | de)                      | <u></u>                                         | 10. SOURCE OF FUR                            | NDING NOS.                          |                                                |            |
| Kini        | lond AE               | א מי                     | Morio                    | 0 87117                                         | PROGRAM                                      | PROJECT                             | TASK                                           | WORK UNIT  |
| V T L (     | Janu Ar               | אפא , uev                | w mexic                  | O OLITU                                         | E EMENT NO.                                  | NO.                                 | NO.                                            | NO.        |
| 11. TITLE / | Include Securit       | v Classifica             | tioni                    | ······                                          | 4                                            |                                     |                                                |            |
| See         | Box 19                | .,                       |                          |                                                 |                                              |                                     |                                                |            |
| PERSO       | AL AUTHOR             | (S)                      |                          |                                                 | <u>.                                    </u> | ·                                   |                                                |            |
| Robe        | rt I. S               | <u>uizu,</u>             | 1st Lt                   | , USAF                                          |                                              |                                     |                                                |            |
| 13a TYPE (  | DF REPORT             |                          | EBOM                     | TO                                              | March 19                                     | Rт (Yr., Mo., Day)<br>)84           | 15. PAGE C                                     | OUNT       |
| 10. SUPPLE  | <u>NESIS</u>          | OTATION                  |                          |                                                 |                                              | Reproved ton passie                 | release! IAW A                                 | FR 190-17  |
|             |                       |                          |                          |                                                 |                                              | LYNN E. WOLAVER<br>Dean for Repears | May y                                          | evelopment |
| 17.         | COSATI                | CODES                    |                          | 18. SUBJECT TERMS (C                            | Continue on reverse if n                     | WeightyPatter iden BF               | a loobadoyy (ATG<br>₿b <b>9ibiotsissu</b> mber |            |
| FIELD       | GROUP                 | ຣບ                       | B. GR.                   | Adaptive Es                                     | timation, F                                  | orward Loc                          | king Infi                                      | rared      |
|             | 05                    |                          |                          | (FLIR) sens                                     | or, Image T                                  | racking, K                          | Kalman Fi                                      | ltering,   |
| 19 48578    | CT (Continue          | on muaraa i              | 1 78684070 07            | I Target Trac                                   | king. Multi                                  | ple Model                           | Adaptive                                       | Filter     |
| Titl        | .e: ENH<br>MUI<br>ADA | IANCED<br>TIPLE<br>PTIVE | TRACKI<br>MODEL<br>FIELD | NG OF AIRBORN<br>FILTERING TEC<br>OF VIEW EXPAN | E TARGETS U<br>HNIQUES FOR<br>SION           | SING                                |                                                |            |
| Thes        | sis Chai              | rman:                    | Dr. P                    | eter S. Maybe                                   | ck                                           |                                     |                                                |            |
|             |                       |                          |                          |                                                 |                                              |                                     |                                                |            |
|             |                       |                          |                          |                                                 |                                              |                                     |                                                |            |
|             |                       |                          |                          |                                                 |                                              |                                     |                                                |            |
| DISTRI      | BUTION/AVA            | LABILITY                 | OF ABSTRA                | CT                                              | 21. ABSTRACT SEC                             | URITY CLASSIFIC                     | ATION                                          |            |
| UNCLASSI    | BUTION/AVA            | TED TE SA                | OF ABSTRA                | CT                                              | 21. ABSTRACT SECULAS                         | SIFIE:                              | ATION                                          |            |
| UNCLASSI    | SUTION/AVA            | TED TE SA                | OF ABSTRA                |                                                 | 21. ABSTRACT SECU<br>UNCLAS                  | URITY CLASSIFIC<br>SIFIE:<br>UMBER  | ATION                                          |            |
| UNCLASSI    | DED/UNLIMI            | ILABILITY                | OF ABSTRA                |                                                 | 21. ABSTRACT SECU<br>UNCLAS                  | URITY CLASSIFIC<br>SIFIE:<br>UMBER  | 4110N                                          | -          |

83 APR

| EDITION OF TIAN 75 15 OBSCILLIE |
|---------------------------------|
|---------------------------------|





Υ.

and a state in the state of the

The state of the s

1

34 34 4

ti's A

## UNCLASSIFIED SECURITY CLASSIFICATION OF THIS PAGE

Previous studies at the Air Force Institute of Technology have developed two tracker algorithms which provide significant improvements in tracker performance against close-range, highly-dynamic, airborne targets, over a currently used direct correlation method. Digital signal processing techniques are used to derive a target shape function from available sensor information. In one formulation, this shape function is used in the measurement update portion of an extended Kalman filter to determin the target position offsets from the center of the sensor field of view. In the other tracker, the offsets are derived and incorporated into the tracking algorithm by using the shape function as a template for and enhanced correlator/linear Kalman filter structure. Combining these offsets with any a priori target information allows the tracker to produce better target position estimates than achievable from a conventional correlator. This research investigates using a multiple model approach for the adaptive expansion of the effective tracker field of view as a means of increasing the dynamic range of the tracker. Two independent Kalman filters, each receiving measurement information from a shared sensor, generate target position estimates. The multiple models are created by tuning the respective filters for "best" performance at differing conditions of exhibited target behavior and differing the physical size of their respective fields of view. Adaptive expansion of the tracker field of view is obtained by summing the weighted estimates of the two filters.

ACCESSION OF A CALL ACCESSION OF A CALL

SECURITY CLASSIFICATION OF THIS PAGE

1 FILMED 6-84 DITIC